Toxic Masculinity, Male Childhood Sexual Trauma, and the Challenges to How Young Men Heal

by

Colin James Morrison

Submitted in partial fulfilment of the requirements for the degree of Master of Social Work

at

Dalhousie University
Halifax, Nova Scotia
June, 2020

DEDICATION

This work is dedicated to male childhood sexual trauma survivors everywhere:

We see you, we hear you, and we stand with you.

TABLE OF CONTENTS

LIST OF TABLESvii
ABSTRACTviii
LIST OF ABBREVIATIONS USEDvii
ACKNOWLEDGEMENTSx
CHAPTER ONE: INTRODUCTION
Prevalence of Sexual Abuse of Boys
Hegemonic Masculinity as Traditional Masculinity5
Toxic Masculine Culture
Narrative Beginnings
Purpose of this Work
CHAPTER TWO: LITERATURE REVIEW15
The Psychology of Men
Social Constructs of Gender
Learning Gender
Raising Boys to Men
Finding Foucault
Hegemonic/Dominant Masculinity43
Toxic Masculinity50
Male Childhood Sexual Trauma
Challenges around Disclosure64
Healing Processes67

CHAP	TER THREE: METHODOLOGY	76
	Research Questions and Objectives	76
	Theoretical Framework	77
	A Turn towards Narrative	79
	Research Design and Methodology	80
	Participants and Sampling	86
	Information and Consent Process	89
	Data Collection	89
	Data Analysis	90
	Rationale for Analysis	94
	Touchstones of Narrative Inquiry as Rigour	95
	Ethical Issues	103
	Conclusion	105
СНАР	TER FOUR: NARRATIVE ACCOUNTS	107
	Descriptive Overview of Participants	108
	Bryce	111
	Peter	115
	Joanna	118
	Daniel	121
	Susan	124
	Kate	127
	Nathan	129
	Gillian	132

Ben	136
Richard	140
Adam	144
Shane	147
CHAPTER FIVE: INTERPRETIVE FINDINGS	152
Invisible Victims/Hidden Survivors	153
Some Moral Distress	159
Dangerous Disclosures	165
Challenging and Renegotiating Masculinity	y171
Misogyny/Fear of the Feminine	175
Homophobia: Am I a Fag?	179
Re-Storying Trauma	183
CHAPTER SIX: ANALYSIS & DISCUSSION	191
Toxic Masculine Culture, Male Childhood How Young Men Heal	
Break the Silence/Into the Light	195
Stigma in Mental Health Systems	199
Moral Distress/Moral Courage	202
Masculine Identity and Male Privilege	206
Challenging Misogyny & Homophobia	211
Re-Storying Trauma/Reconstructing Masc	ulinity217
Pathways to Healing	220
CHAPTER SEVEN: CONCLUSION	227
REFERENCES	238

APPENDIX A	261
APPENDIX B	265
APPENDIX C	266
APPENDIX D	267
APPENDIX E	268
APPENDIX F	270

LIST OF TABLES

Demographic Information1	10)
--------------------------	----	---

ABSTRACT

This study critically examined impacts of toxic masculine culture on males who experience childhood sexual trauma and explored how myths surrounding dominant masculinity serve to impact healing. Through the perspectives of twelve mental health therapists, this work explored gaps in current research in understanding needs of survivors, and highlighted the importance of engaging with social constructs of masculinity and healing through a gendered lens. Data was analyzed through a narrative inquiry approach, utilizing thematic and discourse analysis.

Findings were consistent with current literature that described strong adherence to traditional masculine norms as having negative impacts on male survivors. Stigma related to homophobia and misogyny within a heteronormative, patriarchal society were problematic and impeded disclosure. Gendered male approaches are arguably non-existent at present in mental health systems in Nova Scotia, yet this study demonstrated how clinical service providers might be poised to intervene in important gender specific ways with young male survivors.

LIST OF ABBREVIATIONS USED

APA American Psychological Association

LGBTQI2S+ Lesbian, Gay, Bisexual, Transgender, Queer, Intersex, Two Spirited, Plus

PTSD Post Traumatic Stress Disorder

TF-CBT Trauma Focussed Cognitive Behaviour Therapy DSM-V Diagnostic and Statistical Manual, Fifth Edition

NSHA Nova Scotia Health Authority

NSCSW Nova Scotia College of Social Workers APNS Association of Psychologist Nova Scotia

BSW Bachelor of Social Work
MSW Master of Social Work

DBT Dialectical behaviour therapy
BPD Borderline Personality Disorder

ADHD Attention Deficit Hyperactivity Disorder

ACKNOWLEDGEMENTS

First of all, I want to acknowledge tremendous thanks and extreme gratitude to my thesis committee – Dr. Catrina Brown, Dalhousie University, School of Social Work; Dr. Marion Brown, Dalhousie University, School of Social Work, and Dr. Deborah Norris, Family Studies and Gerontology Department, Mount St Vincent University.

Catrina – I have been in awe of your knowledge, your passion, and your commitment to critical social work practice from the moment I entered your classroom, and it has been my privilege to take this journey with you. As thesis supervisor, I want to thank you for all the support and guidance, and the freedom to allow me to find my own voice. You will always be an important mentor of mine, and you are everything I aspire to be as a social worker, a therapist, a researcher, and a human being.

Marion – Thank you for all your thoughtful and incisive critiques. You pushed me to grow and develop my ideas while still allowing me to forge my own path. Your knowledge and wisdom are at the heart of this work and serve to make it that much better, and for that I am forever indebted.

Dr. Deborah Norris – Thank you for agreeing to be external reader. Your vast knowledge of trauma and its impacts as contributions here are much appreciated.

To Coleen Flynn, MSW RSW, I wish to express thanks and gratitude for your contributions to this study (including serving in an advisory role as supervisor through the IWK and in introducing me to the concept of "moral distress"). Thank you for all your encouragement and guidance, and for keeping me laughing throughout it all – I am proud to be part of "the Cols".

I would like to thank the many inspirational professors I have had throughout my MSW experience at Dalhousie - particularly Senator Wanda Thomas-Bernard, Dr. Cassandra Hanrahan and Dr. Jeff Karabanow - who have helped develop my critical thinking skills and ignite my passion for fighting social injustices through a welcoming learning environment filled with thoughtful, open debate.

I would also like to thank Professor Ginette Abraham and Louise Sabourin of the University of Manitoba for their encouragement in pursuing an MSW and ultimately thesis work.

I wish to thank the family and friends of the Lawrence Hancock Scholarship, the Sonia Weil Bursary, and the Margaret Cragg Award for their tremendous financial support that helped to make this work possible.

I wish to thank all of my MSW classmates – in particular Katrina, Kaitlyn, and Jillian. I am so proud to have taken this MSW journey with you all, and I believe together we *can* and *will* change the world.

Thank you to the twelve amazing clinicians at the heart of this work for being so brave and generous in sharing their stories of experience. I will be a far better social worker and therapist just by virtue of knowing you all.

Thank you to all IWK Managers past and present who have supported me throughout this work, and my coworkers and colleagues, all of whom continue to inspire me every day.

Thank you to my friends for their enthusiastic and unwavering support. I especially want to thank Karen Carey for all the fun, friendship, and laughter over the years, and for encouraging me to follow in her path and return to academia.

Thank you to Barry Nahirnak for his wise counsel and Dr. Kevin Carbyn for his compassionate care.

Thank you to my family for always loving and supporting me, even when you did not always understand my path – although we do not see each other often, you are always in my heart. To Mom, dad, and Raylene – I miss you every day, and while you are not here to share in this moment, I know that you are somewhere together watching and you are proud.

Thanks to Carol for her kind heart and for accepting me as part of her family, and for raising on her own such a strong, caring, and compassionate son.

Thanks to Edward and Mungo, two of the sweetest ginger cats that ever existed, for all the times they kept me good company while I was researching and writing long into the night.

And finally, to my partner Shawn Miner: Thank you for always picking me up whenever I fall, and for lending me your strength when it seems I had none of my own. This work would not exist without your endless encouragement, enthusiasm, and support, and it is your countless hours of constructive feedback and thoughtful critiques that have served to make this work so much better. You make the world a much better place just by being in it, and I am so lucky to be on this journey through life with you. You show me every day what it means to be a good man, and my love for you truly knows no bounds.

CHAPTER ONE: INTRODUCTION

This study critically examines the impact of toxic masculine culture on young men and boys who experience childhood sexual trauma and explores how myths and assumptions surrounding dominant masculinity in the world today serve to influence, impact, and potentially disrupt healing processes for young male survivors. The study also examines how we might work to reduce the stigma, shame, and silence that surrounds this phenomenon, and seeks to contribute to literature around the experiences of young men and boys and early childhood sexual trauma. The study presents an overview of the experiences of young male survivors understood through the perspectives and personal narratives of mental health clinical therapists, those who counsel and provide therapeutic interventions to this population. Through this work, I critically examine gaps that exist in current research in understanding the unique needs of survivors, explore constructs of masculinity and male identity as it relates to experiences of childhood sexual trauma, and highlight the importance of engaging with these issues through a gendered, male-centric lens.

Chapter one outlines the framework of this study and provides some background to the research topic. Therein, I present a working definition of the term "toxic masculine culture" that places it within the context of male survivors' experiences with early childhood sexual trauma and abuse. Chapter two explores relevant literature pertaining to hegemonic or dominant masculinity and toxic masculine culture, early childhood sexual trauma, boyhood socialization, and a brief overview of treatment and healing processes. Chapter three contains the methodology section, including a discussion of theoretical foundations, an outline of the recruitment processes, and an overview of ethical considerations. Chapter four presents the narratives of the study participants as mental health therapists engaged with this unique population. Chapter five

discusses the interpretive findings of the research in a results section. Chapter six provides an analysis and critique and considers some future implications for social work practice. Lastly, chapter seven offers an overall summary of the work, closing remarks, and a final conclusion.

This study explores narratives of clinical therapists detailing their work with young male sexual trauma survivors. Their narratives of that work consist of stories within stories, as participants shared their interpretations and meaning making of the narratives of those under their care. The study examines and interprets the effects of toxic masculine culture on young males through their narratives and considers the resulting impact on their healing process. It also explores the complex gendered nature of the treatment work participants provide to survivors. Research questions that guided this study are: How does the experience of childhood sexual trauma impact the lives of young survivors? In what ways does the phenomenon of childhood sexual trauma intersect with the experiences of being socialized as a young male? How do expectations of dominant masculinity – and particularly those of toxic masculine culture – impact or disrupt healing processes of young survivors? What current treatment interventions and strategies are utilized in treating male survivors, and what appears most effective and what might be missing or lacking? How might we reduce the stigma for males associated with being a survivor of childhood sexual abuse and address the dominant discourse related to poor mental health help- seeking behaviours in males?

I chose to use the word *trauma* to describe these early childhood experiences in place of *abuse*. I understand abuse as the violation, disrespect, or harm that befalls or is visited on a person, while I interpret trauma as the physical, emotional and psychological impact of that abuse. Trauma overwhelms a victim, taking away control, severing connection, and disrupting meaning in their lives (Herman, 2015). Not all men and boys who experience childhood sexual

abuse are traumatized, and many go on to live healthy, productive lives without intervention. However, this study focuses on the lives of those who do experience trauma and a loss of control, connection, and meaning in their lives, and the complex interplay between their traumatic sexual experiences and their masculine identity. My hope for this work is to bring attention and shed light on a topic that is poorly understood and not well recognized, to better capture and define some of the more effective ways and means of providing treatment and service, and thereby increase capacity in others working with this unique and underserved population.

Prevalence of Sexual Abuse of Boys

Research shows as many as one in six men are survivors of early childhood sexual trauma histories, yet many clinicians in the field believe that this number is a gross underestimate of those truly affected, noting how many young men and boys struggle to recognize or define the experience itself as abusive, and possess little understanding of the possible impacts or effects of trauma and how they may relate to present day struggles. Men and boys also struggle in disclosing their experiences to others, initiating help-seeking behaviours, and finding appropriate resources (Cohen et al., 2017; Fisher & Goodwin, 2008; Van der Kolk, 2014). Men and women sexually abused during childhood share many commonalities in the psychological and social aftermath they endure following these experiences, and yet for as many similarities that exist there are an equal number of differences. Of particular focus in this study is the unique way for male survivors that victimization intersects with gender socialization and the formation of sexual orientation identity (Gartner, 2017a; Fisher & Goodwin, 2008). As a result of this intersection, men abused in childhood can often face several harmful myths that can serve "to amplify their trauma, limit the services available to them, and block their entry into a healing process" (Fisher

& Goodwin, 2008, p. 2). This study will demonstrate how these myths are often directly related to that crucial period of adolescent discovery and sexual identity formation, and how that confusion, loss and disruption has staggering and long-lasting impacts for many survivors.

Perpetrators of sexual abuse on young males are often adults who have violated positions of power and trust which results in a shattering of the natural trust he possesses in adults and caregivers (Gartner, 2017a). Trauma survivors are potentially left suffering from damaging effects, with little awareness of how impactful and harmful their experiences may have been over the course of their young lives, or how inter-connected trauma may be to any present-day difficulties or challenges (Fisher & Goodwin, 2008; Gartner, 2017a). These unwanted experiences can have severe lifelong impacts on some of survivors and, if left unresolved, can lead to more severe psychological issues like anxiety, depression, and other mood-related disorders, which in turn can promote high risk behaviours involving substance misuse, sexual compulsivity, self-harm, or eating disorders (Fisher & Goodwin, 2008; Gartner, 2017b; Van der Kolk, 2014). Research now shows a clear link between these same resulting high-risk behaviours and long-term physical health consequences like sexually transmitted diseases, cancer, heart disease, and obesity (Cohen et al., 2017; Currie & Spatz-Widom, 2010; Herman, 2015; Van der Kolk, 2014). Furthermore, the impact of delaying recovery can have serious impacts on the struggling survivor's life and a ripple effect on family and community around him, including social dysfunction issues that may lead to failed relationships, intimate partner violence and significant impacts on career or education goals and objectives (Gartner, 2017b; Lisak, 1994, 2017; Van der Kolk, 2014).

In our school and mental health care institutions, these same young men and boys are sometimes wrongfully misdiagnosed with a wide range of mental health disorders, and

subsequently treated with multiple medications and therapies that ultimately prove ineffective because the clinical formulation is unwarranted, the interventions target only a small cluster of symptoms, or most importantly for sexual trauma survivors, the treatments fail to address underlying root cause and effect of early childhood trauma (Cohen et al., 2017; Fisher & Goodwin, 2008; Herman, 2015; Van der Kolk, 2014). As a result, I argue throughout this work that the current services offered through our mental health systems, with their emphasis on brief interventions and short term therapeutic approaches, are often poorly designed and dangerously ill- equipped to address the unique therapeutic needs of trauma survivors.

Hegemonic Masculinity as Traditional Masculinity

Hegemonic masculinity, as described by Raewyn Connell (1995) refers to the socially constructed dominant form of masculinity that is accepted, privileged and exalted over femininity and other forms of marginalized masculinities, including gay, trans, or gender diverse males. It is an idealized form of male behaviour, which privileges the position of some men, a position they are strongly encouraged to aspire to, while it ideologically legitimates the subordination of women and all things considered feminine. Character traits of hegemonic masculinity include physical strength, wealth and power seeking, risk taking, invulnerability, virility, stoic or suppressed emotionality, control and dominance, excessive competitiveness and a rejection of femininity (Connell, 1995; Connell & Messerschmitt, 2005). While portrayed as only one of many different ways men and boys enact masculinity through its social construction, hegemonic masculinity has historically been recognized as the most privileged and sacred among them (Connell, 1995; Connell & Messerschmitt, 2005; Seidler et al., 2017).

It is embodied at the specific intersections of race, class, ableism and sexuality (Collins, 2015), currently defined in Western culture as white, wealthy, able bodied, and heterosexual. Central to hegemonic masculinity is heterosexuality, constructed as a position that is as much 'not gay' as it is 'not female' (Jewkes et al., 2015). These privileged social identities interact and intersect in specific ways that exclude certain groups of men due to their perceived devalued membership, such as those who are racialized, gay, disabled or working class poor.

In January of 2019, the American Psychological Association (APA) released a series of recommendations centred on gendered mental health treatment approaches towards males (APA, 2018). Titled *Guidelines for Psychological Practice with Boys and Men*, these recommendations, reportedly drawn on over forty years of research, described hegemonic masculinity (sometimes referred to in the report as "traditional masculinity") as psychologically harmful when combined with strict and rigid adherence to masculine norms. It also deemed the Western practice of socializing boys to suppress the depth and range of their emotions as causing damage both internally and externally (Pappas, 2019). Traditional masculinity, when combined with extreme adherence to gendered expectations, manifested in harmful and problematic ways for boys and men but also for those closest to them. This strict adherence was harmful and damaging to men in terms of their physical and mental health by often involving excessive substance use, fighting or risk taking, body dysmorphia, and challenges in expressing emotions. It can further present as damaging to others, in terms of violence, transphobia, misogyny, homophobia, xenophobia, racism, and sexual assault or harassment (De Boise, 2019; Pappas, 2019).

Toxic Masculine Culture

The Good Men Project (2020) describes toxic masculinity as a narrow and repressive description of manhood, reducing masculinity to a cultural ideal of manliness, emotions as weakness, and sexual pursuits and forced aggression as measures of success while supposedly "feminine" traits – ranging from emotional vulnerability to simply not being hypersexual – means by which your status as a man might be taken away or negated. Over the past number of years, "toxic masculinity" has become a catchall phrase used to describe male violence and sexism in modern culture. The appeal of the term, which distinguishes "toxic" traits such aggression and self-entitlement from "healthy" ones such as strength and confidence, has grown to the point that the American Psychological Association, while not explicitly labelling it as toxic, warn that extreme forms of certain "traditional" masculine traits are linked to aggression, misogyny, and negative health outcomes (APA, 2009).

Shortly after its release, the APA *Guidelines for Psychological Practice with Boys and Men* was met with considerable controversy and conflict, with some critics describing it as a declaration of war on traditional manhood, while others saw it as helpful in acknowledging and challenging misogynistic behaviour and homophobic behaviour and violence towards women and the Lesbian, Gay, Bisexual, Transgender, Queer, Intersex, Two Spirited, + (LGBTQI2S+) community (De Boise, 2019; Pappas, 2019). The term toxic masculinity itself became highly charged and politicized, and amid heated discourse, popular media has blamed toxic masculinity for rape, mass shootings, gun violence, online trolling, climate change, and the election of Donald Trump (Salter, 2018). Others have argued using a term such as 'toxic masculinity' is problematic in that it releases men of responsibility for troublesome behaviour by positioning

them as victims of a broader vague entity, rather than highlighting their agency in the social construction and reproduction of masculinity (Beasley, 2013; Waling, 2019a).

I believe that the term toxic masculinity is a repressive and restrictive way of defining manhood, and one that essentializes the way masculinity is socially constructed and portrayed in our world today. My intention is to not essentialize masculinity as fixed or problematic, and in fact, I will draw heavily from social constructionism and postmodern theories of discourse on gender with a view of all gender as fluid, socially situated and socially constructed. Within this work, I speak of the male experience with sexual trauma and gender constructs because that is my understanding and my lived experience, and because I believe it is an area of exploration in our culture that has long been ignored or silenced. For an example of this, we need look no further than to the Nova Scotia Sexual Violence Strategy, released in 2016 as the first provincial coordinated response to sexual violence, with a focus on improved service delivery and access to supports and resources for sexual violence victims and their families. The strategy noted that youth and children make up 20% of the population of Canada, yet they accounted for 55% of all victims of police reported sexual offences (Nova Scotia Department of Community Services, 2015). It also noted that while women and girls were largely more likely to be impacted, victims of sexual violence also include men, boys, and LGBTQI2S+ persons. While acknowledging there was a need for more focused and increased supports for these populations, those needs and their required supports and resources were not explicitly recognized or addressed.

I recognize within this work that girls and women are statistically much more often impacted by sexual violence as victim and survivor, with research that shows as many as one in three girls will likely experience some form of sexual trauma in their early lifetime (Collin-Vezina et al., 2013; Herman 2015). Furthermore, I believe that it is the work that women have

done to advance the conversation around sexual trauma and its impacts that have allowed space for research on male sexual trauma survivors such as this to exist. So, while I urge for more attention and focus on male experiences of sexual trauma, by no means is that meant to detract from the very significant societal issues we face with violence perpetrated by men and boys towards women and girls. I also recognize that gender expectations can impact everyone, and that women can face impossible standards in regard to dominant or idealized forms of femininity. Boys are not naturally strong and aggressive just as girls are not inherently weak or passive – rather, these are stereotypes related to cultural expectations placed on boys and girls that can lead to a form of behaviour that is toxic. This is not because masculinity itself is innately or inherently bad, but it is because the way boys are socialized in our culture is often harmful and potentially dangerous for everyone. Independence, self-reliance, and stoicism can be admirable traditional male traits, but when a man is unable to ask for help, feels incapable to rely on others, or struggles in the expression of vulnerable emotions of any kind, those traits become increasingly problematic (Augusta-Scott, 2020; Fisher & Goodwin, 2008).

Experiencing trauma in childhood can encourage men to deny their victimization, because to be a victim is to be perceived as vulnerable or weak (Augusta-Scott, 2020; Fisher & Goodwin, 2008). Ironically, by not acknowledging their victimization and seeking to perpetually prove they are not vulnerable and therefore not victims, their trauma remains unresolved and all-consuming of their focus (Augusta-Scott, 2020). Research shows that masculinity, when associated with rigid, restrictive enactment and enforcement of traditional male gender roles, can manifest as toxic both in its associated behaviours and its significant harmful impacts and negative influences on the positive health, recovery and overall well-being of males who are survivors of early childhood sexual trauma (Englar-Carlson, 2014; Fisher & Goodwin, 2008;

Gartner, 2017b; Iwamoto et al., 2018). When describing a toxic male culture, I am not describing masculinity itself as toxic, but the form of gendered behaviour that manifests when expectations of what it means to be masculine fail to match up with reality. A toxic male culture is one that implores young men and boys to grow strong and tough, to hide or suppress many of their outward emotions, to present themselves (often at times rigidly) as strong, independent, assertive, stoic, athletic, sexually potent, successful, and powerful, and to believe that anything less may imply a failure of masculinity (Corbett, 2016; Gartner, 2017a; Kimmel, 2015; Kupers, 2005). While hegemonic masculinity shares many characteristics with what I describe as toxic masculine culture, I maintain that the term "toxic masculine culture" is helpful because it helps to delineate and differentiate between those aspects of hegemonic masculinity that are most harmful or socially destructive, such as misogyny, homophobia, and transphobia, and those that are most honoured or valued, such as courageousness, self-reliance and the desire to provide protection and promote wellbeing in others. Through this work, I will explore this crucial distinction, and deconstruct how strict and rigid adherence to masculine norms can prove especially problematic to young male survivors of sexual trauma in terms of their healing and recovery process.

Narrative Beginnings

As a researcher, I come to this project as a queer cis gendered male, a social work scholar, a youth care worker, a case manager, and a mental health clinician. I have worked with this unique population of male sexual trauma survivors and witnessed first-hand the devastating impact of unresolved and untreated trauma experiences in young men and boys and the farreaching effects on their lives and those around them. I have been challenged and I have struggled with membership in a health care system whose predominant approach to this

phenomenon either pathologizes or minimizes the very severe and long-lasting implications of trauma. My positionality within this work is grounded by my own experiences of childhood sexual trauma and my interactions as a survivor within systems and agencies that failed to protect me or to provide the even the most basic care and treatment I so desperately sought at a young age. These early experiences with trauma led personally to troubles with anger, challenges with healthy relationships, disruptions at school, and problematic substance use in the past many of the same issues my work experience in mental health and the research literature explored in this study has shown as commonplace among male childhood sexual trauma survivors.

As a gay male who grew up in a small community that was heteronormative and often largely homophobic, I have faced many challenges within my own understanding, portrayal, and enactment of masculinity. I was raised in a family where my three primary male role models — my father and two older brothers — worked as a firefighter, a naval diver, and a truck driver respectively, and all subscribed to very traditional and stereotypical male gender role patterns. Meanwhile, I was a queer kid who preferred books, art, dance and nature. In struggling to fit in, I identified much more strongly with my sisters and their interest — interests that were deemed too "feminine" for a boy — and therefore rejected many of those traditional masculine norms that were expected. But over time, to avoid ridicule, I had to consciously be aware of things like the way I spoke or carried myself, or the way I dressed. My experiences of gender and how masculinity was performed and enacted was far removed from the traditional normative ideals of my family. I was loved and cared for by my mother and, although at times distant and removed, my father as well, and I shared a close bond with both my sisters. Yet every single day of my childhood was a struggle to belong combined with pressures to conform, and at times my

brothers, in their relentless teasing, were my worst tormentors. Therefore, while I cannot claim to be neutral in exploring the accounts of my participants in their trauma work due to my personal history and narratives, I recognize that all researchers come to their research with prior knowledge, experience and social location that serve to inform and influence their work. Through my experiences, I believe I am uniquely positioned and qualified to explore this research as a survivor of childhood sexual trauma and as someone who has dedicated so much time and effort to helping others who have experienced similar dark journeys and pathways. My work specifically has also afforded me some privilege as an insider, as a mental health colleague to those who share their work narratives and as a fellow survivor to those young boys and men under my care. It is through that privilege that I have learned and grown as person, and how I came to understand the power of the survivor's voice in telling their narrative. For that reason, I choose to illustrate and highlight the voices of my participants and their understanding of the male sexual trauma survivor's voice, and therefore position them as the true experts in this phenomenon and in this research.

Purpose of this Work

The capacities that men and boys have for processing vulnerable emotions can be fostered and cultivated, and with support and guidance can come healing and recovery from trauma. However, the healing work for young male survivors is often explicitly complicated due to the intersection between male boyhood socialization and male sexual victimization (Fisher & Goodwin, 2008; Gartner, 2017b). Within the context of masculine gender socialization, many young men and boys may struggle to perceive themselves as potential victims, which hearkens back to the dominant discourse that a man as victim equates a man who is not strong, powerful, or able to protect himself, and therefore "victimhood" constitutes a potential failure of

masculinity (Corbett, 2016; Gartner, 2000; Kia-Keating et al., 2009; Kimmel, 2012; Kupers, 2005). Under the shadow of abuse, these experiences are very much at odds with one another, setting up powerful and potentially damaging discourses within lives of young male survivors (Fisher et al., 2008; Gartner, 2017a; Lisak, 1994; 2017). Furthermore, as David Lisak (1994) so eloquently tells us, the path to recovery from childhood sexual trauma "winds straight through masculinity's forbidden territory: the conscious experience of those intense, overwhelming emotional states of fear, vulnerability, and helplessness" (p. 262).

As a mental health professional engaged with this distinct population, I will argue that it is crucial to explicitly acknowledge, address, and explore the complex intersection between how young men and boys who experience sexual abuse are at the same time experiencing what it means to be socialized as a man in our world today. Boys and men are often conditioned through socialization to think that the only way to remain safe is to feel angry and avoid any vulnerable emotions (Augusta-Scott et al., 2017). As part of treatment processes, I argue that we must work to actively contend with the intersection between childhood trauma and male socialization through exploration and unpacking and consider resulting impacts on the healing process. Trauma influences masculinity by normalizing avoidance of pain, fear, and sadness, while encouraging only the expression of anger (Augusta-Scott, 2020). Within our mental health systems, the reaction to this expression of anger is often pathologizing in that survivors are either denied entry to service based on what is deemed too hostile a presentation or an unwillingness to engage, or their problems and issues are mislabelled or misdiagnosed based on what is quite possibly a valid and honest response to past unresolved traumatic experiences. These unhelpful responses from professionals, combined with rigid and unrealistic rules and expectations around societal norms of masculinity and systemic barriers that limit pathways to treatment, may serve

to compound experiences of trauma and move young men away from the very capacities they need to heal and overcome the negative impacts of childhood sexual trauma. It is my hope that this work will contribute to the conversation about how we approach problematic notions of masculinity that impact male sexual trauma survivors and offer agency in our work so that they may process emotions and overcome their experiences without shame, humiliation, or defeat.

CHAPTER TWO: LITERATURE REVIEW

This chapter will explore relevant literature pertaining to male early childhood sexual trauma, challenges with disclosure, the psychology of men, hegemonic/dominant masculinity, toxic masculine culture, and a brief overview of current treatment and healing processes. It will also serve to demonstrate the significant gaps in current literature related to experiences of male early childhood sexualized trauma and highlight both the relevance and importance of this research topic and current study.

The Psychology of Men

Interest in the psychology of men grew as a study topic in academia decades ago and saw application in feminist theory around notions of patriarchy and its harmful impacts (Connell, 2005; Fisher et al., 2008; Kimmel, 2015; Levant & Pollack, 1995). It also served to contextualize social determinants of health relative to male experiences and to increase awareness of male antiviolence campaigns (Fisher et al., 2008; Kimmel, 2012; Levant & Pollack, 1995). Fields in academia including sociology, anthropology, literature, and cultural studies also saw a significant increase in scholarly writing that used the feminist critique of traditional gender roles as a starting point for a discourse on masculinity and a cultural analysis of masculine traditions, as well as producing new portrayals of men and boys in literature and popular culture media (Cochran, 2010; Levant & Pollack, 1995).

Masculinity has been described as a "culturally defined phenomenon, contradictory and inconsistent in nature, and individually enacted by men in specific situations that differ and vary across different ethnic groups, family traditions and cultures" (Cochran, 2010, p. 50). Kimmel (2012) says that:

Manhood is neither static nor timeless; it is historical. Manhood is not the manifestation of an inner essence; it is socially constructed. Manhood does not bubble up to consciousness from our biological makeup; it is created in culture (p.120).

Therefore, masculinity is not a fixed biological state but rather an ever-changing social construct, historically and culturally related. Masculinity refers to the behaviours, relationships, and social roles of men within any given society, and the different ways of being a man. Schrock and Schwalbe (2009) state that studying masculinity involves a marked shift from studying the male sex role and masculinity as a singular trait to examining how men enact diverse masculinities in different contexts. This means understanding men as gendered beings in various contexts – for example, from as individuals, to relationships, to the workplace, and to global politics.

When boys or men violate, deviate from, or fail to meet gender norms of masculinity, they may experience gender role strain (Pleck, 1995). As interpreted by Joseph Pleck (1995), the concept of gender role strain sought to conceptualize the psychological and interpersonal conflicts in both men and women that theoretically stemmed from the application, interaction, and intersection of traditional gender codes (Fisher & Goodwin, 2008; Levant, 2011, Pleck, 1995). Research based on this paradigm focused primarily on examining how gender role conflict potentially impacts men's' psychology, stress, and health outcomes (Cochran 2010; O'Neil, 2013). While acknowledging fundamental differences between genders, gender role strain holds that these concepts of masculinity and femininity were socially constructed from biological, psychological, and social experience and further influenced and dictated by society and culture (Fisher & Goodwin, 2008; Kimmel, 2015; Levant & Richmond, 2007; Pleck, 1995). This view purported that males and females essentially share the same fundamental humanity but that cultures obscure this sameness through an enforcement and policing of gender codes.

The concept of trauma strain has been applied to specific groups of men whose gender role strain may at times be experienced as particularly harsh, rigid, or limiting (Levant, 2011). By virtue of their race, occupation, or sexual orientation, this potentially includes racialized men; gay, bisexual, trans, or gender diverse men; professional athletes; military; veterans; first responders; and survivors of childhood trauma (Levant, 2011; Lisak, 1994; Messner, 2004). Furthermore, it is also widely acknowledged that gay and bisexual men are often traumatized by gender role strain (Levant, 2011; Sanchez et al., 2010), largely due to homophobia and marginalization through being raised in a heterosexist society with dominant expectations around masculinity.

By adopting this framework for the psychology of men, scholars and theorists were able to "break with the then-dominant academic view of masculinity as an inherent, essential, and universal expression of biological maleness" (Levant, 2011, p. 765), something that arguably still remains for many the dominant discourse today. Interest in the psychology of men has also allowed for a critical examination of traditional norms of masculinity, such as an emphasis on aggression, extreme self-reliance, and restrictive emotionality, and to view certain problems that are prevalent among men, such as the devaluing of women or sexual minorities, the detachment from relationships, or the neglect of certain health needs, as both an unfortunate yet predictable potential outcome of male gender socialization processes informed by restrictive traditional masculine ideologies (Levant, 2011). As a result, a deeper understanding of the psychology of men is warranted in our mental health models and among mental health practitioners to help inform more effective treatment processes and outcomes. This understanding may help increase our understanding of why men are disproportionately represented in population groups such as men who use violence against women and children, men who struggle with problematic

substance use, men who are incarcerated, victims of suicide and homicide and the homeless (Gartner, 2017a, 2017b; Levant, 2011). It also helps in the development of a positive new vision of what it means to be a man in the world today, and how that vision contributes to the healthy development of men, women, children, and families (Levant, 2011).

Some gender scholars such as Robinson (2003) and McCarry (2007) argue that focusing on the experiences of women and other subordinated groups is essential in working towards gender equality. Focus on studying men and centering on their experiences therefore leads to a significant risk of drawing attention and resources away from women's' issues, which in turn allows for continued support of male dominance in a patriarchal society (Peretz, 2016). At the same time, feminist scholars and activists are increasingly recognizing the importance of addressing men and understanding their experiences as a means of analyzing gender relations and increasing support for gender equality (Connell, 1995; Gardiner 2000; Kimnel, 2015; Pascoe 2007; White & Peretz 2010).

An overview of the literature clearly demonstrates the vital impact feminist scholars had not only on the study of men but on gendered male approaches to the treatment of men. As Fisher and Goodwin (2008) note, the women's movement introduced the idea of gender analysis in research and theory and gender awareness in therapeutic practice. Masculinity studies owes feminism a tremendous intellectual debt as an example of how to question, challenge, and deconstruct patriarchal power and privilege and unpack and explore gender construction and meaning (Kimmel, 2015; Waling, 2019). Feminist contributions gave momentum to a male-centred approach with an applied gendered perspective to the study of men (Lisak, 1994). Other theorists, however, have criticized masculinities studies' commitment to feminist roots noting how in much of the related literature previous feminist work was disregarded or misrepresented,

particularly in addressing gender inequality and power relations (McCarry 2007, Robinson, 2003). Some, like Robinson (2003), argue that male masculinity theorists draw selectively on feminist theory, engaging only with certain perspectives that appear sympathetic to male issues or problems. Anthony Messner, a self-described pro-feminist masculinity theorist, believes that many prominent texts on masculine studies do not acknowledge feminist theory at all, and the literature that does explore it often appropriates forms of feminism that de-emphasize key issues related to sexual politics between men and women (McMahon, 1993). As Connell (1995) notes in her work, the challenge that presents itself in studying masculinity is the tension in asking men to examine their practices in order to reject those practices, even those they may benefit from, whether individually or collectively.

Overall, the literature demonstrates how gendered perspectives on masculinity have been helpful in particular in understanding men's use of violence by helping to illustrate problematic gender norms, masculine ideals such as emotion restriction and conformity to aggression, and the notion of men's self-perceived failure of living up to masculine expectations. It provides greater insight into the unique ways violence and aggression can intersect with masculine norms, and the potential for impact on physical and mental health. Studying masculinity also allows for further exploration of the work of feminist scholars and activists around the feminization of victimization and the masculinization of oppression. However, critics of men and masculinity studies are concerned about implicit messaging in the literature that disembodies and reifies masculinity (McCarry, 2007, Waling, 2019), which can have a profound effect on how we understand male violence. If we attribute masculinity and the impact of gender norms on certain men as the root of male violence, then there is a risk placing blame on masculinity as some disembodied yet highly influential fixed entity, rather than the behaviour of the men and boys

who perpetrate and perpetuate that violence as part of their culture. Waling (2019) and Beasley (2005) argue that violence against women is often heavily premised on the positioning of women as lesser in society. Therefore, to consider male violence as an effect of toxic masculinity "disembodies men from their actions... [and] denies the long standing history in which women have been consciously been systematically and institutionally marginalized by men" (Waling, 2019, p.370). I argue that this interpretation of the interconnectedness between male violence and toxic masculinity is problematic as it essentializes one dominant form of masculinity with anger as its expression and strips agency from men and boys in both managing and taking responsibility for their aggressive or violent behaviour. The challenge is that these behaviours exhibited by some male survivors are often trauma influenced (Augusta-Scott, 2020) and reinforced by culture's narrow and conformist view of masculinity. Although it serves to advance the conversation around men's use of violence and bring it into the public discourse, the tendency to associate victimization solely with females and perpetration solely with males – and by extension, females as non-perpetrators - is problematic in our understanding of male childhood sexual trauma because it erases the possibility of men and boys as victims, a theme that will be explored further throughout this work.

Social Constructs of Gender

A social constructionist paradigm of gender views masculinity as culturally derivative, contextual, constrained, and often contested within social and interpersonal interactions that are directly related to that individual's experience (Addis & Mahalik, 2003; Cochran, 2010; Jewkes et al, 2015). While gender roles may be reflective of traditional or dominant cultural norms, they are neither considered completely universal nor necessarily entirely healthy (Cochran, 2010), and can serve as explicitly unhealthy in their enactment. Socially constructed men's gender roles can

therefore demonstrate problematic ways individual men enact gender, which may result in barriers to their access to health care and treatment. This may compound problems if physical and mental health interventions are not sought out when necessary or in a timely fashion (Addis & Mahalik, 2003; Cochran, 2010).

However, critiques of gender theory and gender identity models have noted reliance on false, essentializing assumptions that men need to subscribe and conform to a dominant/hegemonic form of masculine behaviour in order to experience successful, healthy development (Kilmartin & Smiler, 2015; O'Neil, 2013). Such approaches to gender theory and identity models have also been criticized for assuming that efforts to follow and adhere to cultural expectations are inherently problematic, because it fails to take into consideration a more positive or desirable outcome or reality, such as traditional male characteristics like strength of character, self-reliance, and a desire to protect other. It also does not acknowledge space for multiple ways to enact masculinity, including those of queer men or those who identify as gender diverse or gender fluid (Kilmartin et al., 2013; Kimmel, 2008; O'Neil, 2013). Kiselica and Englar-Carlson (2010) argue that focus on strict and rigid adherence to male traditional norms and the negative impact of traditional gender roles on men's health is problematic. They suggest a shift in future research on both positive social and adaptive aspects of the psychology of men and boys, such as a willingness to persevere and sacrifice personal needs to provide for others may offer a more complex and nuanced understanding of the construction and performance of masculinity (Addis et al., 2003; Cochran, 2010, Kia-Keating et al., 2009; Kimmel, 2012). At the same time, researchers also make a compelling argument for the need for research to explore the context of the social construction of gender roles, and to consider and investigate more fully those aspects of masculinity that are related to men's violence against women, other men, and

children (Addis et al., 2003; Cochran, 2010, Kia-Keating et al., 2009; Kimmel, 2012; Gartner, 2017a; 2017b).

Where dominant forms of socially constructed masculinity exist, there are also subordinate ones (Connell, 1995; Connell et al., 2005). Dominant/hegemonic masculinity has historically been based on the exclusion of men where not white, privileged, cis-gendered, ablebodied and heterosexual (Liu, 2005). Some forms of masculinity may generally be deemed more highly socially acceptable, such as the privilege Western society gives to white, heterosexual, middle class males, and are associated more closely with authority, social power, and influence within society (Connell & Messerchmidt, 2005). In heteronormative society, LGBTQI2S+ youth who identify or are perceived as feminine are typically subordinated and socially marginalized as they do not fit dominant masculine stereotypes and subsequent expectations because heteronormative assumptions falsely conflate masculinity and sexual identity for men and boys. Therefore, in those instances, the construction, intersection, and renegotiation of masculinities can potentially occur in a highly oppressive and constraining environment of homophobia, transphobia, misogyny and heterosexism (Addis et al., 2003; Kia-Keating et al., 2009; Kimmel, 2012; Gartner, 2017a).

Racialized and ethnically and culturally diverse men and boys are often subject to various forms of prejudice, marginalization and micro aggression and experience conflict between dominant/hegemonic masculinity and their experiences as marginalized (Kiselica et al., 2011; Liu, 2005; Liu & Concepcion, 2010). Men and boys who are racialized, LGBTQI2S+ or lower class, working poor men, among other intersecting identities, can face painful and marginalizing experiences that lead some to conform to dominant masculine ideals rather than face disapproval or adopt behaviours that endorse a form of masculinity that may conflict with their preferred

identity (Kimmel, 2008; Liu & Concepcion, 2010). Marginalizing experience can also lead to these men and boys rejecting their cultural or sexual identity or social class identity as a result. But men and boys can still have agency and in fact often resist and part from traditional norms and create their own communities. Men and boys are diverse in terms of their race, culture, socioeconomic status, abilities, sexual orientation and gender identity, and these identities intersect and interact in unique ways that contribute to their physical, relational, and psychological outcomes in positive and negative ways (Kimmel, 2012; Kiselica et al., 2011).

The sex binary allows for only two rigidly fixed roles or options, either male or female, and yet within modern Western society, we more readily acknowledge today there are many other sex and gender roles available, and that sexual orientation is wholly separate and distinct from sexual gender identity (Kilmartin et al, 2015; Kimmel, 2008). However, dominant discourses maintain there are two distinct sexes (male and female), two gender roles (masculine and female) and three possible sexual orientations (heterosexual, bisexual, and homosexual), when in fact there are multiple ways of constructing and enacting sex, gender roles and sexual orientations. These constructions include, but are not limited to, asexuality, pansexual, transgender males, transgender females, gender fluid, gender queer, gender diverse, non-binary, and intersexed people (Kilmartin et al., 2015, Stainton-Rogers; 2001).

In *Gender Trouble* (1999), Judith Butler examines the division between socially constructed gender and presumed biological sex and describes how that division ultimately fosters the illusion that the subject's gender is grounded in some fixed and binary biological essence. Within Western culture, sex, gender, and sexual orientation are viewed as essential qualities, with biological sex as binary (i.e., male vs. female), essential, and natural, and that it forms the basis for binary gender, the cultural interpretation of sex, and sexual desire. This is the

belief that a baby born with a penis will grow up to identify and act as a man as defined by his specific culture, and, as part of an expected gender role, be sexually attracted to women. Similarly, there is a belief that a baby born with a vagina will grow up to identify and act as a woman and be sexually attracted to men. These configurations of sex, gender, and sexual desire become the only "intelligible" genders – the normative accepted gender identities - in hegemonic patriarchal heterosexual culture.

Butler (1988) argues that gender should not be viewed as biological fact or an internalized state of being, but rather as an enacted performance and an active way of doing and being. Butler (1988) believes that there is a meaningless distinction between sex and gender, noting evidence for the variability in chromosomes, genitalia and hormones that do not always align in the expected binary manner. Drawing on the work of theorist Michel Foucault, in particular his work on the nature of power and subject positions, Butler maintains that sex and gender are discursive products of modern power. Butler (1988, 1999) asks that we rethink our understanding of gender – not through the cultural meanings that are attached to any predetermined sex, but through the process that it is performatively constructed. In that sense, Butler (1999) views gender as neither essential or biologically determined, but rather created through its own enactment or performance. It is important to note that through this argument, Butler is not saying biological processes do not exist or do not affect differences in hormones or anatomy. However, what she does argue is that bodies are material and discursive and the meaning and value attached to bodies are always in the grip of culture. The dominant discursive views of sex and gender are simplistic, limiting, binary and often oppressive.

By redefining gender from a passive and natural 'state of being,' to an active and performative 'way of doing,' Butler (1999) transcends the idea that bodies are merely passive

victims to the ways and mean they are influenced and produced. While Butler (1988) acknowledges that gender is always performed within limiting and often policing framework, she describes many possibilities to destabilize, resist and reject this order. For instance, Butler (1988) notes that subversive gender performances, such as cross-dressing, drag, and parody, can be used to undermine the essentialist belief that there is a 'true gender identity', and she furthermore suggests that labels such as 'queen', 'butch', 'dyke' and 'queer' can be discursively reappropriated, which through enactment provide concrete tools for both agency, resistance and subversive action in combating both problematic or restrictive constructs of gender and the policing of gender (Butler, 1988; 1999; Fraser, 1989). In other words, the performance of drag serves to challenge and disrupt by destabilizing gender norms and illustrating the performative nature of gender.

The binary performance of gender is often further reinforced by the reactions to those who fail to adhere to gender norms. Butler (1999) says that those who fail to do their gender right, i.e. do not subscribe to traditional gender norms or somehow violate these norms, are punished. This punishment includes the oppression of women and the stigma towards those who challenge the binary, such as trans people, who in a sense disrupt the link between gender and sex, and lesbian and gay men who disrupt the link between sex and sexuality. These are abject identities, aspects of the individual or subject that are removed or cast aside and labelled "not me". What has been expelled becomes "the Other" and because of its otherness, becomes the object of the subject's repulsion. For Butler (1988; 1999), this process allows the subject to create boundaries for itself between internal and external, or between where their body stops and "the Other" begins. Butler explains that this process is present in homophobia, racism, and sexism, which involves expulsion, exclusion, and repulsion from society when certain identities

come to symbolize the Other. There is clear evidence of this stigma in the form of violence and hatred towards transwomen, particularly transwomen of color, homophobia in the form of assaults on both the rights and the bodies of queer people all over the world, and surgeries performed on intersex babies to achieve accepted or normative sex characteristics.

Butler (1988; 1999) argues that the binary performance of gender serves as tools of patriarchal power structures in society that seek to sustain and reproduce itself – with women serving as a means of reproduction to men as mothers and wives. She calls these power structures prohibitive or proscriptive in that they repress deviations in gender performance, and generative or prescriptive in how they create binary heteronormative gender performance. Butler tells us that if we consider gender identity as some inner truth versus a product of gender performance, we play into the hands of patriarchy and the notion of compulsory heterosexuality and therefore serve to reinforce the gender binary.

Learning Gender

Upon the birth of a child, the first question most commonly asked is whether the child is a boy or a girl? Here, the sex of the child denotes the physical, biological categories of "male" and "female" - a binary of contrasting glands, chromosomes, genitals, hormones, and secondary sexual characteristics. This division between what constitutes male and female, masculine and feminine, "appears fundamental to the operation of most contemporary societies, and so it seems essential that a baby is categorized, assigned to one sex or another, as early as possible" (Paetcher, 2007, p. 5). Gender socialization, therefore, begins from the moment we are born with the presumed ability to look at an infant and assign a sex. From the outset, biological sex then serves to become a central organizing principle around which children learn their gender identity.

Intersexed children – those born with atypical or ambiguous sex organs or markers – disrupt that binary- and the medical profession often pushes a choice on parents to decide which gender, and then proceed with surgeries and medical interventions to bring that infant's gender in line with a chosen sex. Although intersex children may account for 2-4% of the population, intersexuality remains highly stigmatized, with the common discourse of abnormality or unnaturalness surrounding it. However, intersexuality occurs about as often as cystic fibrosis and Down syndrome, two conditions that are more familiar to most of us and certainly cause less shame for parents and family members (Preves, 2003). Intersexuality therefore only becomes abnormal when a society subscribes to only two biological sex genders as normal and natural.

Social theories of gender focus on the impacts of broader social systems on a child's gender development (Fagot et al., 2000) through defining rules and customs about what is and is not appropriate for those boys and girls. Children's gender assignment becomes a powerful social identity that shapes children's lives,, because as soon as a child is identified as a boy or a girl, parents form expectations about the child's interests, skills, and behaviours, and these expectations appear in gendered parenting practices (Martin et al., 2002; Mesman & Groenveld, 2017). During early childhood, girls and boys look to parents and older siblings for guidance, and they provide children with their first lessons about gender in ways that can have profound impact on their gender development. This might include role modelling, gendered division of labour and care in the home, and encouraging gendered activities and interests. When parents consistently buy female stereotyped toys (e.g., dolls, tea sets) for their daughters, and malestereotyped toys (e.g., trains, dinosaurs) for their sons, they are implicitly linking their children's sex to particular gender roles that are often encouraged as the children play with these toys.

Commercial advertising aimed at children is highly gender stereotyped, as are many children's cartoons (Mesmean & Groenveld, 2017).

The way parents evaluate and provide feedback to children on their behaviour – those that are stereotyped versus those that go against stereotype – is another form of gendered parenting that can have tremendous influence. Research shows that parents are generally less likely to respond negatively to a son's risk taking behaviour, and more likely to notice and reinforce a daughter's positive prosocial behaviour, which can serve to convey and overstate a message that boys are outgoing and adventurous and girls are well behaved and kind (Marks et al., 2009; Mesmean & Groenveld, 2017; Murnen et al., 2015). There are also subtle ways that parents may reinforce gender stereotypes even when they are not overtly encouraging them. This is commonly seen in parents' use of essentialist statements about gender. Examples would be "only girls like dolls" or "only boys like football." In these instances, the parent is expressing what is known as a descriptive stereotype (i.e., describing general patterns or "essences" about each gender) rather than prescriptive stereotype (i.e., stating what should occur).

Gender-typed expectations may relate to personality traits (e.g., "boys are aggressive"), abilities (e.g., "girls are good at schoolwork"), activities (e.g., girls try out for cheerleading), and roles (e.g., "men are scientists and engineers"). As views around gender equality have increased in recent decades, research shows more variation among parents with some holding traditional expectations and some expressing egalitarian expectations for their daughters and sons (Marks et al, 2006; Murnen et al, 2015). Some may also express equal views around certain domains (such as future occupations) but are more traditional about other domains (such as family roles and responsibilities).

Of course, in contemporary Western societies, many women with children work outside of the home. Men's average involvement in childcare and housework has increased, although in most two career two parent households, domestic responsibilities are still predominantly held by women (Marks et al, 2003). Research finds that fathers' active engagement in the family as a coparent, with equal sharing of roles and responsibilities and modelling of positive relationships between spouses, can have a significant positive influence by demonstrating an adult male in a nurturing, caring role (Buswell et al., 2018). This can be highly influential to male sons, particularly those who identify more with their same sex parent as it can motivate them to start imitating some of those same interests.

Although the literature is limited, research on LGBTQI2S+ families indicate that, when compared to children raised in two-parent heterosexual families, children raised by same-gender parents are less likely to endorse certain gender stereotypes and are therefore more tolerant of children's varied choices of toys or types of play. However, when same-gender parents divided labour with one parent as primary caregiver and the other parent as the primary breadwinner, their children were likely to express stereotyped views similar to children in opposite sex families. Biblarz and Stacey (2010) concluded that single-sex parenting (i.e., single-parent, gay and lesbian parents) employ different socialization practices, they are also models for nontraditional gender roles to their children. Single parents' behaviour indeed is often less traditional, because these parents have to fulfill both roles of economic provider and caretaker. The same is true for gay and lesbian parents, who are more likely to share the roles of caretaker and economic provider (Solomon et al., 2005; Stacey & Biblarz, 2001).

Gender socialization works, according to social learning theory, by rewarding or challenging and punishing children for engaging in sex-typed behaviour that is consistent with

their assigned sex category. A strong example of this is crying - while a little girl may be soothed when she cries, a little boy may be told that boys don't cry. Crying therefore becomes a sextyped behaviour - mostly acceptable for girls and therefore not a punishable behaviour, but widely not seen as an appropriate behaviour for boys, and so the little boy may be punished, shamed, or corrected for his crying behaviour. Through these kinds of interactions – some subtle and some more explicit - gender socialization occurs. Gendered stereotypes are modelled and communicated through children's dress, their play, and even their household chores from an early age (Kilmartin et al., 2015). In fact, this begins from the moment of birth or even earlier—from the colour of a child's bedroom to the clothes they are presented. Children are provided gender-typed toys such as dolls for girls and trucks for boys, and often receive a more positive response and attention from their parents for choosing those to play with, which in turn can be reinforcing. In contrast, boys are often discouraged from playing with dolls or playing dress-up, while girls are often prevented from engaging in risky sports or activities (Halpern & Perry-Jenkins, 2015; Kilmartin et al., 2015; Stainton Rogers & Stainton Rogers, 2001).

Developmental psychology often views early play and family activity as a rehearsal for later social roles (Kilmartin et al., 2015). As a result, from a very early age, children can demonstrate highly gender stereotypical beliefs that are dominant within their families, their communities, and their culture (Halpern & Perry-Jenkins, 2015). Like their parents and caregivers, young children learn to essentialize gender in ways that make assumptions about what constitutes meaning as male and female based on biological sex (Halpern & Perry-Jenkins, 2015; Kilmartin et al., 2015; Stainton Rogers & Stainton Rogers, 2001). Comparisons of fathers' and mothers' gender typing reveal that fathers, in particular, tend to be more stereotypical in their definition of gender-appropriate activities, especially regarding their sons.

A major criticism of social learning theory is its essentialist application, as it is limiting in the sense of describing behaviour solely in terms of either nature (our genetic makeup) or nurture (our environment) rather than a combination of both (Addis et al., 2010). Studies also show that children do not always model same sex parent behaviour. Children raised in single parent families do not always solely identify with their parent's sex or gender. The influence of children on their environment is not always considered in social learning theory, and they are largely seen as passive recipients, which ignores the active role that they play in their socialization process.

Gender norms are shaped by a broad set of cultural influences and can vary across cultures. In developing countries, many parents have a strong preference for sons based on what they might provide for the family, and as a result, access to resources ranging from health care to education are prioritized to sons over daughters. The United Nations has estimated that female infants and children suffer higher rates of abuse and neglect in countries such as China, India, and Papua New Guinea related directly to the preference for male children. Historically, female infants survived at much lower rates than male infants because parents generally fed the girls less and neglected their basic needs (United Nations, 2000).

In many cultures, the family name can only be carried on by sons because daughters take their husband's name upon marriage. In Hindu religious traditions, rituals surrounding the death of a parent are entrusted solely to sons. While the reasons for this preference for boys vary, they have important impacts on the gender socialization of boys and girls. "Good wife, wise mother" is a Japanese proverb that continues to have strong influence on Japanese culture with its messaging to women that their role is to stay at home and be devoted entirely to the family's needs. These examples, say researchers, lie in stark contrast in the differential treatment of sons

and daughters in Western culture (Bisin & Verdier, 2010; Halpern & Perry-Jenkins, 2015). I would argue, however, on the virtue of being born male, that men have access to many rights, advantages, and freedoms available solely to them. When considering cultural differences, researchers also failed to fully take into account how meaningful traditions or customs might be meaningful to those who enact them, and thereby strip women and men of agency by reducing their actions to the product of their socialization. Also important is the recognition that gender roles are culturally and historically situated, meaning that they cannot be analyzed or interpreted outside of the cultural and historical context of their time.

Nonetheless, there are common ways that parents in these societies may socialize girls and boys differently. Research shows that explicit messages to children (e.g., dolls are for girls) are relatively less common today in most Western societies that value gender equality (Mesman and Groenveld, 2017). There has always been variations in what constitutes family that challenge a Western heteronormative discourse around notions of two parent, opposite sex, male and female led families, and the traditional image of the two-parent heterosexual family with the father serving as the provider and the mother as the homemaker is no longer the accepted norm in many industrialized countries. Instead, many women pursue jobs outside of the home and many men are involved to varying degrees in childcare. In addition, many children are raised by single parents and by lesbian/gay parents. Despite these role changes, studies show that there are relatively few truly equal parenting arrangements in terms of responsibilities in raising children. Also, studies suggest that parents who espouse gender equal attitudes may still act differently with daughters and sons in terms of roles, expectations and freedoms (Gelman et al., 2004).

Gendered lives begin even before we are born because the societies that we enter into have already made decisions about how our sex and gender will be determined. In addition, the

content of the cultures into which we are born will partially dictate exactly what gender lessons we learn. Families surround us with explicit and implicit messages about what it means to be a gendered person. Even those whose biological makeup does not fit within established sex categories are expected to conform to gender expectations. Gender is all around us, and it is therefore not surprising how it quickly becomes internalized by infants and children.

Raising Boys to Men

Because gender is a social category that organizes virtually every segment of society, there are multiple sources of socialization in children's gender development aside from parents and immediate caregivers. Siblings, other family members, peer groups, friends, the media, coaches, group leaders, and teachers all serve to influence gender-related behaviours (Jaffee, et al., 2003; Lippa, 2005; Paetcher, 2007). As children grow into adolescence, the impact of peers and social media become even more profound.

Research shows teachers often respond differently to boys and girls when addressing unwanted or undesirable behaviour, sometimes even when they demonstrate the same behaviour (Jaffee et al., 2003; Lippa, 2005; Paetcher, 2007; Stainton Rogers & Stainton Rogers, 2001).

Peer influences become especially important and influential as children grow into adolescence and same-sex segregation can sometimes intensify differences in boys and girls (Lippa, 2005; Paetcher, 2007) reinforced culturally through school and community (Kilmartin, 2015; Lippa, 2005). For example, boys tend to police one another through encouraging masculine behaviour (running, climbing) and ridicule perceived female behaviour (singing, dancing). Sex segregation, which studies show may start as early as the second or third year of life and continue to intensify as children grow older, primarily begin due to family and peer influence, and later reinforced in schools and communities. A hypothesis as to why children themselves choose to engage in sex-

segregated play is that males in general are more aggressive, competitive and group-oriented than females, while girls consider boys more domineering and unresponsive to verbal requests or the possibilities of negotiations (Lippa, 2005; Stainton Rogers & Stainton Rogers, 2001).

Popular culture and mass media continue to be saturated with gender stereotypical behaviours, and children can actively and readily learn expectations about gender from merely observing or being exposed to these influences through film, music, and literature (Jaffee, et al., 2003; Lippa, 2005; Paetcher, 2007). For preschoolers, characters on television or in books exemplify obvious masculine and feminine appearances – a superhero is adventurous and has huge muscles, a princess is beautiful and has long flowing hair. These representations translate to character traits – the superhero is brave and heroic while the princess is fearful and in need of saving. A few years later and these same characters might be interpreted as dominant and aggressive and as submissive and weak, which can be problematic in its messaging. And an important contributing factor to this is that they are often socialized to react and respond to these characterizations and by extension one another in this way.

Research shows that differences in how we raise and socialize young boys and girls contribute to teaching girls about the salience of feelings in early stages of language acquisition, and teaching boys to focus more on the action of doing rather than on the feeling of doing (Jaffee, et al., 2003; Lippa, 2005; Paetcher, 2007; Stainton Rogers & Stainton Rogers, 2001). Collectively, a lack of emphasis on internalized feelings combined with a focus on activities that are outside of the home contribute to definitions of masculinity that serve to emphasize the external, such as material possessions, athletic performance, status and influence over others (Jaffee, et al., 2003; Lippa, 2005; Paetcher, 2007). In contrast, girls are often encouraged, both subtly and explicitly, to internalize their thoughts and feelings and to be more considerate and

demonstrate a heightened awareness of others' perspectives (Jaffee et al., 2003; Lippa, 2005; Paetcher, 2007; Stainton Rogers & Stainton Rogers, 2001).

The literature demonstrates how traditional masculinity ideology influences the socialization of children and the behaviour of adults through social interactions that involves observational learning, punishment and reward (Berger et al., 2005; Levant & Richmond, 2007). The process of toughening boys consists of verbal bullying, calling them sissies and equating their behaviours, responses and actions as something womanly, which in turn serves to separate them from the masculine ideals to which they are expected to aspire (Kimmel, 2012). Words can take on a discursive power and do damage and harm as surely as physical actions (Brown, 2013), cutting a young person's self-esteem, poisoning their self-confidence, and undermining their self-image. This can in turn encourage boys and men to conform to the prevailing male role norms by adopting certain socially sanctioned masculine behaviours and avoiding others, with one very central masculine norm the restriction of emotional expression (Levant, 2011).

However, although a lack of expressive emotion tends to be a defining characteristic of hegemonic masculinity discourse, critics note that throughout the world there is a great deal of cultural variation in the gendered expression of emotion, with some cultures considering women to be the emotional sex, some allowing a wide latitude and range of expressiveness in both sexes, and some expecting both men and women to police, constrain, and regulate their emotions (Kilmartin et al., 2013; Kimmel, 2012). It is crucial to recognize that although cultures influence people and exert normative pressure on people to think, feel and act in specific ways, responses to cultural expectations can vary widely. As a result, even in cultures that expect women to be highly emotional and men less so, there are very expressive men and very inexpressive women.

Gender awareness for many men involves "not only their understanding of the pressures of traditional masculinity but also an appreciation of their advantaged social position" (Kilmartin & Smiler, 2015, p. 19). However, the privilege of masculinity does not fall to all males in equal share or measure, and certainly not to marginalized men who reflect differences in race, culture, class, or sexual orientation. The opportunity to reap the social benefits of being "a real man" is, in fact, available to relatively few. Social constructs of masculinity can exert pressure on all men to experience themselves and the world in specific ways, but these demands intersect and interact in significant ways with the oppression and marginalization of racism, classism, and heterosexism (Kilmartin & Smiler, 2015; Kimmel, 2008). For instance, dominant masculine ideology can serve to impose extreme self-reliance and reflects that quality as some part of an expected order of things. It can also impose assumptions and expectations around power, wealth, and success that for various reasons – ability, opportunity, and ambition - may not be in their grasp (Kia-Keating et al., 2005; Kimmel, 2012).

bell hooks (2005) once said that in order "to indoctrinate boys into the rules of patriarchy, we force them to feel pain and to deny their feelings" (p. 22). Hegemonic masculinity equates stoicism with strength, even when in the face of danger, where it quickly becomes imperative to either dismiss or not acknowledge feelings of fear, uncertainty, or helplessness (Kia-Keating et al, 2005). A consistent argument lacking in much of the literature related to gender socialization of boys and their experiences with trauma is that it is that it ignores or pays little attention to this conceptualization of pain and indoctrination through denial that hooks (2005) so eloquently describes, something that can have profound influence and impacts on male sexual trauma survivors. Therefore, a significant obstacle for male survivors and a likely necessary one to overcome as part of healing includes an ability to set aside this expected stoicism and find ways

and means to reconnect with others, to not depend on self-reliance and instead discover a means of reaching out and asking for help and guidance, and a develop tools and supports so they can acknowledge and accept their feelings of shame, sadness, anger, and vulnerability as they relate to their traumatic histories.

Finding Foucault

Michel Foucault, the French philosopher, has been hugely influential in shaping our understandings of power, turning away from the notion of power belonging to the few, who wield it an instrument of coercion, and toward the idea that power exists everywhere, dispersed and embodied through discourse, knowledge and regimes of truth (Foucault 1991). Foucault (1991) argued that knowledge and power are intimately connected, creating the term "power/knowledge" to signify that one is not separate from the other. Power/knowledge is constituted in society, which is constantly in flux and negotiation, through accepted forms of knowing, understanding, and what is accepted and functions as true.

According to Foucault, each society has its own regime of truth: the types of discourse which it accepts and makes function as true; the mechanisms and instances which enable one to distinguish true and false statements, the means by which each is sanctioned; the techniques and procedures accorded value in the acquisition of truth; and the status of those who are charged with saying what counts as true' (Foucault, 1991). Regimes of truth are the result of discourse that is reinforced, renegotiated, and redefined constantly throughout society, by way of media, educational institutions, healthcare, and justice systems, as well as through our political and economic ideologies. Therefore, there is no absolute truth that can be discovered and accepted, but there does exist a battle about the status of truth and the economic and political role it plays (Foucault, 1991). Foucault (1991) believed that power is not just a negative, coercive or

repressive thing that forces us to do things against our wishes, but that it can also serve as a necessary, productive and positive force in society:

We must cease once and for all to describe the effects of power in negative terms: it 'excludes', it 'represses', it 'censors', it 'abstracts', it 'masks', it 'conceals'. In fact power produces; it produces reality; it produces domains of objects and rituals of truth. The individual and the knowledge that may be gained of him belong to this production (p.194).

Foucault wrote about issues to do with mental health in several of his works, such as the history of psychiatric practices, psychoanalysis, sexuality, and the formation of identity and subjectivity. His work on the human sciences focused on the rise of professional power throughout the eighteenth and nineteenth centuries, mainly through the development of medical discourse (1990, 1991). Foucault recognized how power is also a major source of social discipline and conformity and pointed to schools, mental hospitals/asylums and prisons in eighteenth century Europe as systems of surveillance and assessment that no longer required force or violence because people had learned to discipline themselves and behave in expected way. Foucault was interested in the mechanisms of prison surveillance, school discipline, population controls, and the promotion of norms about bodily conduct, including sex. He studied psychology, medicine and criminology as bodies of knowledge that define norms of behaviour and deviance.

In critiquing hospitals, asylums, prisons, and schools, sexuality and science, Foucault's underlying themes showed a continuing concern with the relationship between knowledge and power (Bracken et al., 2007). In examining psychiatric discourses, Foucault "historicizes' the concept of mental illness' by showing how it emerged at a certain historical point within a culture and society, and 'politicizes' that concept by showing how its emergence was

inextricably bound to the political concerns, norms and values of that culture and society" (Roberts, 2005, p. 37).

Foucault's work centered on how some of our most deeply held beliefs and assumptions about ourselves and our societies have been developed over time, and how the human sciences have become a constitutive element in the formation of power-relations in society (Bracken et al., 2007; Randall & Munro, 2010). Through a Foucauldian analysis, psychotherapy may be viewed as a manifestation of one of the most pervasive examples of the power-knowledge relation exists within western societies, that of 'the confession' (Foucault, 1990). Foucault believed that the individual's experience with mental illness is "pathologized through the exercise of the psychiatrist's power to create knowledge about it" (Swerdfager, 2016, p. 291). Thoughts, feelings and behaviours of clients are interpreted based on current psychiatric discourses that categorize specific thoughts, feelings and behaviours as mental illness (Roberts 2005). However, Foucault saw this power as only somewhat localized in the psychiatrist, because he believed it could not be entirely distilled into one body but would flow instead through many points' that make up the psychiatric apparatus of mental health services (Foucault, 1990; Swerdfager, 2016). For Foucault, psychotherapy is not an "emancipatory alternative to psychiatric hospitals and drug treatments. Rather, it incorporates a disciplinary power that is productive but also controlling in its own way" (Bracken et al., 2007, p. 608).

Key to Foucault's approach to power is that it transcends politics and views power as an everyday, socialized and embodied phenomenon. His work has been hugely influential in illustrating how norms can be so embedded they are beyond our perception – causing us to survey and discipline ourselves without any willful coercion or force from others.

Foucault's interest lies in how people are both made into and turn themselves into subjects within our culture (Brown, 2007; Foucault, 1991). Foucault rejects a negative representation of power while suggesting it is the "widely accepted view because it, in fact, masks and hides the complexity and insidiousness of how power operates" (Brown, 2007, p. 17). By criticizing the classical view of power as solely repressive with the focus on the powerful and the powerless, Foucault (1980) viewed power as constitutive or productive while still recognizing its constraining and oppressive traits, (Brown, 2007). However, Foucault believed in possibilities for action and resistance. He was an active social and political commentator concerned with our capacities to recognize and question socialized norms and constraints. To challenge power is not a matter of seeking some 'absolute truth' (which is in any case a socially produced power) but, "of detaching the power of truth from the forms of hegemony, social, economic, and cultural, within which it operates at the present time" (Foucault, 1991, p. 75).

Foucault challenges us to not only reconceptualise our understanding of power but also the role that we, as individuals, might play in those power relations. As Brown (2007) notes, Foucault privileges how power manifests, its techniques and strategies, versus who wields it: "a repressive approach to power is only ever toxic; there is no room for productive power' (p. 15). Foucault (1980) situates the human experience in a particular power-knowledge dynamic that is always in a state of flux, and which in turn defines practices, customs, and discourses through which we become subjects. Foucault (1980) did not believe in a fixed or essential human subjectivity. He argued that "we should be trying to discover how multiple bodies, forces, energies, matters, desires, thoughts, and so on are gradually, progressively, actually and materially constituted as subjects, or as the subject." (Foucault, 1980, p. 28).

Foucault defined technologies of the self as techniques that:

permit individuals to effect by their own means, or with the help of others, a certain number of operations on their own bodies and souls, thoughts, conduct, and way of being, so as to transform themselves in order to attain a certain state of happiness, purity, wisdom, perfection, or immortality" (p. 18).

Defining characteristics such as age, class, sexual orientation, and religion are socially constructed through technologies of self and these constructions converge in a variety of different ways to influence identities. The concept of self is then constructed and deconstructed through changing discursive practices and circumstance, limited by those available to us and by those we choose to occupy or embody.

Foucault describes individuals as vehicles of social power by the way they reify and enact prescriptive and normative practices of self (Brown, 2007). Power becomes insidious because, while individuals might see themselves as enacting their true inherent selves they are enacting or resisting socially constructed discourse of what constitutes the self (Brown & Augusta-Scott, 2007; White & Epston, 1990). Therefore, they often become willing participants in ensuring conformity to normalizing practices of self (Brown, 2007; Foucault, 1991). This is something Foucault (1990) referred as disciplining the docile bodies, the processes and practices through which individuals engage in normalizing, self-monitoring, and disciplinary practices of the self (Foucault, 1980, 1991; Brown & Augusta-Scott, 2007). Foucault (1980) conceptualized the 'normalizing society" by considering how a society composed of individuals becomes trained, modified, and categorized within a system of expected norms (Foucault, 1980; 1991). He then analyzed how these norms and practices reflected and influenced social institutions, which included investigating how discourse and knowledge are produced, reproduced, and then ultimately serve to change our realities (Foucault 1980).

According to Foucault (1980), power is also a major source of social discipline and conformity. Foucault describes a new kind of 'disciplinary power' that could be observed in the administrative systems and social services that were created in 18th century Europe, such as prisons, schools and mental hospitals. He noted how their systems of surveillance and assessment did not require force or violence to conform, as people learned to monitor and discipline themselves and behave in expected ways. As Brown (2007) suggests:

Through a process of subjectification, we turn ourselves into subjects, absorbed by the creation, improvement, management and performance of self. The process turs our gaze inward, as we seek to cultivate and manage ourselves as individuals and renders invisible the social and cultural context in which the subject is made. In anticipating being seen and judged by others, we often shape ourselves accordingly (p. 109).

By making ourselves into subjects we are both constrained by and resistant of power and its effects. The performance of self in culture and society involves agency, self-determination, limitation and social constraint (Brown, 2007).

A Foucauldian analysis of the discourse of masculinity and the notion of technologies to self-illustrate how men might give voice or use silence to turn themselves into subjects. In other words, men may give voice to or embody specific positions and be silent on other subjects, which in turn can contribute to the discursive patterns and applications of power within masculinity. Foucault's (1980) understanding of the process of subjectification as a means of resisting, disrupting, interpreting, and at times transforming dominant discourses of masculinity and what it "means to be a man" may be useful in this study as it puts focus on survivors as active participants in policing their own and others performance of gender.

Brown (2007) notes that by adopting a Foucauldian stance on power, we are able to move beyond the idea that one either possesses or does not possess power, and that this allows us the means to interpret stories as evidence of both social agency and social constraint. "Refusing to conceptualize power and identity as fixed or static produces new possibilities and thus the possibility of new stories" (Brown, 2007, p. 15). These possibilities, along with knowledge of both agency and constraint, become meaningful in our work with trauma survivors by offering that possibility of a new or different narrative and a new or different way of understanding trauma experiences.

Hegemonic/Dominant Masculinity

Our current understanding of hegemonic or dominant masculinity was conceptualized in the mid-1980s through the work of R.W. Connell (1995) as a specific form of masculinity relative to historical and societal settings. Hegemonic/dominant masculinity legitimates men's dominant position in society and therefore legitimizes unequal gender relations between men and women, between masculinity and femininity, and among other types of masculinity, such as those subordinated (LGBTQI2S + persons) and those marginalized (men who are racialized, culturally diverse, or hold lower socio-economic status). Hegemonic masculinity is distinguished from other forms of masculinities, especially subordinated ones, and although not assumed to be normal, it *is* viewed as normative through its enactment, while embodying a current time-honoured and traditional way of being a man, requiring other men to position themselves in relation to it (Connell & Messerschmidt, 2005; Courtenay, 2000). Certain traditional guiding beliefs about hegemonic masculinity include: men and women are inherently different; men are superior to women and superior to men who do not live up to certain prescribed ideals of manhood; activities customarily associated with women are demeaning for men to be engaged

in; men should not express vulnerability or sensitive emotions; the most typical manly emotions are desire and anger; toughness and domination are essential to man's identity; sex is less about pleasure and more about proving manhood and asserting power; and gay or homosexual men are failed men who do not subscribe to masculine norms (Connell, 1995; Fisher & Goodwin, 2008; Kimmel, 2008; Kivel, 2010).

Intersectionality is a term coined by feminist scholar Kimberle Crenshaw (1993) to denote the ways in which race and gender interact and intersect both simultaneously and inseparably, and contribute to marginalization and oppression of racialized women. Intersectionality provides a framework and an analytical lens to examine how each individual experiences social structures differently because of the intersection of identity markers such as gender, race, age, class, and ability and how that contributes to their privilege or their disadvantage. Race, ethnicity, class, sexuality, age, and ability shape the enactment of masculinity, as well as the social contexts in which these performances take place. These inequalities also shape the extent to which men benefit from gender inequality—what Connell (1995) refers to as the "patriarchal dividend" (p. 79). Because individuals share multiple social statuses and claim multiple identities, performances of masculinity are interconnected with performances of race, ethnicity, class, age, and other social identities (Collins 2004; Pascoe 2007). Expectations of masculinity and masculine performance are therefore shaped by the institutional structures and broader social contexts in which they are situated. However, among masculinities lies a hierarchy in which men and masculinity are privileged over women and femininity, but also as a hierarchy within which some men, and some versions of masculinity, are privileged over others (Connell 1995; Connell & Messerschmidt 2005). The more these identifies intersect the more marginalized or oppressed the individual might be – for example, a poor, black gay male might face overlapping marginalization due to his race and sexual orientation. Race, ethnicity, class, sexuality and other identities therefore shape not only the ways in which men enact masculinity, and the social contexts within which these performances take place, but also the extent to which men are able to benefit from systems of gender inequality.

Masculine scripts refer to internalized gender norms that serve as ideals of acceptable ways for boys and men to think, feel, and behave (Mahalik et al., 2003). Masculine studies scholars such as Richard Gartner, Michael Kimmel and Jackson Katz refer to these scripts often throughout their work. Psychologist Robert Brannon (1976) introduced four phrases to describe masculinity, phrases that encapsulate what is commonly referred to as masculine scripts.

Brannon's four phrases included: 1) "No Sissy Stuff!" which translates to never engaging in any activity or behaviour that even remotely suggests femininity. Masculinity, therefore, is the relentless repudiation of the feminine. 2.) "Be a Big Wheel", meaning that masculinity is measured by power, success, wealth, and status. 3. "Be a Sturdy Oak." with masculinity presenting as a calm and reliable demeanor at times of crisis, partly through holding emotions in check. Proving you are a man means either not showing or minimizing your emotions.4. "Give 'em Hell." which calls for an aura of manly daring and aggression, and the willingness to always take risks (Brannon, 1976; Kimmel, 2012).

These rules contained succinct elements of the definition of masculinity upon which men are measured. They are restrictive and repressive, and I do not believe that most men subscribe to these notions. However, a young male impacted by trauma that forces him to question issues around masculine identity and sexual orientation can become entrenched in a belief system that tells him he has failed at masculinity due to his experience of victimization. Failure to embody

these rules, therefore, becomes a source of pain and confusion for many young men and boys, even though their achievement as a whole is unrealistic if not impossible. But masculinity can serve as a relentless test, and first among it – regardless for the most part of race, sexual orientation or class – is the call to *not* be like a woman.

Within any culture, there can exist a hierarchy of masculinities that are compared with a dominant or hegemonic ideal (Connell, 1995). In Western culture, the normative form of hegemonic/dominant masculinity is defined by race (white), sexual orientation (heterosexual), socioeconomic status (middle class) and the possession of certain traits such as assertiveness, dominance, control, physical strength, and emotional restraint (Courtenay, 2000; Kimmel, 2008). Men who subscribe to these standards are often those who have the specific capital or resources necessary to perform hegemonic/dominant masculinity, such as the physical prowess and capabilities to succeed at competitive sports or win physical fights, or economic capital that allows for financial independence and security or access to higher education (Coles, 2007). Kimmel (2008) notes that the definition of hegemonic masculinity is central to:

a man in power, a man with power, and a man of power. We equate manhood with being successful, capable, reliable, in control. The very definitions of manhood we have developed in our culture maintain the power that some men have over other men and that men have over women (p. 125).

In expanding the definition of hegemonic masculinity to encompass one of *masculinities*, Connell (1995) describes complicit masculinity as men who may benefit from hegemonic masculinity but do not explicitly enact it, while subordinated masculinity describes men – primarily LGBTQI2S+ or gender diverse men - who are oppressed by definitions of hegemonic masculinity. Marginalized masculinity describes men who may potentially possess power in

terms of their gender, but not in terms of their class, race or culture, and are therefore subject to discrimination or marginalization (Connell 1995; Connell & Messerschmidt, 2005; Pascoe, 2007). As noted earlier, Connell and Messerschmidt (2005) emphasize that these configurations of gender are often contextual, and relatively few men are hegemonically masculine, but most men do benefit, to different extents and varying degrees, from this guiding definition of dominant masculinity in society. For example, relative privileges by virtue of being a white, heterosexual male in contemporary society include, among other things, the ability to go out in public without fear of being followed or harassed; the option of seeing people who look and act like them positively portrayed in the media; and the opportunity of having people of the male sex disproportionately represented in government, education, and other institutional settings (Kilmartin & Smiler, 2015, Kimmel, 2015). Men can distance themselves or engage directly with hegemonic masculinities based on their interactional needs, and therefore masculinity does not represent a certain type of man but the way that men position themselves through discursive practices (Messerschmidt, 2019). As a result, masculinity does not mean the same thing to all men, but instead it is varied in how is interpreted, experienced, and lived out (Coles, 2007). Therefore, ways of being masculine are contextual, open to reinterpretation and renegotiation and connected to historical change.

Men who are more marginalized in society due to socio-economic factors, cultural differences, or race are often denied access to the resources and social power needed to enact such standards of hegemonic/dominant masculinity which in turn serve to validate or legitimize their sense of manhood – they are therefore seen as *less than* by those men that are more privileged (Courtenay, 2000). Factors such as class, race, gender, ability, and sexual orientation contribute to marginalization, and viewing male privilege through an intersectional lens (i.e. how

those factors intersect within the context of one's life) helps in our understanding of the complexities of male privilege. Given the dictates of hegemonic/dominant masculinity, physical and verbal dominance and at times violence become more "readily accessible resources for structuring, negotiating, and sustaining masculinities" (Courtenay, 2000, p. 1391). Research shows patterns of more frequent use of overtly coercive behaviour and incidents of intimate partner violence involving verbal and psychological abuse and physical force among poor and working-class men (Schrock & Schwabe, 2009). Other studies illustrate how working-class men engage in "bar culture" by demonstrating their masculinity through heavy drinking, over sexualization of women and aggressive posturing (Eastman, 2017).

While gender socialization is often the central focus in studies of masculinity, it is important to pay close attention and distinguish how cultural expectations intersect with the ways in which society perpetuates masculine norms (Kia-Keating et al., 2005; Kimmel, 2012; Levant, 2011). For instance, many Latino and African American cultures often emphasize the most traditional views of masculinity and maintain rigid standards (Kia- Keating et al., 2005; Levant, 2011), and therefore these already marginalized young men and boys are positioned to face even higher pressures to conform to societal norms. However, while marginalized men of colour might accept that there are culturally dominant masculine ideals, their everyday experiences are not necessarily experienced as feeling subordinate, and rather they might construct and perform their masculinities as dominant in relation to other men's masculinities and challenging to standard norms of hegemonic/dominant masculinity (Coles, 2007). Racialized men might therefore negotiate masculinity by drawing on those aspects of hegemonic masculinity for which they have the capacity to perform or even excel and establish their own standards and meanings of masculinity (Coles, 2007; Griffith, 2018). Despite being located "in a subordinate position in

the field of masculinity, these men's lived experiences of masculinity are not of being marginalized or subordinated, but of being legitimate and dominant" (Coles, 2007, p. 246).

Research has demonstrated the role of masculinity in aggression toward those who do not conform to strict gender narratives, leading to violent and often fatal attacks against transgender and gender diverse people (Gruenwald & Kelly, 2014), something that seems particularly relevant in the recent epidemic of hate crimes against trans women of color in the United States. Traditional masculinity ideology excludes LGBTQI2S+ males because they violate a fundamental criteria for being considered masculine due to their sexual attraction to or engagement with other men (Sanchez et al., 2010). As a result, gay males who value or subscribe to traditional masculine norms "may experience stress, shame, or guilt because being truly 'masculine' is unattainable due to their same-sex romantic attractions" (Sanchez et al., 2010, p. 82). At the same time, other sexual diverse and gender diverse individuals may not correlate traditional masculine norms as an essential component of their male gender identity and view masculinity on a spectrum or continuum without experiencing a need or pressure to conform (Bockting et al., 2009).

The way we construct masculinities is often by default – to be masculine is to be the opposite of feminine - and societal discourse suggests that conversations around gender more appropriately guides the conversation for women and girls but not for men and boys. Some have critiqued discourse around hegemonic masculinity as static and unchanging, questioning how men "conform to an ideal and turn themselves into complicit types without anyone ever managing to exactly embody that ideal" (Wetherell & Edley, 1999, p. 337).

Hegemonic/dominant masculinity and its' characteristics are increasingly being viewed from different and varied perspectives. Nevertheless, what we understand as femininity has and

continues to change in the sense that women actively continue to renegotiate, challenge, and redefine it, while masculinity appears to be much more resistant to change (Kilmartin et al., 2015; Kimmel, 2012; Lisak, 1994). This resistance can potentially result in more psychological distress and dysfunction for men and boys and leading to challenges to the way masculinity is constructed, understood, and enacted.

Toxic Masculinity

The term "toxic masculinity" is generally used to describe hegemonic masculinity at its most extreme form. It denotes how some aspects of masculinity such as entitlement, misogyny and homophobia can harm women, children, families, and can impact men's health outcomes (Fisher et al., 2008, Kupers, 2005). Toxic masculinity has been linked to acts of violence in Western cultures such as mass shootings, demonstrations of racial violence and discord, attacks on trans women of color, and campus sexual assault (Haider, 2016; Katz, 2006; Kimmel, 2012; Kupers, 2005). As noted earlier, it is a term that has become highly politicized and polarizing in society and culture, often portrayed and perpetuated in ways that construct masculinity around themes of domination, control, and violence (Katz, 2006; Kimmel, 2012; Kimmel, 2015).

Although growing in popularity over the past few years, the term toxic masculinity was first coined during the short-lived mythopoetic men's movement of the 1980s and 1990s, influenced by the writings of Robert Bly, and motivated in part as a reaction to second-wave feminist thinking. This movement – consisting of male only workshops, wilderness retreats and drumming circles – was designed to rescue masculinity from a society that supposedly feminized boys, supposedly by separating them from their fathers and having them learn what constitutes masculinity from their mothers. Followers reasoned that this denied boys access to rites and rituals to realize and achieve their true inherent masculine identity (Messner, 2004; Salter, 2019).

Ironically, the movement defined what they labelled the feminization of men as toxic masculinity, and blamed society, and by extension women and in particular mothers, for denying men and boys access to their inner selves, which they related to archetypes such as king, warrior, and wild man (Messner, 2004). Their overtly sexist ideology was problematic, particularly in their essentialist ideas of what constituted gender and their lack of analysis of patriarchal institutions and power structures that particularly favoured men.

Through various research studies, interpretations of harmful forms of masculine behaviour, what I describe here as toxic masculine culture, have been closely linked with early alcohol and drug use, delinquency, early sexual intercourse, aggression after drinking, and high risk driving (Edwards et al., 2014; Fisher et al., 2008; Katz, 2006; Kimmel, 2012). Toxic masculinity has also been strongly correlated with personality traits such as high impulsivity and lack of empathy, and with attitudes such as rape myth acceptance, homophobic and transphobic ideas, and misogynistic beliefs (Edwards et al., 2014; Katz, 2006; Kimmel, 2008; Kimmel, 2012). In addition to these noted correlational studies, empirical support for the relationship between toxic masculinity and sexual assault has also been consistently found, with the notion that extreme forms of dominant masculinity "can be thought of as the common thread which binds together the fundamental constructs of coerciveness against women" (Peters et al., 2007, p. 179). Furthermore, certain men who ascribe to this form of masculinity, in an effort to actualize more power, may partake in the use of violence and control to subordinate others, and engage in extreme competition and aggressive athletics and risk-taking behaviours as the true measures of an authentic man (Katz, 2006; Kimmel, 2012; Kivel, 2011).

Today, dominant discourses in contemporary Western societies at times share a similar toxic construct of masculinity, wherein white, middle class heterosexual males specifically might

aspire to reach an unrealistic and at times unhealthy stereotype of manhood, because they believe by virtue of their gender, race, sexual orientation and class they are privileged to do so (Katz, 2006; Kimmel, 2015; Lisak, 2017). The elusiveness of this enactment of manhood means that no man can ever feel totally and permanently confident that he has truly made the masculine grade, and living up to an idealized image of manhood, it seems, becomes almost a virtual impossibility (Kaufman, 2002; Kimmel, 2015). A further rationale behind that 'failed men' assertion is that since masculinity is contingent upon a solely heterosexual orientation, a man is immediately considered substandard should he desire individuals of the same sex (Benke, 1997; Lisak, 1994).

Studies have shown that men who subscribe to these toxic or extreme traditional notions of masculinity are significantly more likely to engage in high-risk behaviour (Courtenay, 2000; Fisher et al., 2008; Katz, 2006; Kimmel, 2012). These studies also find that racialized men of colour are more likely to hold more traditional ideas of masculinity that are reaffirmed by their culture and community (Courtenay, 2000; Fisher et al., 2008; Katz, 2006; Kimmel, 2012). If heterosexual, middle class, white, able bodied men are the signifiers of dominant masculinity, then gay, working class, non-white disabled men fail to measure up or conform. These men may come to struggle and see themselves as powerless due to systemic prejudice and racism in certain contemporary societies, which means they may therefore be more likely to try to assert their manhood through thrill-seeking or risky behaviours (Courtenay 2000). In that sense, such risk-taking behaviours arguably provides a way for marginalized males to prove themselves as men, since they lack other more efficient means and symbols to demonstrate power or authority.

In the United States, men suffer more severe chronic conditions than women, have a life expectancy that is seven years shorter than that of women, and experience higher death rates for all leading causes of death, including suicide, heart disease, cancer and accidental injury

(Courtenay, 2000). Masculinity for some men might mean going to work when sick, refusing to seek medical help for a chronic issue, driving when sleepy or intoxicated, refusing to wear safety equipment, engaging in unsafe sexual behaviour, or ignoring risks to their health associated with tobacco use or an unhealthy diet (Mahalik, Walker, and Levi-Minzi, 2007). Researchers believe that typically when boys are socialized to conform to a rigid adherence to traditional masculine norms, they experience harmful physical and mental health outcomes (Courtenay, 2000; O'Neil, 2008, Pleck, 1995).

Although the majority of young men may and boys may not subscribe to misogynistic, anti-feminine or sexist beliefs, for some sexism may become deeply entrenched in how they understand and construct masculinity ((McDermott & Schwartz, 2013; O'Neil, 2015). From an early age, many young boys are gender policed and may experience negative consequences for violating prescribed masculine gender norms (Reigeluth & Addis, 2016). For example, parents of young children might hold essentialist beliefs that dolls are for girls, or only boys play hockey, and impose their beliefs on the son that loves Barbie and the daughter that wishes to try out for the school hockey team.

Sexual minority, transgender and gender-diverse persons may be seen as transgressing dominant masculinity by not adhering to gender norms. As noted earlier, a great deal of research has detailed the role of masculinity in homophobic and transphobic violence and aggression (both verbal and physical) against those who do not conform to strict gender rules or narratives. On a daily basis, media coverage details horrific violent assaults and often fatal hate crimes against transgender and gender-nonconforming people around the world, with threats and challenges to the perpetrator's masculinity often the root cause (Gruenwald & Kelly, 2014; Kimmel, 2015). 'Toxic masculinity' takes away responsibility of men for their engagements in

violence against LGBTQI2S+ peoples and against women, and instead a vague entity is to blame (i.e., masculinity compels me to act in this way). It renders invisible the acts of misogyny and homophobia/transphobia by men through an ambiguous blaming on the social construct of masculinity (Banet-Weiser and Miltner, 2015; Gruenwald & Kelly, 2014; Waling, 2018). As discussed earlier, in this sense, masculinity becomes positioned as something that is done to men, or something they are victim of, rather than something that men may actively engage, and therefore removes their agency and a sense of responsibility for potentially harmful actions (Waling, 2019).

Another broader challenge with the term toxic masculinity is that it does not consider intersecting identities in men's lives, or consider closely enough the structural issues that shape men and boys. In drawing heavily upon and applying Raewyn Connell's (1995) work on multiple masculinities, Michael Salter (2019) argues that the term toxic masculinity encourages assumptions that the causes for male violence and other social problems are the same everywhere rather than specific to that society and that culture, and that it therefore assumes the solutions are the same as well. Salter believes that in focusing on men as individuals versus culture or broader social structures, those that are most marginalized because they are so impacted by structural inequalities, such as racialized men or men who live in poverty, are targeted as bad men enacting toxic behaviours. Connell (1995) believed that if some men and boys feel inadequate or not masculine enough, then they might act out aggressively or violently as a means of living up to a dominant standard of masculinity. Women and girls hold themselves to similar high standards of femininity, but I argue a key difference is that when some women act out as a means of attaining those standards it can often present as self-harming behaviours such as, for example, restrictive eating or over exercising. However, as Brown (2018) describes, this also serves as a means of

women using their bodies to comply and resist to dominant cultural norms. But for some men, when they are left feeling frustrated, angry, or emasculated, they act out in an external way that causes harm to others.

Although notions of toxic masculinity have been qualitatively theorized extensively in recent times, it is important to note that quantitative studies in this area are scarce and limited. Furthermore, while it is well acknowledged that the development of a masculine identity plays a crucial role in the health and well-being of men and boys, only a few studies have explored that sense of identity as it relates to male survivors of childhood sexual abuse sexual abuse (Easton, 2014; Gartner, 2000; Kia-Keating et al., 2005). What these studies do tell us is that experiences of childhood sexual abuse can and do have significant impact on developing masculine identity and has far reaching impacts well into adulthood for male survivors (Easton, 2014; Kia-Keating et al., 2005; Lisak, 1994). Male survivors can face enormous pressures from parents, peer groups, and their community at large to demonstrate traditional masculine norms while contending with their sexual abuse histories, something directly at odds with basic tenets of dominant masculinity insisting they appear strong and invulnerable (Kia-Keating et al, 2005). Strict adherence to more toxic masculine norms makes it even more challenging for survivors to develop functional and integrated identities (Kia-Keating et al, 2005). As I further discuss the impact of early childhood sexual trauma on the lives of these young men and boys, managing this disconnect between experiences of early childhood sexual abuse and the notion of living up to prescribed cultural standards and norms of masculinity becomes crucial to survivors' healing processes.

Male Childhood Sexual Trauma

Childhood sexual abuse is traumatic and can overwhelm the lives of its victims and result in profound challenges to emotional regulation, physiological arousal, interpersonal relationships, cognition and memory (Herman, 2015; Van der Kolk, 2014). As renowned expert in trauma and abuse Dr. Judith Herman (2015), tells us:

at the moment of trauma, the victim is rendered helpless by overwhelming force...Traumatic events overwhelm the ordinary systems of care that give people a sense of control, connection, and meaning. . . Traumatic events are extraordinary, not because they occur rarely, but rather because they overwhelm the ordinary human adaptations to life...They confront human beings with the extremities of helplessness and terror, and evoke the responses of catastrophe (p. 34).

Trauma survivors may face triggering events that results in intense emotions without clear memory, or a flashback of memories disconnected from one's emotions. Central to Herman's (2015) work was her assertion that that psychological trauma and outcomes for a survivor were influenced heavily by the society in which it occurred. Herman (2015) argued that by contextualizing the environment and a person's reaction as a normal response to an overwhelming experience or event, it allowed for the shifting of blame away from the survivor, who was previously perceived as weak or defective in their suffering.

Experiences of trauma can result in far-reaching changes in the way meaning is given to life, with those changes mainly centered on the loss of trust and the sense of disconnection from other, and feelings of despair, terror, rage, and hopelessness. After a traumatic experience, "the world is experienced with a different nervous system. The survivor's energy now becomes focused on suppressing inner chaos, at the expense of spontaneous involvement in their lives"

(Van der Kolk, 2014, p. 53). Survivors may find themselves in a constant state of hypervigilance and irritability yet be unable to identify the cause of their arousal state (Herman, 2015). Even when far removed from the traumatic experience, they still sometimes find themselves trying to organize one's life and surroundings as if the trauma is still happening, unchanged and immovable, with each new experience or event somehow tainted or contaminated by the past.

For both males and females, research shows that risk of child sexual abuse may be directly correlated with such factors as divorce and domestic violence within the family, parents or guardians who suffer substance abuse issues, or parents or guardians who have mental health challenges, and as a result are physically or emotionally unavailable or incapacitated in their care-taking roles (Dube et al., 2005). Childhood sexual abuse can also commonly involve the use of force or threats, and there is some correlation between the perpetration of sexual abuse of boys and a statistically significant use of violence as part of that sexual abuse (Dube et al., 2005, Kia-Keating, et al., 2009). Studies of childhood sexual abuse have demonstrated strong relationships to several adverse health, behavioural, and social outcomes among male and female survivors such as severe psychiatric disorders, suicidality, substance abuse, chronic health issues, obesity, and high risk sexualized behaviours (Dube et al, 2005; Fisher & Goodwin, 2008). Impacts of sexual abuse on both genders demonstrate that the most common negative outcomes include emotional and behavioural problems, post-traumatic stress disorder, depression and anxiety, suicidality, substance abuse, poor self-esteem, academic problems, and sexualized/sexual acting out behaviours (Allagia et al., 2008: Kia-Keating et al, 2010: Sorsoli et al, 2008; Van der Kolk, 2014).

The available research literature does suggest that while there are many similarities in the way males and females experience childhood sexual abuse and trauma, there are also some

distinct differences. Some studies using standardized measures, such as the Trauma Symptom Checklist, have shown no significant differences in trauma impacts between men and women who have experienced sexualized trauma in childhood (Allagia et al., 2008; Briere, 1988; Roesler & McKenzie 1994). Other studies concluded that male survivors are less likely than female survivors to experience anxiety, depression and engage in acts of self-harm, while females were more prone to prolonged periods of depressive episodes (Allagia & Millington, 2008, Gartner, 2017a; Kia-Keating et al., 2009). Still other studies found that male sexual abuse survivors are more likely than female survivors to experience suicidality, have more significant substance abuse issues, and show more externalized aggressive characteristics and issues with managing anger and resulting violence (Allagia & Middleton, 2008; Denov 2004). These outcomes are significant to this study, as issues related to challenges with anger, emotional regulation, outbursts of violence, and problematic substance use histories were common in participants' narratives of their therapeutic work with survivors.

While some of these studies noted here have used population-based samples (for e.g., Denov, 2004; Dube et al., 2005), it is important to note that many have been limited by examination of child sexual abuse among very specific clinical samples, such as individuals in substance abuse recovery programs, persons incarcerated or involved in the justice system, and psychiatric in-patients (for example, Allagia, 2004; Sorsoli et al., 2008). A major cross Canada study- *The Canadian Incidence Study of Reported Child Abuse and Neglect 2008* - examined the incidence of reported child maltreatment and the characteristics of the children and families investigated by child welfare by considering almost 16,000 reported allegations over 112 different agencies. Among investigations of childhood sexual abuse, 37% of cases involved boys and adolescent males (Trocme et al, 2003). In the US, a nationwide study that looked at children

in community mental health services, 47.5% of reported childhood sexual abuse cases involved male children (Walrath et al., 2006). These propensity for a high degree of reported sexual trauma in boys and adolescent males among these specific populations is an important distinction to make, as these are incidences of childhood sexual abuse that come to light and are reported, for example, through investigations by child welfare or by youth mandated to mental health or substance use - in other words, they involve an intervention that led to the discovery or disclosure of instances of childhood sexual trauma amongst very specific populations vs. general populations where disclosure is much less common. Reasons for this were not entirely clear from these studies, but I would argue it may be related to factors such as the vulnerability of these youth in relation to the perpetration of sexual trauma, or a heightened awareness among professionals that statistically these boys and adolescent males are more likely to have early childhood sexual trauma histories.

The literature also demonstrated the degree to which young men and boys will not readily disclose experiences of sexual abuse for fear society will neither condone nor even accept it (Fisher & Goodwin, 2008; Lew, 2004; Van der Kolk, 2014). Major studies on the phenomena of early childhood sexual trauma share similar findings, in that the problem is much more common than we assumed, it is vastly under-reported, it is grossly under-recognized, and is often undertreated (Cohen et al., 2017; Denov, 2004; Fisher & Goodwin, 2008; Lisak, 1994; Van der Kolk, 2014). As noted previously, issues that can occur in the aftermath of male childhood sexual abuse include challenges in defining the unwanted experience as sexual abuse or trauma; fear, isolation, and shame; the impact of masculine gender socialization on how the experience is understood; the possibility that abuse by women or even same age same sex peers will be defined

as sexual initiation or exploration; and the impact on sexual and other intimate relationships into emerging adulthood (Fisher & Goodwin, 2008; Gartner, 2000, 2017a).

Sexual abuse perpetrated by a male against another male can bring questions and doubts around one's sexual orientation and sexual identity (Corbett, 2016; Fisher & Goodwin., 2008). Survivors are often left struggling with heightened issues related to homophobia, shame, and stigma (Easton, 2014; Lew, 2004). This is an outcome of being raised and socialized in a heteronormative society that reifies and privileges heterosexuality in its customs, traditions, institutions, and ways of understanding. As such, it illustrates one of the most heightened differences in outcomes for male and female survivors, as abuse by a male perpetrator for many men and boys leads to questioning sexual orientation and sexual identity and challenges in reconciling their membership as masculine within a heteronormative and predominantly homophobic society.

As a means of coping, some male survivors adopt a hyper-masculine persona in which they display exaggerated and toxic masculine attitudes and norms such as violence, aggression, homophobia, and misogyny (Dorais, 2002; Kia-Keating et al., 2005). Hegemonic/dominant masculinity, influenced by a heterosexist society, communicates to young males that, in sexual situations, they can expect to be the one who desires another and to also be the one who typically initiates sex (Corbett, 2016; Kimmel, 2012). Sexual abuse of men and boys subverts dominant expectations of masculinity in such a profound way that it is often experienced as an attack on the male self that is emasculating (Corbett, 2016).

One particularly pervasive myth or untruth that recurs throughout the brief literature available involves the phenomena of sexual abuse of young men and boys by older female perpetrators. If we accept that hegemonic/dominant masculinity and those that subscribe to it

often prides itself on notions of sexual conquest and prowess, then we must acknowledge that young men are sometimes actively encouraged to recognize or accept such experiences as initiation, rites of passage or healthy adolescent exploration, rather than the exploitative abuse of power and control that they are in reality (Denov, 2004; Gartner, 2017b; Lisak, 1994). The fact that this is such a widely accepted myth is dangerous in that it limits the ability of a survivor to speak of his abuse, as to do so would result in ridicule and, in rejecting these sexual advances, call into question their manhood. As constructed within hegemonic/dominant masculinity, male sexual exploits can sometimes appear to young men and boys as something that seems necessary, often aggressively pursued, and an activity they should readily seek to participate in and enjoy (Kimmel, 2008). But when we normalize, trivialize, or sensationalize abuse perpetrated by an older female against a male youth, on both the personal level and the cultural level we overlook the seriousness and the implications of the act itself, both as a transgression of power and a legitimate form of abuse or trauma.

For heterosexual youth, surviving sexual abuse challenges the norms of their perceived masculinities set forth by a heteronormative and patriarchal society. For LGBTQI2S+ youth, it raises concern they were abused because they were perceived as gay after showing some indication or hidden signal to others, or that the abuse experience and the sexual transgression itself somehow contrived to "make them gay" (Cassese, 2000; Corbett, 2016). Male survivors may also feel the need to act out sexually with female partners in an effort to re-establish or reclaim their heterosexual identity. They may also feel hurried into identifying their sexual orientation, and later come to associate their sexual identity and their sexual activity with feelings of betrayal, exploitation and secrecy (Gartner, 2018; Lew, 2004). Within early adolescence in particular, self-concepts of both sexual orientation and gender identity have yet to

fully coalesce, and fears, prejudices, and misinformation regarding identity and orientation can be particularly prominent and therefore especially problematic within developing adolescence (Gartner, 2000). As a result, perhaps one of the greatest crimes of male childhood sexual abuse is how it robs young men of their ability to discover their orientation and sexual interests in developmentally appropriate ways.

As limited as the literature in general is on male childhood sexual trauma experiences, it is even sparser when considering the experiences of LGBTQI2S+ youth. James Cassese's Gay Men and Childhood Sexual Trauma: Integrating the Shattered Self (2000) is one of the few works based solely on the experiences of gay and bisexual male survivors, studies and researched primarily by predominantly gay male scholars and therapists. Most of the major gender scholars writing or researching male childhood sexual trauma – Gartner, Lisak, Kimmel, to name a few – are white, cisgender heterosexual men. Their positionality as such arguably brings assumptions to their work that may not fully encapsulate the experience of marginalized masculinities, and in turn views the issue through a privileged heterosexual lens. Missing from their work are more prominent examples, for instance, of a male child who identifies as LGBTQI2S+ prior to the experience of sexual abuse or a racialized child challenged by cultural expectations. From childhood abuse to homophobic jokes that shame and ridicule to hate crimes, certain gay, bisexual and trans men and boys experience dominant culture as traumatic. Experiences of childhood sexual abuse only compound these issues further and exacerbate that trauma; however, it is a phenomenon rarely addressed in current literature.

Herman (2015) describes how traumatic experiences impact not only an understanding of self, but also the systems of attachment and meaning that provide connection and linkage to families and community. Traumatic events can destroy our underlying assumptions around safety

in the world and erode the positive value of self (Herman, 2015). The very act of being sexually traumatized is conceptualized as "contempt for the victim's autonomy and dignity and thus...destroys the belief that one can be oneself in relation to others (Herman, 2015, p. 53). If we accept that a secure connection with caring people is a necessary foundation of healthy personality development, then along with the shattering of this connection through trauma comes doubt, fear, and insecurity. As Herman (2015) tells us, developmental conflicts, which may have been long thought negotiated or resolved, are now often painfully reopened and left uncertain.

Because of the overarching context of hegemonic masculinity and toxic masculine culture, male survivors can struggle with a yearning for connection that conflicts and constricts with an inability to establish closeness to another for a variety of reasons, including fear of some re-occurring abuse, or fears of perpetrating abuse themselves (Kia-Keating et al., 2005; Lew, 2004). It may also seem a requirement of masculine culture that they remain stoic or express little emotion regarding the traumatic experience (Kia-Keating et al., 2005; Lew, 2004). Lisak (1994) describes that "paralleling the damage to the survivor's self is an equally pervasive assault on his connection to others" (p. 545) and notes how the resulting alienation can prevent or prove debilitating to the formation of positive interpersonal relationships. According to Lisak (1994), it is these connections that are needed to mitigate the resultant fundamental mistrust of others, including the expectation that others can and will potentially do us harm. And yet as Herman (2015) tells us, recovery takes place within the context of relationships – connection is recovery, and recovery cannot occur in isolation. Therefore, what becomes key to the treatment of male trauma survivor's experience is the reparation of that sense of connection, and the discovery of safety in trusting relationships that all form parts of his healing process.

Challenges around Disclosure

Several compounding factors impact the identification of childhood sexual abuse victims, from issues such as constructing the experience as abusive to impacts on disclosure rates (Fisher & Goodwin, 2008; Gagnier & Collin-Vezina, 2016; Gartner, 2017a). One significant issue impacting our understanding of the experience of men and boys is the scarce research that is available explicitly detailing the experience of disclosure and help-seeking of young male survivors. Furthermore, the studies that do exist consider experiences of two genders have a disproportionate number of female voices due to challenges with recruitment of males, while other studies have comparatively a small sample of males as participants (Allagia & Millington, 2008). There has also been a tendency to measure only certain types of outcome, such as disclosure that resulted in official accounts being provided to police or child welfare agencies, or the pursuit of criminal charges and prosecution through justice system (Sorsoli et al, 2008). This excludes an act of disclosure that involves conveying or attempting to convey to another person their experiences of trauma without necessarily contacting or involving authorities.

In studies that compared male and female disclosures, findings showed that males struggled with disclosure in similar ways (for e.g., feelings of shame, not wanting to hurt others by disclosing) but in a number of studies there are several distinct factors that stood out as specific barriers to disclosure for male survivors. These factors are what Fisher and Goodwin (2008) call cultural myths that leaves male survivors with profound feelings of isolation and stigmatization. A central myth is the common discourse that boys are rarely victimized. Because of a belief that men and boys cannot be victims, male survivors struggle with feeling vulnerable and weak versus tough and strong, and equate their abusive experience to a *female experience*, something that only happens to girls and women. This serves as a violation or transgression to

their masculine identity (Allagia, 2005; Gagnier & Collin-Vezina, 2016; Kia Keating et al., 2005) influenced by a patriarchal, heteronormative, and misogynistic culture that devalues women and views them as weak and powerless. Boys and men are emasculated by these cultural biases. They also fear being perceived as homosexual or gay, as the result of homophobic and heterosexist biases in culture, and the enormity of pressures in heteronormative society to conform to dominant (i.e. hegemonic and heterosexual) ways of being a man (Allagia, 2005; Gartner 2017b; Lisak, 2017). These same studies demonstrated how female survivors generally felt more conflicted about their responsibility in an abusive experience, and struggled with feelings of shame and blame, while at the same time largely anticipated being blamed or not being believed by those to whom they disclosed their trauma (Allagia, 2005; Gagnier & Collin-Vezina, 2016).

Since experiences of childhood sexual abuse and feelings of hopelessness or terror are seemingly at odds with certain masculine ideals, confiding a history of abuse or the resulting challenges become particularly problematic for some survivors (Kia-Keating et al., 2005; Lew 2004; Lisak, 2017). However, studies show that disclosure is often key to relational development and that the inability to talk about these experiences contributes to disconnections in relationships to families, communities and at times disassociation within themselves (Gartner 2000, Kia-Keating et al., 2005).

In their work with female survivors of sexual assault, McKenzie-Mohr and Lafrance (2011) invoke the term "tightrope talk" to describe survivors' attempts to make meaning of their experiences by negotiating and navigating both agency and blame. McKenzie-Mohr and Lafrance (2011) challenge the oversimplification of binaries such as powerful or powerless, or strong or vulnerable. This is particularly relevant in how mental health clinicians help young

male survivors unpack their stories. Brown (2018) notes that disclosing trauma can be perceived as dangerous by the survivor, as it exposes them to uncomfortable or painful emotions and fears of being blamed or not believed. This danger may, therefore "shape the storytelling, and caution and self-surveillance may render invisible or disqualify aspects of the story (Brown, 2018, p. 46). As Brown (2018) cautions, we should not misinterpret uncertainty in an account as a lack or absence of past traumatic experience.

The powerful messages about what it means to be a man restrict and constrain how men and boys talk about their trauma (Fisher et al., 2008; Gartner, 2017a; 2017b)). When faced with trauma in such forms as childhood sexual abuse, powerfully overt and subliminal messages around masculinity make it difficult for men to acknowledge trauma and seek help, and leave them feeling ashamed to disclose any pain or suffering, or to seek comfort and support (Fisher & Goodwin, 2008; Gartner, 2000; Lisak, 1994). Studies show that men who are better able to locate, recognize and process their emotional experiences within the context of their male identity, including those specific to the trauma experiences themselves, were more likely to be successful in recovery (Kia-Keating et al., 2005; Fisher & Goodwin, 2008; Gartner, 2017b).

Another challenge facing survivors' help-seeking is the lack of a recognizably comfortable or accepting atmosphere within our current mental health settings for the disclosure, and the tendency of mental health professionals to underestimate or dismiss the prevalence of male childhood sexual abuse (Gartner, 2008; Teram et al., 2006). Ironically, it is dominant masculinity's privileged position in our Western culture that "contributes to these impediments, making it difficult for male survivors to acknowledge themselves as victims and furthermore for health professionals to even view them as such" (Teram et al., 2006, p. 513). The myths surrounding male childhood sexual abuse continue to perpetuate in society and serve to

"exacerbate the difficulties men have in disclosing the experience of sexual assault and increase their stigma while hindering the development of appropriate services and empirical research" (Stermac et al., 2004, p. 7). Therefore, addressing these myths directly and working to decrease stigma is key to the male survivor's recovery process.

Healing Processes

Childhood trauma due to physical abuse, neglect, or sexual abuse is recognized in the research as a severe problem in our society today, and it well documented how traumatic experiences may result in disruption or injury to the developing brain that affect a child's developing functioning and can potentially result in lifelong problems and challenges if left untreated (Ford & Courtois, 2016; Thomason & Marusak, 2017; Van der Kolk, 2014). Treatment modalities used today with young survivors of early childhood sexual trauma include, but are not limited to, trauma-focused cognitive behavioural therapy (TF-CBT), narrative therapy, feminist based approaches, attachment therapy, dialectical behavioural therapy (DBT), acceptance and commitment therapy (ACT), sensorimotor psychotherapy, eye movement desensitization and reprocessing (EMDR), adventure therapy, and mindfulness strategies (Fisher & Goodwin, 2008; Ford & Courtois, 2016; Gartner, 2017b; Herman, 2015; Van der Kolk, 2014). Individual therapy is often observed as the best starting place for men and boys in developing trust and rapport with a clinician (Fisher & Goodwin, 2008; Lew, 2004), but studies show that men and boys also draw strength and healing from therapy delivered outside of traditional officebased settings, such as outdoor nature therapy; self-help books and guided journaling; practiced mindfulness; and, in particular, group therapy processes (Kia-Keating et al., 2005; Fisher & Goodwin, 2008; Gartner, 2017b). Research on group therapy shows that it can serve to reduce isolation throughout the process of healing by demonstrating that people are not alone in having

these experiences, and in providing survivors with a sense of feeling more accepted, validated, visible, and heard (Fisher & Goodwin, 2008; Gartner, 2017b; Lisak, 2017).

For boys and adolescents, employing informal settings outside the office (e.g., playground or recreational areas), involving musical instruments or art activities, using humor and self-disclosure, and providing psychoeducational groups aimed at challenging myths around male sexual trauma may prove more effective than traditional psychotherapy offered in traditional office like settings (Kiselica & Engar-Carlson, 2010; Kiselica et al., 2011).

Combining psychological interventions with some form of sporting activities has also been demonstrated to reduce barriers to male-help- seeking, including an impact on what is theorized as a perceived social threat associated with needing help (Kiselica et al., 2011).

Men and boys cope with masculine expectations throughout the therapeutic process, first through deconstruction and subsequent reconstruction of the masculine order, and finally a renegotiation or a new understanding of what it means to be masculine (Fisher & Goodwin, 2008; Gartner, 2017b; Kia-Keating et al., 2005). Ultimately, survivors may seek to develop skills to amend problematic behavioural patterns that may have hindered relationships or impacted their lives in a myriad of different ways (Fisher & Goodwin, 2008; Kia-Keating et al., 2005). Through healing from past violence and abuse, they may also start to seek out, as an alternative, more stable, non-abusive, and supportive relationships with others (Fisher & Goodwin, 2008; Kia-Keating et al., 2005). However, while the ability to connect or reconnect with others is described as a critical component to any healing processes in the literature, the idea of "embracing connection, empathy, and vulnerability continue to be a difficult renegotiation for many survivors in many contexts" (Kia-Keating et al., 2005).

Judith Herman has been described as a pioneering clinician in the field of trauma work

with a tremendous influence on the theoretical understanding of the sequelae of trauma. Although her book *Trauma and Recovery* (2015) was first written and published almost three decades ago, it is still considered one of the most essential theoretical works in the field of trauma (Suleiman, 2008, p. 285). Herman was highly critical of the treatment survivors of child abuse received in mental health systems, and believed in treating symptoms as causes, professionals working in these systems served to perpetuate the culture of blaming the victim, by focusing on the victim's character as the source of the problem. In *Trauma and Recovery*, Herman (2015) proposed the diagnosis of Complex Post Traumatic Stress Disorder (C-PTSD). Herman's definition of C-PTSD differs from the current definition of Post-Traumatic Stress Disorder as it seeks to address the occurrence of multiple, repeated trauma throughout the lifetime as opposed to a single acute traumatic experience (Herman 2015; Van der Kolk, 2014). Although C-PTSD has yet to be recognized as an official diagnosis, it is used extensively as a guideline of practice and treatment in the field of trauma by both therapist and researchers (Van der Kolk, 2014).

Herman (2015) proposes a three-stage model for recovery in the treatment of trauma. According to Herman, these key stages of recovery include establishing safety, reconstructing the trauma story, and restoring the connection between survivors and their community. In stage 1: "safety and stabilization", Herman (2015) notes that establishing safety takes precedence over everything else because, without that secure base, the trauma work cannot begin. In stage 2, "remembrance and mourning", the survivor tells the story of their trauma. As Herman (2015) notes, they tell it "completely, in depth and detail. This work of reconstruction transforms the traumatic memory so that it can be integrated into the survivor's life story" (p. 175). Through the therapeutic relationship, the survivor may be empowered by the clinician in this stage, as "the

choice to confront the horrors of the past rests with the survivor. The therapist plays the role of witness and ally, in whose presence the survivor can speak of the unspeakable" (Herman, 2015, p. 175). Herman (2015) notes that together the patient and therapist must learn to "negotiate a safe passage between the twin dangers of constriction and intrusion" (p. 176). She cautions that "avoiding the traumatic memories leads to stagnation in the recovery process while approaching them too precipitately leads to a fruitless and damaging reliving of the trauma" (Herman, 2015, p. 176). In stage 3: "reconnection," Herman (2015) describes how the survivor has reconciled their traumatic past and is tasked with creating a new future. She notes how the trauma has forever changed relationships, and the survivor must develop new ones. Thoughts, feelings and beliefs once sustaining have now been challenged, and the survivor must look for new ways to reclaim faith (Herman, 2015). As Herman (2015) describes it:

Traumatic events destroy the sustaining bonds between individual and community. Those who have survived learn that their sense of self, of worth, of humanity, depends upon a feeling of connection to others. The solidarity of a group provides the most reliable protection against terror and despair and the most potent antidote to the traumatic experience. Trauma isolates; the group re-creates a sense of belonging. Trauma shames and stigmatizes; the group bears witness and affirms. Trauma degrades the victim; the group exalts her. Trauma dehumanizes the victim; the group restores her humanity (p. 214).

In *The Body Keeps the Score: Brain, Mind, and Body in the Healing of Trauma*,

Bessel Van de Kolk (2014) explores "the extreme disconnection from the body that so many people with histories of trauma and neglect experience" (p. 91). Van der Kolk (2014) draws on three main areas of study as a focus of treatment for trauma survivors that include neuroscience

(function brain processes), developmental psychopathology (the impact of painful traumatic experiences on brain development) and interpersonal neurobiology (a person's behaviour and how that impacts on psychoemotional and neurobiological states of those around them). Through a neurodevelopmental lens, Van der Kolk (2014) describes how parts of the brain that have evolved to monitor for danger remain over activated for trauma survivors, and that the slightest sign of danger, whether real or imagined, can trigger an acute stress response, painful emotions, and overwhelming sensations. Van der Kolk (2014) proposes that these post-traumatic reactions make it difficult for survivors to connect with other people since closeness can sometimes signal a sense of danger, and yet the close contact a survivor may come to dread is often the thing needed most in order to begin healing. While helpful in understanding some of the physiological impacts of trauma, der Kolk's (2014) theories do not fully address impacts of social determinants of health on trauma survivors or, in particular, discursive social forces that can impact how a survivor understands and integrates past trauma into their lives.

It is increasingly recognized that more specific gender-sensitive training is needed for mental health professionals when addressing the unique needs of male survivors (Addis & Mahalik, 2003; Gartner, 2018; Kia-Keating et al., 2005; Mahalik & Burns, 2007; Mahalik et al., 2007). Current treatment practices can pathologize male survivors in terms of their focus on symptomology, particularly within the context of problematic or externalizing presenting behaviour (Gartner, 2000, 2017a; Lew, 2004). This, in turn, can contribute to how survivors often internalize blame and guilt for these traumatic experiences (Lew, 2004).

Studies show that mental health professionals rarely screen their male clients for specific experiences of sexual abuse (Addis & Mahalik, 2003; Gartner, 2017b; Mahalik et al., 2003; Teram et al., 2009), which raises the concern that perhaps one of the reasons why male survivors

have not come to the forefront in our research is that clinicians, physicians, and other mental health professionals are failing to assess appropriately around past experiences of childhood trauma. Clinicians are often less likely to recognize potential symptoms of past trauma in male survivors as internalized (i.e., viewed and formulated as a mood disorder, such as anxiety or depression) and tend to instead focus on externalizing behaviours, such as difficulties with displays of anger, violence, or emotional regulation (Addis & Mahalik, 2003; Kimmel, 2008; Mahalik et al., 2003, 2007; Teram et al., 2009). Many clinicians describe possessing little or inadequate specialized training in regard to working with male survivors of sexual abuse, which may in turn heighten their anxiety and impact their approach to the treatment of male sexual trauma survivors (Teram et al., 2009).

In studies related to sexually abused men's' experiences in psychotherapy, survivors highlighted the need to work slowly, be open to alternative approaches that may lie outside office based norms, and the importance of providing specific psychoeducation at critical points in treatment to assist with development and implementation of healthy emotional skills (Addis & Mahalik, 2003; Fisher & Goodwin, 2008, Gartner, 2000). However, survivors participating in these same studies frequently noted that it is problematic to presume that male clients are emotionally incompetent or unable to manage complex emotional states, and that therapist should not avoid intricate emotional work as part of treatment, but rather be more attuned to helping survivors manage those complicated emotional responses and triggers as they may arise (Fisher & Goodwin, 2008; Gartner, 2000; Mahalik et al., 2007).

To varying degrees, a male survivor is often faced with the challenge of rejecting the traditional standards of masculinity. Lisak (1994) believes that male survivors of childhood sexual abuse must be "given the tools to deconstruct the gender system and their individual

experience of gender socialization in order to fully engage in the process of healing from abuse" (p. 258). While still successfully engaging in traditional expectations of masculinity, survivors must potentially allow themselves to express fear or uncertainty or show vulnerability, all of which are experiences that are often at odds with traditional expressions of manhood. Therefore, a sexual trauma survivor must work to both contain, resist and redefine traditional masculine roles as part of his healing process (Kia-Keating et al., 2009), which ultimately may prove key to a healthy recovery.

Numerous studies have documented that exposure to interpersonal trauma during childhood is related to increased incidence of affect and impulse dysregulation, alterations inattention and consciousness, disturbances of attribution and schema, and interpersonal difficulties. Affect dysregulation may present as lability, flat affect, explosive or sudden anger, or incongruous or inappropriate affect response, while behavioural expressions of affect regulation may include withdrawal, self-harm, heightened aggression, oppositional and defiant behaviour, substance use, or other compulsive behaviour (Cicchetti & Rogosch, 2018; D'Andrea et al., 2012; Pollak et al., 2009). Disturbances of attention and consciousness may manifest as dissociation, which Herman (2015) describes as the numbing of one's self, memory disturbance, inability to concentrate or sustain attention, and poor executive functioning, such as the ability to plan or problem solve (Kaplow et al. 2008).

Children exposed to trauma in early childhood often have distorted attributions about themselves and the world around them that may act to exacerbate feelings of shame and guilt, promote negative or distorted cognitions, and result in a loss of control and poor self-efficacy (Gibb & Abela, 2007; Kim & Cicchetti, 2006; Valentino et al., 2008). Interpersonal difficulties in children following abuse or neglect may include disrupted or poor attachment styles, issues

with trust, diminished social skills, poor perspective taking, an inability to understand social interactions, expectations of harm from others, and poor boundaries (Kim & Cicchetti, 2006; Perlman et al., 2008). Children who are exposed to trauma and abuse are also often at a much higher risk for further victimization through witnessing interpersonal violence in their homes, which has shown increases risks and severity of internalizing, externalizing, relational, and academic problems in later childhood (Finkelhor & Turner, 2016). One of the few major studies that looked at the association between male childhood maltreatment, male childhood sexual abuse and adult mental health outcomes found that men who experienced early childhood sexual trauma with or without other forms of child maltreatment had significantly higher rates of mental health issues, were more likely to be diagnosed with mental health disorders, and were more likely to self-harm and attempt suicide than men who did not experience childhood sexual abuse (Turner et al., 2017).

The scope of this work limits a more comprehensive review of effective treatment modalities and healing processes. However, what is clear from the studies reviewed and from the work of theorists such as Herman and van der Kolk is that following childhood victimization there is a spectrum of specific symptoms commonly experienced by survivors of complex trauma, and these symptoms cannot be accounted for fully by any existing diagnoses or combination of diagnoses, found in the current Diagnostic and Statistical Manual (DSM-V), including Post Traumatic Stress Disorder (PTSD) (D'andrea et al., 2012). A mental health diagnosis can serve the purpose of validating people's experiences, and empower them to seek treatment while recognizing their resiliencies, but it can equally serve to pathologize their experiences, perpetuate stigma related to mental health, and potentially reduce the likelihood of positive treatment outcomes. Sexual trauma survivors are more than their trauma, and more than

their manifestations of pain. By applying interventions that comprehensively address the full range of biopsychosocial issues that men and boys exposed to childhood sexual trauma face while conceptualizing and planning treatment approaches through a gendered lens, I believe we have the potential as a mental health system to dramatically increase the likelihood of positive treatment outcomes and better the lives for male childhood sexual trauma survivors.

Chapter Three: Methodology

This chapter provides a detailed overview of the methodology and research design in the study. It begins with the research question and objectives of the study, and then outlines the research design and methodology that was chosen and utilized. This chapter will also outline the methods used for participant sampling, data collection, and explain the rationale and process related to data analysis. Finally, it also provides an overview of how confidentiality and anonymity was maintained as well as a framework for ethical considerations.

Research Questions and Objectives

This study explored narratives of trauma therapists and their work with young male sexual trauma survivors. It described the experiences of young survivors from the viewpoint of participants as therapists, and interpreted trauma narratives through an understanding of the influence of toxic masculinity culture on young males and the resulting impacts on their healing processes. It also explored the challenges and complexities of the treatment work participants provide to survivors. Research questions that guided this study included: 1) how does the experience of childhood sexual trauma impact the lives of young survivors? 2) In what ways does the phenomenon of childhood sexual trauma intersect with the experiences of being socialized as a young male? 3) How do rigid expectations of dominant masculinity – and particularly those of toxic masculine culture -disrupt healing processes of young survivors? 4) What current treatment interventions and strategies are utilized in treating male survivors, and of those what appears most effective and what is missing or lacking? 5) How might we reduce the stigma for males associated with being a childhood sexual trauma survivor and the stigma related to mental health help seeking in males?

Analysis in this study centered on how society fosters assumptions and myths around meanings of masculinity, which in turn can perpetuate incidents of gendered violence, misogyny, homophobia, and transphobia, all of which serves to hinder the way boys and young men heal from sexual trauma. The study explored how participants, as therapists, approach treatment processes to address the gendered nature of male childhood sexual trauma and how they might challenge rigid and constraining constructs of masculinity that serve to impede healing and recovery. There was a particular focus on the complex relationship between male childhood sexual trauma and homophobia and misogyny, and how these experiences become internalized in some survivors and result in significant challenges with sexual identity and a masculine sense of self (Allagia et al., 2008, Fisher & Goodwin, 2008; Gartner, 2000; Kia-Keating et al., 2005, 2009; Lisak, 2017). This in turn led to deeper exploration of the dangers of injurious speech (Brown, 2007; McKenzie-Mohr & Lafrance, 2011) and the need for counter-stories related to available trauma discourse (Brown, 2007; Brown, 2011). Participants also described struggles with systemic constraints and barriers to treatment inherent in the medical model in which they were situated, and ways in which they worked to resist these constraints. As previously reported, experience of male childhood sexual trauma is a phenomenon that is

As previously reported, experience of male childhood sexual trauma is a phenomenon that is under-studied, under researched, and not well understood. My ultimate hope is that this work will add to the literature available on male childhood sexual abuse, and contribute to the conversation about how we might offer agency in our work with young male survivors, so that they may process emotions and story experiences without guilt, shame, humiliation, or defeat.

Theoretical Framework

Participants' narratives of experiences are critically examined through a social constructionist and postmodern lens throughout this study. This work also draws on parts of

feminist and queer theory, in particular their transformative nature, their politicized stance, their interest in knowledge possession, and exploration of gender and sexual identity. Social constructionism seeks to recognize, capture, and honour multiple meanings through consideration of a complexity of views (Berger & Luckmann, 1966; Creswell, 2013). Social constructionism rejects an objectivist viewpoint and purports that there is no objective truth, and believes that knowledge arises from engagement with the world, with knowledge and the knower as interdependent and embedded within history, context, culture, language, experience, and understandings (Berger & Luckmann, 1966; Denzin & Lincoln, 2017). Therefore, what we take as true and objective is the result of discursive social processes that take place within historical and cultural contexts. Understanding experiences and perceptions from a social constructionist perspective offers potentially new ways to consider constructs of masculinity and how they intersect with experiences of sexual trauma outside of currently constructed meanings.

Like social constructionism, postmodernism is interested in challenging what is known and how it becomes known. Postmodernism believes there is no one absolute, universal, or discoverable truth, or one fixed essential self or identity that exists outside human experience (Brown, 2007; Denzin & Lincoln, 2017). Postmodernism calls for a critique of foundational knowledge and privileged discourses, also called grand narratives, and any taken for granted assumptions. The central message of postmodernism is the inherent danger of the *one* story that leaves no room for alternate versions (Payne, 2014). Social constructionism and postmodernism assist in understanding what is constructed, how it is constructed and the very question of what it means to *be* constructed (Payne, 2014). They also offer a framework to deconstruct meaning making of experience, and subsequently consider how that meaning constructs future experiences. Situating men's stories in this study within a social constructionist/postmodern

framework allows for their experiences to be understood within the various contexts that emerge. It also allowed for greater understanding of men's experiences through how they process their experiences, in this case primarily their traumas, and how they consider and construct their identities.

A Turn towards Narrative

As social workers, we are interested in hearing, understanding and working with people's stories, and those experiences told through stories are often very central to our practice. Narrative inquiry, as defined by Clandinin and Connelly (2000), is the study of experience as story rooted in a situational way of knowing. Within that knowing, researchers and participants together can come to understand experience under certain contexts, during certain periods of time, and through interacting with certain others. Narrative inquiry involves a paradigm shift from thinking about stories to engaging with stories, with stories positioned as a way of both being in and engaging with the world (Bruner, 1991; Clandinin & Connelly, 2000). It promotes a change in power relations, moving away from the pursuit of objective knowledge as a researcher, and towards acknowledging the value of relationship between researcher and participant. Narrative inquirers arguably conduct research that is less generalizable in favour of research that is more specific, with emphasis placed on the use of words and narratives to impart meaning versus numbers and statistics (Pinnegar & Daynes, 2007; Reissman, 2007). Narrative inquiry involves a change from seeking more definitive answers towards valuing subjective and situated knowledges. It rejects one way of knowing and recognizes multiple ways of knowing and understanding human experience (Pinnegar & Daynes, 2007; Riessman, 2007).

As human beings, our lived and told stories become important ways that we fill our world with meaning and engage in relationships with one another to build lives and to build

communities (Clandinin and Rosiek, 2007). My narrative beginnings, as presented in Chapter One, briefly explored my identities within this research as a mental health clinician, as a gay male, and as a survivor of childhood sexual trauma. But a further identity, that of creative writer and storyteller with a great passion and respect for the written word, was what drew me in particular to narrative research and ultimately to narrative inquiry as design and methodology. Story has always served a crucial role in my life, and has provided me a source of knowledge, a means of expression, and at times an escape from difficult realities. Narrative inquiry and its fundamental beliefs in the power of story, story as knowledge, and knowledge as relational provided me with connection and linkage I needed between creative writer and academic writer, and therefore allowed me to position myself as a researcher and creative writer, and as an academic and a storyteller. It is for these specific reasons that I position this study within a narrative inquiry paradigm.

Research Design and Methodology

This qualitative study centered on a narrative inquiry research design and was theoretically drawn from parts of social constructionism, postmodernism, feminist and queer theory. It critically used a discourse analysis to explore themes related to social constructs of masculinities in our world today and how those constructs and the belief systems that surround them might impact male survivors of early childhood sexual trauma. A qualitative research design was chosen as it involves an interpretive approach to the world through studying things in their natural setting while attempting to make sense or interpret phenomena in terms of the meanings that people bring to them (Denzin & Lincoln, 2017). Qualitative research allows for a final research report that includes the voice of participants, the researcher's reflexivity, complex

descriptions and interpretations of the problem, and a contribution to the literature or a call for change (Creswell, 2013; Willig, 2013).

For this research, I have aligned my understanding of narrative inquiry primarily with the writings of Jean Clandinin and Michael Connelly (2004), whose methodology and approach are informed by the work of philosopher John Dewey (2008), and in particular his theory of experience. Dewey (2008) believed in an experiential continuum, where experience takes up something from those experiences that have gone before and modifies in some way the quality of those experiences that occur later. In other words, the way we live, learn, think, and remember is rooted in experience and in how we make sense of experience. All experience is grounded in stories - in how we tell those stories and in how we challenge them. Through conversation, we form and reform our life experiences by creating and recreating meanings and understandings, and therefore constructing and reconstructing our realities and ourselves (Brown & Augusta-Scott, 2007). Narrative inquiry, or "the study of experience as story... is first and foremost a way of thinking about experience" (Clandinin & Connelly, 2004, p. 375). It serves as a "collaboration between researcher and participants, over time, in a place or series of places, and in social interaction with milieus" (Clandinin & Connelly, 2004, p. 20). In a narrative inquiry, the storying of experience becomes both the phenomenon under investigation and the method of research (Pinnegar & Daynes, 2007). As Brown (2007) tells us, experience cannot be treated as absolute truth, conflated as authoritative, self-legitimizing and therefore uncontestable. Stories of experience are "interpretive, political, and contestable. Like all stories, they are multiple, fluid, changing, contradictory, and full of gaps: Like all stories, they are imperfect" (Brown, 2007, p. 192). In this research, I explore with participants their narratives as mental health therapists working with male childhood sexual trauma survivors. Their experiences, and the experiences of

those they care for as stories both lived and told, are the phenomena that I investigate, inquire within, and tell. I chose to follow the Clandinin and Connelly approach to narrative inquiry because it aligns with my own epistemological view of story as experience and experience as relational, with story as a way of being in the world and a way of knowing one's self and of knowing others. Clandinin and Connelly (2004) centred much of their research on teaching and teacher knowledge and worked in exploring and understanding a myriad of experiences in school systems. Their narrative inquiry methodology has also had interesting implications in recent years in nursing research by providing new and rich knowledge and understanding in the patient/doctor relationship (Wang & Geale, 2015). Their approach seemed highly relevant to the study at hand and works influenced by it mirrored what I hoped to accomplish through this study about survivors, treatment approaches, healing processes, and the gendered nature of male childhood sexual trauma.

As a method of narrative analysis, Clandinin and Connelly (2004) describe use of a three-dimensional narrative inquiry space to study experience. Understanding a three-dimensional narrative inquiry space – experience bound by temporality, sociality, and place - and what it means to live and work within that space over the course of a study is something that makes it unique and sets narrative inquiry apart from other forms of qualitative research (Clandinin and Connelly, 2004). Three-dimensional narrative space – temporality, sociality and place -are drawn from a Deweyan perspective that distinguishes experience from what people live through every day versus and an experience which has specific boundaries. An experience is therefore bound by temporality (it has a starting point and has, or will have an end point), sociality (it occurred in relation to one's self or others), and place (it happened somewhere) (Clandinin & Connelly, 2004). As noted earlier, stories of experience and the knowledge that arises from them are

embedded within history, context, and culture (Berger & Luckmann, 1966). Therefore, stories of experience are interpretations of events situated within historical, cultural, and temporal contexts.

Situating and analyzing each story within this three dimensional inquiry space, narrative inquirers are called upon to adopt a multidimensional and interdisciplinary lens; to treat the story or narratives as a whole; to pay attention to form, content and context of narratives; and lastly, to become participants in the understanding and construction of reality (Clandinin & Huber, 2010). Attending to temporality – past, present and future -means thinking about the ways people make links between various experiences, how researchers and research participants write themselves into and out of stories, and how narratives are created. Temporality presented itself as participants reflecting on past experiences with trauma survivors and telling their narratives. It also involved participants reflecting back on earlier times in their career, describing how they came to the work, what theories and principles guide them, and how that might have changed or evolved over time. Within temporality, movement occurs as reflections into the past and future, but also as lateral movements within the present (Clandinin & Connelly, 2004). This continuous nature of experience allows narrative inquirers to look not only to the present of experience, but also to its past by looking backward and to its future by thinking forward (Clandinin & Connelly, 2004).

The sociality dimension of narrative inquiry – the personal and the social - is based on the idea that people are always in personal and social interaction with both situations and experiences (Clandinin & Connelly, 2004). Personal conditions may include hopes, desires, and beliefs of both participants and researchers, while social conditions might include the environment and people who inform a person's context (Clandinin & Rosiek, 2007). People experience things either alone or with others, but as noted earlier, experience itself is always in

relation to others. Sociality is the looking "inward to thoughts, emotions, and moral responses and outward to events and actions" of the participants and researchers (Clandinin et al., 2018, p. 167). Therefore, attending to relationship within a narrative inquiry means considering how people's interactions with others shape their experiences. Within a narrative inquiry, the relationship between researcher and research participant becomes an especially important context for exploring experience. Clandinin and Connelly (2004) tell us that "inquirers are always in an inquiry relationship with participants' lives. We cannot subtract ourselves from relationship" (p. 480). The third dimension of experience is place (Clandinin & Connelly, 2004). According to Clandinin and Rosiek (2007), every experience takes place in specific, concrete, physical, and topological boundaries of place although a narrative understanding of place goes beyond the physical or geographical concept of place to a landscape of experience. Thereby, situating a narrative inquirer within the landscape of where experience takes place enables the inquirer to move in four directions (i.e., backward, forward, outward, and inward) from that particular place (Clandinin & Connelly, 2004). Stories lived and told serve as conceptual places where experiences occur, and it is within these conceptual places that we meet and live out the context of our lives (Clandinin, & Connelly, 2004).

Attending to the commonplaces of experience (temporality, sociality, and place) is crucial in order to gain a holistic understanding of experience (Clandinin et al., 2018; Clandinin & Huber, 2010). Experience is "always more than we can know and represent in a single statement, paragraph, or book. Every representation therefore... involves selective emphasis of our experience" (Clandinin & Rosiek, 2007, p. 39). A story might serve as a brief snapshot of an experience, incomplete but framed by temporality, sociality, and place. Narrative inquirers view this incompleteness as a strength rather than as a limitation in narrative inquiry. If a story can

only tell a partial tale of experience, then storytelling becomes an act where situated knowledge and subjective experiences combine to create multifaceted understanding (Connelly & Clandinin, 2004; Pinnegar & Daynes, 2007).

This study also employed a discursive analysis in understanding and interpreting the narratives of participants as therapists working with young male survivors. As discussed in an earlier chapter, a Foucauldian notion of discourse holds that discourse is a socially and culturally constructed representation of reality (Foucault, 1980; 1991). Discourses are interactions expressed in language that enable people in social groups and societies to build up a shared understanding of the meaning of behaviour (Fairclough, 2013; Payne, 2000). Dominant social discourses can shape therapeutic work in terms of influences of both clients' stories of themselves and therapists' interpretations of those stories (Brown & Augusta-Scott, 2007). Taken for granted and dominant accounts of mental health, trauma and violence are often constituted from larger organizing resources that rely upon notions of truth, knowledge, power, gender, experience, the self, and identity (Brown, 2007). Stories and experience then become essentialized, which is problematic in that it does not consider the impact and influence of social and historical forces.

Discourse governs what is possible to talk about and what is not and the taken for granted rules of inclusion and exclusion, and as a result it produces and reproduces both power and knowledge simultaneously (Foucault 1980, 1991). Foucauldian discourse analysis seeks to examine power and knowledge through the meaning-making that has been constructed through discourse. It uncovers how and why some categories of thinking and lines of argument have come to be generally taken as truths while other ways of thinking or being are marginalized. It asks questions such as what is being taken as truth, how that knowledge of truth came to be, and

how meaning is constructed, including what evidence is used and what is left out (Brown & Augusta-Scott, 2007; Foucault 1980, 1991). In considering unhelpful stories and preferred alternatives, such as those that arise from social conversations and culturally available discourses, we need to consider the relationship between knowledge and power. Central to the deconstruction of stories is the notion that through discourse, knowledge and power are joined (Brown, 2007; Foucault, 1991). Using Foucauldian analysis means questioning how power happens and how it operates and functions. It also considers that strategies and techniques that sustain it (Brown, 2007). Through data analysis, I sought answers to questions from the research texts such as: how has this come to be? What is being represented here as a truth or as a norm? What alternative meanings are ignored or left out? What interests are being mobilized or served? What is allowed and what is disallowed? This work will take a Foucauldian discourse analysis approach by seeking to destabilize some of the myths and assumptions around male experiences with early childhood sexual trauma, and uncover the ways in which dominant discourses might exclude, discount, marginalize, underrepresent, or oppress the voices of young male survivors.

In this inquiry, the stories of trauma therapists add a unique perspective to the landscape of therapeutic work with young male survivors of sexual trauma by critically examining myths and assumptions. It also serves to locate socially available discourse for male survivors and considers counter-stories and different narratives in understanding the experiences of childhood sexual trauma for young men and boys.

Participants and Sampling

Ethics approval to begin this study was sought through the IWK/NSHA Ethics Approval Board, and granted in September 2017 for one year, and has since subsequently been renewed in 2018 and again 2019 for the remainder of the project, with a plan completion of Spring/Summer

2020. Recruitment began in early December 2017, in the hopes of recruiting eight to twelve participants who met criteria in having provided treatment to young men and boys impacted by early childhood sexual trauma. Because there is so little research specific to male early childhood sexual trauma experiences, I hoped to recruit at least ten to twelve participants so that I might develop insight from a number of different perspectives. In the end, the participant sample consisted of twelve mental health practitioners who currently or had previously worked in some capacity as a clinician treating male youth under the age of 19. Three other therapists were interested initially in participating but, due to scheduling issues, were unable to commit.

Participants were recruited through purposive and snowball sampling, first from within IWK Mental Health and Addictions and Nova Scotia Health Authority (NSHA) Mental Health and Addictions programs, and later by word of mouth through mental health practitioners working privately in the community. I initially contacted managers of mental health and addictions services through the IWK and Nova Scotia Health Authority via email and provided a brief description of my proposed research. (This recruitment e-mail is included in the appendices section of this study). I received a response from three managers from the IWK and two from NSHA who agreed to distribute my recruitment email through their teams. I also searched through the Nova Scotia College of Social Workers (NSCSW) registry and the Association of Psychologists Nova Scotia (APNS) database and sought out mental health clinicians who worked in the area of male childhood sexual trauma, and contacted approximately ten clinicians to gauge potential interest in participating in the project. During the first few interviews, participants gave suggestions of therapists working in the community for the study who might be interested in the project, so I reached out via e-mail with the recruitment e-mail to those names provided. By late December 2017 I had confirmed and scheduled my first six participants, derived from both the

purposive and snowball sampling described above, and interviews began in mid-January 2018. By early April 2018 I had interviewed six participants with plans to interview six more, but as noted three participants dropped out of the study due to prior work commitments and challenges with scheduling, and so I continued to seek out further participants through word of mouth following interviews until I was able to confirm and schedule a dozen participants in total. Interviews for all twelve participants were completed by early June 2018.

At the time of recruitment, I was serving as a part time social worker/access navigator with IWK Central Referral and a casual youth care worker with Children's Intensive Services, both programs within IWK Mental Health and Addictions Services. I purposefully avoided recruiting in either of those work areas as I did not want potential participants to be current colleagues who might potentially feel pressured or obligated to participate upon receipt of a recruitment email. As there was potential to have some familiarity with other colleagues within the IWK Health Centre or NSHA, I encouraged managers to forward the recruitment email or post a copy of it in their care area and allowed it to serve as an explanation of the study. This email also provided participants the option to connect with either myself or the study supervisor, Dr. Catrina Brown, if interested in obtaining further information. While managers did distribute recruitment emails throughout their teams, I did not seek their direct help in advocating for active recruitment, as I wanted participants to learn about the study from the e-mail or posting and decide of their own volition whether they wanted to participate versus any implication of potential coercion or duress through being directly encouraged by their supervisors to participate.

Information and Consent Process

Participants who expressed interest in being interviewed were then provided a detailed understanding of the purposes of the study, and I sought out written informed consent specific to being interviewed and audiotaped, and noted that this consent will extend to any future publication of the study should that present as a possible outcome. Signed consents were completed at the start of each interview. Participants also had the option, at any point during the process of their involvement in this study, of fielding questions or comments through my supervisor Dr. Catrina Brown rather than myself as principal investigator, and were informed of hat option at the start of recruitment and prior to each interview.

Data Collection

Between late December 2017 and early June 2018, I conducted twelve semi-structured qualitative interviews, using open-ended questions and queries followed by occasional guided prompts. Interviews were recorded on a personal voice recorder device, and electronic files stored on a personal computer used solely for this project, with both computer and files password protected. The interviews lasted anywhere from one hour and fifteen minutes to nearly three hours in length and were rich with robust, detailed stories and perspectives of participants' work with young male survivors. Participants discussed at length the ways they worked to both challenge and mitigate the harmful impacts of dominant constructs of masculinity in their work, how they seek to combat stigma, and how they tailor their approaches to meet the distinct needs of male survivors. Using a narrative inquiry approach allowed for a rich and detailed description of these experiences, and furthermore provided space for an exploration of the meanings that the research participants derived from their experiences, both as practitioners and from within their own lived experiences.

Data Analysis

In this particular study, the stories told are reconstructions of another's experience in terms of the therapist relaying client narratives, and therefore encompass interpretations of stories within stories, all of which impacts on how the stories are told, which stories are told and which are excluded, and how they are both presented and in turn interpreted. They do not represent life as lived but serve as representations of those lives as they were told to the participant as therapist, and later relayed to myself as researcher. According to Bruner (1986), knowledge gained this way serves to bring together layers of understandings about a person, their culture and how they may have created change through their struggles to make sense of the past and create meanings. It is therefore situated, transient, partial and provisional, and is characterized by multiple voices and different perspectives, truths and meanings (Bruner, 1986; Riessman, 2007).

Data analysis in this study was considered through multiple lenses - first through a descriptive analysis and exploration of the detailed narratives of participants experience, then through the use of thematic analysis which served to extend the detailed narratives of experience provided by participants toward understanding and exploring more universal constructs related to trauma experiences and masculine identities and then key texts re-examined through a Foucauldian discourse analysis. Foucauldian discourse analysis allows for a means to consider and think differently about predominant truth claims and challenging taken for granted knowledges. Chase (2005) encourages multiple modes of analysis and suggests that limiting a narrative study to a unilateral mode of inquiry may serve to inhibit the richness of data and preclude more universal theorizing or application.

The process of interpretation or analysis in narrative inquiry involves "many hours (of) reading and re-reading field texts in order to construct a chronicled or summarized account of what is contained within different sets of field texts" (Clandinin & Connelly, 2004). During the initial stages, the interviews were first transcribed verbatim from the audio recorder within 24 to 2 hours of each interview. It was important to complete the transcriptions myself as I believed that would allow me to immerse myself into a participant's narrative and provide me with a richer understanding and deeper insight. I also believed in setting a goal to transcribe as soon as possible following each interview might allow me to better recall the conversations from my memory and capture the intended meanings. However, as Clandinin and Connelly (2004) tell us, meaning in narrative inquiry can often become distorted and cannot ever be fully captured. Following transcription, I studied the raw narratives in their initial rough drafts, and from there I moved toward focusing on and analyzing individual experience. This served as a means to deconstruct the life stories of the participants while attending to the multiplicity of meanings within the interactions between researcher, storyteller (in this case, participants) and listener or audience.

Following transcription, I created a storied draft of each participant's individual narrative (these narratives are described in the following chapter) using a blend of both my voice as narrator and the narrative stories of the participants. From this perspective, I sought to explore and analyze each narrative "temporally, spatially and in terms of the personal and the social" (Clandinin & Connelly, 2004, p. 89) while attempting to understand and retain the participants' intended meanings within the context of their experiences. Clandinin and Connelly (2004) suggest three specific analytical tools – broadening, burrowing, and storying and restorying. Broadening refers to analyzing the transcript interview while reflecting on the literature review

and field notes to better understand participants' values and beliefs within the social, cultural, and historical milieu the research takes place. Burrowing is to focus on specific details of the data, with attention paid to participants' feelings, understandings or dilemmas and impacts of events and happenings. Storying or restorying comes after broadening and burrowing, where the researcher considers ways to rewrite the participants' story to move their experience to the forefront. From there, the narrative interviews were coded by laying each narrative side by side and exploring the data horizontally, looking across narratives to establish "patterns, narrative threads, tensions and themes" and as such, create interim texts of descriptive data (Clandinin & Connelly, 2004, p.132), while also forming a foundation for the next stage of thematic analysis.

Originally, I had planned only to compare and contrast these themes between this stage of analysis and the next and not use the descriptive accounts in the final report (in an earlier and in less cohesive draft I chose not to use them and focused instead on thematic analysis). However, in analyzing and re-analyzing the data, I realized that by not using the texts I was losing important aspects of the participants' rich voice of experience and perspectives. A large part of the insights gained around this phenomenon came from hearing participants own narratives — how they came to this work, what motivated them day to day, how they identified themselves and how their identity was shaped and constructed — and to not include it, in their own words, devalued their contribution to the finished work. The aim of the descriptive analysis was to build brief narratives of the experiences and lives of the trauma therapists and their descriptive retelling of their work with young male trauma survivors, and in so doing, the narratives sought to describe how participants constructed their personal and professional identities as shared in their storied experiences. These narratives are included in Chapter Five: Narrative Accounts.

Following the descriptive analysis, thematic analysis sought to discover and explore common themes. Braun and Clarke (2006) suggest that the analytic process should ideally evolve from description where the data is summarized according to patterns, with focus placed on the significance of the patterns and their broader meanings and implications. Thematic analysis was considered within the framework of narrative analysis by drawing on Braun and Clarke's (2006) techniques. I familiarized myself with the data through transcription and rereading several times and made comments regarding initial possible ideas related to my interpretations of themes. I generated initial codes across transcripts and field texts, looking for interesting data related to research questions and objectives and then connected it back to codes. Codes were collated as I searched for themes, and from there I gathered relevant codes and data under each emerging theme, with ongoing written commentary in journals to help understand content and context. I then reviewed themes for broader themes across the data, and then collapsed and combined themes or considered sub-themes as I defined and named final themes and considered the overall story the analysis told. From there, I considered vivid or compelling examples from across the research texts that would support themes I highlighted and focused upon and inserted those into the final written report. In this way, the thematic content was considered from a postmodern lens in order to "hear the operation of broader social discourses shaping that person's story of their experience" (Clandinin & Rosiek, 2007, p. 55). These findings will be presented in Chapter Six: Interpretive Findings.

A final step to the analysis was to apply a Foucauldian discourse analysis to the narrative accounts and examine elements of power and knowledge through the meaning-making constructed in and as discourse. Analyzing data through this lens uncovers how and why some categories of thinking and lines of argument have come to be generally taken as truths while

other ways of thinking or being are marginalized. In applying discourse analysis to this study, I ask questions of the data, such as what is being taken here as truth or norm, how has it come to be, how is meaning constructed, what evidence is used, and what is left out. This analysis will be discussed as part of Chapter Seven: Discussion and Analysis.

Rationale for Analysis

Thematic analysis allowed me to paraphrase and summarize a large amount of data in relation to the research questions, with a goal of capturing descriptions of participants' thoughts, feelings, behaviours and actions as succinctly and accurately as possible (Braun & Clarke, 2006). Through the process of recoding I was able to observe and uncover broader social discourses that influenced these stories of experience. The descriptive analysis and the use of broadening, burrowing and re-storying allowed entry into the three dimensional narrative space (Clandinin & Connelly, 2004). This allowed me to inhabit participants' stories and understand them contextually in terms of time, space, and in relation to the personal and social. Discourse analysis was always central to this work in considering the relationship between masculinity as a construct and toxic masculine culture as its most extreme form, and the experience of childhood sexual trauma. As an analytic tool, discourse analysis allowed me to examine key discourses through which masculinity, heteronormativity, homophobia, misogyny, and sexual trauma are understood. It also provided a means of understanding the relationship between these discourses by identifying tensions, points of resistance, contradictions, and silences (Willig, 2013). As I considered each of these methods individually, thematic analysis seemed almost too mechanical in its application, and the descriptive analysis, while attractive in its creativity, appeared to lack sufficient scientific rigour on its own. Discourse analysis, while a critical aspect, was highly specific to the meaning, knowledge and beliefs produced through language and social practices.

By combining methods, I gained entryway to these rich and robust stories of experience that in the end allowed for themes to emerge and avenues to explore that might otherwise have gone undiscovered. Ultimately, it also allowed for more universal theorizing and application of a phenomenon – male childhood sexual trauma in relation to the construction of masculinity - that is both poorly understood and sadly under-researched.

This section described the three-part approach to data analysis, including descriptive analysis (narrative accounts), thematic analysis, and discourse analysis. It also provided a justification for using multiple lenses in analyzing the data. This analyses will be incorporated throughout the next three chapters following this section: Chapter Five: Narrative Accounts; Chapter Six: Interpretive Findings; and Chapter Seven: Analysis and Discussion.

Touchstones of Narrative Inquiry as Validity and Rigour

Clandinin (2013) describes a touchstone as a "quality or example that is used to test...
excellence or genuineness" (p. 169). There are twelve touchstones that a quality narrative inquiry
must meet, and that researchers can use to judge the validity and rigour of the study (Clandinin,
2013). Touchstones of narrative inquiry include: recognizing and fulfilling relational
responsibilities; being in the midst; having a commitment to understanding lives in motion;
negotiating relationships; narrative beginnings; negotiating entry to the field; moving from field
to field texts; moving from field texts to interim and final research texts; attending to
temporality, sociality, and place; interacting with relational response communities; explaining
justifications (personal, practical, and social); and attending to multiple audiences (Clandinin,
2013, p. 212). According to Clandinin (2013), narrative inquirers must attend to these
touchstones if they are to engage in a narrative inquiry that is sound in its ethical and
methodological components. In the following section I will address each of the touchstones in

the context of my inquiry and explain the methods of fieldwork that I used while conducting my research.

Recognizing and fulfilling relational responsibilities

Within the researcher/participant relationship it is important for narrative inquirers to negotiate expectations, time constraints, next steps, and outcomes (Clandinin & Rosiek, 2007). By attending to relational responsibilities, we also attend to the emotional needs of participants. In this study, I attended to the emotional needs of my participants by respecting their time constraints, assuring them confidentiality, reviewing mental health supports available to them following our discussions, making myself available for future debrief, and acknowledging and honouring their important contributions to this work.

Being in the Midst

When I began this research, I was living stories of my life as an MSW candidate, a mental health social worker, a trauma advocate, a partner, a friend, and a gay man, among many other things. My participants were also living their stories of being mental health therapists, managers, partners, teachers, parents, sports enthusiasts, writers, painters, and so on. When we met, we were each "in the midst of living and telling, reliving and retelling, the stories of the experiences that make up lives, both individual and social" (Clandinin & Connelly, 2004, p. 20). A researcher begins an inquiry in the midst, progresses in the midst, and concludes it in the same way.

Having a Commitment to Understanding Lives in Motion

Stories are always unfolding and partial and can never tell a complete story of experience. Within a narrative inquiry, Clandinin and Caine (2013) tell us that:

There is no final telling, no final story, and no one singular story that we can tell. While this is troubling to researchers who rely on the truth or accuracy and verifiability of data, it is opening the possibility of narrative inquirers to continuously inquire into the social fabric of experience and not lose sight that people are always becoming (p. 176).

When we began our research relationship, and when we ended it, the research participants and I were each living complex and busy lives and attending to those complexities and the continuity of fluctuating experience guided this narrative inquiry.

Negotiating Relationships

Narrative understanding of what it means to live alongside each other refers to how we negotiated what our research relationship would look like, beginning with informed consent, to where and when we might conduct the interview, to what was comfortably shared (Connelly & Clandinin, 2000). While some aspects of negotiation were explicit, others emerged organically during our interactions, such as some participants opting to share aspects and insights into their personal lives.

Narrative Beginnings

Narrative beginnings are autobiographical accounts of the "personal, social, and political contexts that have shaped our understandings" (Clandinin, 2013, p. 55). They are reflective and help situate the researcher in relation to the topic, the participants, and the literature, and are revisited throughout the inquiry (Caine et al., 2013). Revisiting narrative beginnings is important, because as a researcher engages in narrative inquiry, perceptions and understandings shift.

Clandinin (2013) explained, "Readers often understand an inquiry in more depth when they are able to see the researcher's personal justification in the research texts" (p. 36). By offering some context, narrative beginnings invite the reader to consider from where the inquirer was working

and allows the reader to make judgements about how an inquiry relates to their own understanding of the world. For this inquiry, I asked myself: who am I in this narrative inquiry - Who am I as a man who wants to study experiences with and around male trauma? Who am I as a mental health social worker? Who am I as an academic researcher? Who am I as a male childhood sexual trauma survivor? These questions helped me to write my narrative beginning to this research, which I have shared in Chapter One. It helped me to understand who I am in relation to those participants that I lived alongside and to those fellow survivors for whom I chose to dedicate this thesis work.

Negotiating Entry to the Field

Clandinin and Huber (2010) describe the research field as where research takes place. This may be in reference to a specific three dimensional or geographic space, but it does not have to be. In using a narrative understanding of experience, experiences take place within stories, and thus the field can mean a conceptual place where researcher and participant meet and engage in the narrative inquiry process (Clandinin & Huber, 2010).

Moving from Field to Field Texts

The term "field texts" is used by narrative inquirers to differentiate the various texts and modes of communication that come out of narrative inquiries from the research texts that are meant to be more objective and are called "data" in other methodologies (Caine et al., 2013). As Caine, Estefan, and Clandinin (2013) remind us, "narrative inquirers understand data as field texts that are to be experienced as they are lived and told as narrative compositions. Living is field text" (p. 579). For this narrative inquiry, I recorded the conversations that I had with my research participants on a small audio device, and by using it I hoped to more fully capture the thoughts and words they shared that reflected the nuances of our discussions and conversations.

This also allowed me to rely less on taking actual extensive notes during the interviews, and at times I was able to set notepad and pen aside and listen in more detail, which in turn led to participants seeming more engaged, relaxed and receptive in our conversations and interactions.

Beyond audio recordings, I used the following types of field texts: interview transcriptions, field notes, voice memos, emails, documents, and journal entries. Each of these are described by Clandinin and Connelly (2004) as important for interpreting field experiences. While I initially did not imagine that I would have accumulated such a depth and range of field texts, each became naturally imbedded into the process of my inquiry. For example, some participants provided documentation that explained the goals of their clinical areas, while others shared previous research articles they had written. Others conversed a number of times over email to share further thoughts on our interviews or elaborate on some point or argument they wished to elaborate. I incorporated all these different modes of communication as field texts in an effort to more fully understand the different dimensions of the participants' experiences.

While the transcriptions served as word for word records of the conversations that my research participants and I had, my research journal allowed me to reflect broadly on what I was thinking and feeling during our meetings. The reflections in my research journal helped me to provide much needed context to the transcriptions when I read them later. As I incorporated all the many modes of communication - transcriptions, emails, shared articles, and voice notes -with my research journal reflections, I shifted towards developing interim texts, which ultimately became the final research text that is this thesis work.

Moving from Field Texts to Interim and Final Research Texts

Within a narrative inquiry, interim research texts are narrative accounts of experiences from the field (Connelly & Clandinin, 2004). Field texts detail the experience of living alongside

participants, while interim texts serve as a researcher's attempt to make sense of the field texts as they relate to the larger research puzzle (Connelly & Clandinin, 2004). Interim texts are often shared with participants and are used to further the researcher-participant relational way of knowing. Final research texts may be traditional academic reports such as dissertations, journal articles, and books (Clandinin, 2013), and in my case the final research text will serve as my master's thesis. For this research, I shared written field notes and reflective comments with my participants at the end of our interview, and welcomed their involvement in making sense of what I had written. Sometimes they simply provided a quick approval, and other times this led to re-engaging in deeper conversation following the interview. Participants were provided with contact information and were welcomed to follow up with me at any time should they wish to elaborate on any part of the interview or add further thoughts. It was important to honour the stories that were shared and the important work these therapists provide. Part of how I cared for their stories was by trying to represent them in a way that was genuine, meaningful and respectful.

Attending to Temporality, Sociality, and Place

As discussed earlier, temporality, sociality, and place comprise the three-dimensional narrative inquiry space and must be attended to at every stage within a narrative inquiry (Connelly & Clandinin, 2004). The interviews I had with my participants began with their stories, and looked forwards, backwards, and sideways to their experiences as trauma therapists. In the field, the conversations and interactions that I had with the participants occurred in places, in offices, private homes and coffee shops. I also paid careful attention to how place was attended to in the stories that the participants shared, because where an experience occurred was often important to the telling of a story. Looking backwards on experience was reflective for

participants, and in that process, they storied themselves with insight that comes from living through varied experiences. Storied experiences carried our conversations from thinking backwards, to thinking forwards. For instance, participants reflected on their training in early stages, where perhaps there was less focus on trauma work specifically, and what sort of mentor and teacher they try to be today. Looking forwards allowed participants to speculate on things that might change in our approach to trauma work with young male survivors, and in that contemplation, they carried past and present experience with them as a way of informing their discussion.

Looking sideways is another way of that narrative inquirers attend to experience (Clandinin, 2013). Looking sideways takes the focus off of the obvious and shifts it to something that otherwise might be beyond our attention. Within these interviews, some participants talked about their experiences as parents and partners, and through this sideways looking, I was able to see aspects of their identities as trauma therapists more clearly. For instance, one of the participants described how being the mother of two boys sparked her desire to work with young men and boys, while another described being a parent of teenage girls with a diverse group of friends opened him up to a better understanding of gender fluidity and expression.

Interacting with Relational Response Communities

Relational response communities are people and places where a narrative inquirer can turn to discuss their research and "often consist of people the researcher values and trusts to provide responsive, and responsible, dialogue about his or her unfolding inquiry" (Clandinin, 2013, p. 210). For this research, my response community primarily consisted of my research committee, but also came to informally be comprised of several others. My thesis supervisor Dr. Catrina Brown and thesis committee member Dr. Marion Brown and I met on several occasions

to debrief and discuss processes, theory, methodology, and content and their offered feedback proved invaluable to the final research text. We also regularly corresponded over email as a means of providing updates as work progressed. Their contributions also helped ensure I was navigating my research in a safe and ethical way. My social work candidacy supervisor Coleen Flynn provided lots of opportunity for debrief around processes and offered tremendous emotional support. My partner Shawn was a constant source of strength and encouragement throughout this entire journey, and I owe him a huge debt of gratitude for his unwavering support and thoughtful contributions. I was also fortunate to have a number of peers and friends who had genuine interest in research and provided much needed encouragement along the way. These relational response communities helped me to better understand the importance of relational ways of knowing, and in doing so strengthened my appreciation for narrative inquiry.

Explaining Justifications – Personal, Practical, Social

It is important to offer personal, practical, and social justifications for engaging in narrative inquiry as it fuels both our passion and dedication to the work (Caine et al., 2013). In personally justifying a narrative inquiry, researchers create autobiographical narrative beginnings and include parts of them at the beginning of final research texts (Caine et al., 2013). Practical justifications can emphasize why an inquiry is important and offer a deeper understanding of experience (Caine et al., 2013). Goals of this narrative inquiry were to give voice to the silent phenomenon of male childhood sexual trauma and provide a greater understanding of the lives of trauma therapists and the complex work that they do. Social justifications can be thought of in terms of how an inquiry can impact social policy, how it may contribute to social change, and how an inquiry can contribute to new discipline knowledge (Clandinin & Huber, 2010). It is my hope that sharing stories of trauma therapists and young male survivors can help facilitate new

perspectives, new ways of understanding, and contribute to the literature around the poorly understood phenomena of male childhood sexual trauma.

Attending to Multiple Audiences

Clandinin (2013) tells us that final research texts should be constructed with the intended audience in mind. Because what is shared is ultimately a small part of the research texts, it therefore becomes important that the researcher identify the process through which particular stories were chosen to be highlighted or presented (Clandinin & Huber, 2010). In this study, I attend primarily to the relational experiences of my twelve participants as trauma therapists, and chose to emphasize stories from their individual narratives that I believed helped to explore their identities and experiences, highlighted the phenomenon of male early childhood sexual trauma, and illustrated the unique needs of young male survivors.

Ethical Issues

Ethics approval was received through the IWK/NSHA Research Board in September of 2017, and, as it became necessary due to timelines, ethics approval was renewed in 2018 and again in 2019, until the completion of this study, projected for spring/summer 2020. The difficulties inherent in qualitative research can be alleviated by awareness and use of well-established ethical principles, specifically autonomy, beneficence, and justice. Under autonomy, the researcher is responsible for ensuring that all study participants are well-informed about the purpose of the research they are being asked to participate in as well as their right to withdraw from the study at any time without explanation or reprisal. As part of this study, I provided participants a detailed understanding of the purposes of the study and sought out written informed consent specific to being interviewed and audiotaped and noted that this consent will extend to any future publication of the study should that present as a possible outcome. In

maintaining beneficence, the idea of doing good for others while preventing harm, I ensured participants understood, to the best of my ability, any foreseeable risks to being a part of research as well as any benefits. One foreseeable risk was the emotional nature of some of these discussions, bringing up past difficult or painful thoughts or memories within challenging contexts. In preparation for such issues, I asked about strategies around self-care and maintaining health and well-being, and reflected on those during and at the end of each interview. I made myself available for debriefing immediately following the interview and for the duration of the study if warranted by participants.

Furthermore, in respecting the confidentiality of our subjects, I kept all recorded and written material stored in a locked cabinet in my place of residence. I will destroy all documentation, written and recorded, using IWK/NSHA research protocol standards, after the project is complete and thesis is submitted for final evaluation. I have kept the identities of the participants anonymous throughout our study and use pseudonyms of my choosing in transcribed interviews, research texts, and in the final written report. I demonstrated principles of justice by avoiding any exploitation and abuse of participants, using fair procedures and outcomes in the selection of participants, and noting in the final product their significant contributions to the study. I reminded participants of their right to withdraw from the study at any point in time, and to have any collected data removed up to the point of thesis submission, projected for spring of 2020.

I have used reflexivity throughout the thesis work and separately journaled my thoughts, feelings, and experiences. Reflexivity refers to the researcher's critical self-reflections and impressions of the story, and any personal meaning that it might have for them (Reissman, 2007). Because narrative inquiry is such a reflexive and reflective methodology, as researcher I

spent time self-reflecting before, during and after each participant interview (Clandinin & Connelly, 2004; Clandinin, 2013). Furthermore, I recorded my personal feelings and responses throughout the duration of the study by way of detailed journal entries as a means of documenting my beliefs, values, thoughts and feelings which could influence the interpretations of the field texts. As Riessman (2007) notes, reflexivity through journaling fosters critical self-reflection and awareness of how the research was done and any critical decisions made along the way.

Through reflexivity I have been able to observe, maintain and comment on the impact I have on the research and the impact it has on me as researcher. I have remained vigilant and aware of any unconscious biases, and attempted to employ ethical mindfulness as a means of being "consciously aware of the researchers' values and routinely asking moral questions of ourselves, our practice, and our professional relationships" (Danchev & Ross, 2014, p. 16). Furthermore, I have managed my own self-care and well-being, and have been fortunate to have close support system at hand throughout this process. By following the guidelines set forth through the IWK NSHA Research Board and the Dalhousie University Faculty of Graduate Studies, I believe that the research was conducted safely and ethically.

Conclusion

In this chapter, I have provided a detailed overview of the methodology and research design in this study. I reviewed research questions and objectives, and I described the theoretical framework of social constructionism and postmodernism and the influence of feminist and queer theory on this work. I described the narrative inquiry research design and methodology, and provided rationale for why it was chosen. I discussed the three-part analysis stage, from descriptive, to thematic to discourse. I outlined the methods used for recruitment, participant

sampling, and data collection. I introduced and addressed twelve touchstones by which the validity and rigour of this narrative inquiry can be measured. Finally, it also provides an overview of how confidentiality and anonymity was maintained as well as a framework for ethical considerations which I believe have been maintained throughout this work.

Chapter Four: Narrative Accounts

In Chapter Four, the participants' voices are heard, and stories of their experience and practice with male sexual trauma survivors are presented. While analyzing the research texts, I attended to Clandinin's (2013) belief that in narrative inquiry the participant's voice is the most influential. I engaged in this research by thinking narratively and by attending to Clandinin and Connelly's (2000) commonplaces of three-dimensional narrative inquiry space of temporality, sociality, and place while constructing these research texts, a process that included looking forward and backward with participants, reflecting with them on the social and personal meaning of their experiences in practice, and recognizing the different places and contexts in which their stories are situated. Stories of experience provide insight into the treatment of male sexual trauma survivors and the exploration of discourses in which knowledge that surrounds it occurs. In this study, insight and exploration occurs through the meaning making participants bring to their experiences and of their clients' experiences. It also comes from the combined meaning making that comes from my shared understanding and interpretation of these stories with participants, from the understanding and interpretation as I researcher I make through navigating the final research texts and finally, from the meaning making, understanding, and interpretation that falls to the reader.

The purpose of this chapter is to provide the reader with as much of the rich textual detail and depth of the individual participant's experiences as possible that emerged during the study. Through these texts, I began thinking with their stories and the stories that emerged over the course of each of our conversations, which were co-constructed within the dimensions of temporality, sociality, and space. My hope is this chapter serves to honour the voices of those participants who have chosen to share their experiences in this inquiry and offer an entry point

into the lives of participants as therapists, and a means of exploration to those who read these narratives. I chose to end each narrative with a quote from the participant that was either a significant focus of our conversation or a theme that resonated following the interview. The following stories serve as accounts of working together (Clandinin, 2013) with the participants to understand their individual experiences in the context of their therapeutic practice with male sexual trauma survivors.

Descriptive Overview of Participants

All twelve participants in this study were considered primarily clinical therapists in their individual workplaces. All participants currently either worked directly or indirectly in a clinical and therapeutic role with a wide range of youth and adults, although two participants admittedly had to rely mainly on past or previous experiences with young male survivors to guide their conversations.

Once participants responded to the recruitment e-mail and expressed interest in possibly participating in the project, I arranged for a brief phone chat to introduce myself as a researcher and fellow mental health practitioner interested in the field of trauma work, and from that point scheduled the study interview. This introduction provided a means to position myself as an "insider" with "inside knowledge" to mental health systems, structures, and approaches, which I believed may have helped put participants at ease with the interview process due to our shared understandings of complex mental health systems. Ten of the interviews were conducted at the participants' own office space, all within Halifax Regional Municipality, which I hoped might foster and nurture a comfortable and safe environment for participants to partake in these conversations. This also had the added bonus of allowing me to inhabit the physical space of these therapeutic settings as a client would experience it. Another interview was conducted in a

private space at a local coffee shop in Halifax, and the last at my private work office in Dartmouth, Nova Scotia.

Overall, eight of the participants identified as cis-gender males, and four as cis-gender females. Participants ranged in age from late 20s to early 60s. Ten of the participants were Caucasian or of European descent, one was African Nova Scotian and one was South East Asian. Four of the participants self-identified as members of the LGBTQ+ community. Four of the participants held clinical roles within the IWK Health Centre, three worked within the Nova Scotia Health Authority or NSHA, five were private practice clinicians, and one worked within a community agency. Two of the hospital-based participants had dual roles as managers/team leaders along with their clinical responsibilities, while four did some degree of private practice counselling aside from their hospital-based work. Six of the participants held Master of Social Work Degrees, five held Master of Education in Counselling, one held a Doctorate of Psychiatry and one held a Bachelor Degree in Social Work. The information is displayed on the following page, in Table 1, using pseudonyms to describe participants. Some demographic information, such as specific practice areas, was excluded in order to protect confidentiality.

Participant	Age Range	Sex	Role	Education
Bryce	Mid 30s	M	Clinical Therapist	Master of
			(Private Practice)	Education
				(Counselling)
Daniel	Late 30s	M	Clinical Manager	Master of
			(Mental Health) &	Social Work
			Clinical Therapist	
			(Private Practice)	
Joanna	Mid 50s	F	Clinical	Master of
			Therapist/Trainer	Education
			(Private Practice)	(Counselling)
Peter	Late 50s	M	Clinical Practice	Doctor of
			Leader	Psychiatry
			(Mental Health)	
Susan	Early 60s	F	Clinical Therapist	Master of
			(Health & Private	Social Work
			Practice)	
Kate	Mid 40s	F	Social	Master of
			Worker/Navigator	Social Work
			(Health)	
Nathan	Early 50s	M	Clinical Therapist	Master of
			(Private Practice)	Education
				(Counselling)

Gillian	Early 50s	F	Clinical Therapist	Master of
			(Mental Health &	Social Work
			Private Practice)	
Adam	Early 30s	M	Clinical Therapist	Master of
			(Private Practice)	Education
				(Counselling)
Ben	Late 30s	M	Clinical Therapist	Master of
				Education
				(Counselling)
Roger	Late 50s	M	Clinical Therapist	Master of
				Education
				(Counselling)
Shane	Mid 30s	M	Community	Bachelor of
			Outreach	Social Work
			Worker/Educator	

Bryce

I met Bryce at his self-described office space/music studio, a high ceilinged room with large sunny windows and an impressively large collection of musical instruments. I wondered if he incorporated music as part of his therapeutic work, and while he said he did not directly, he found for many clients being able to relate to a shared passion for music was often a "way in" – an introduction to therapy that helped put clients at ease, particularly men and boys he noted, and gave them "talking points to sort of ease into a conversation." In describing his career path that

led him to become a therapist today, Bryce described some early struggles in his life that, in the end, set him on a specific course: "Without sounding a bit like a cliché, I was living in a recovery house in Vancouver in my late teens speaking about my experiences, and a counsellor said 'you should be a therapist someday". And so with some support and guidance from that counselor, he followed the advice, and while in the early stages of recovery, he went on to complete a Bachelor Degree in Psychology, then later a Masters in Counselling, and today is working on a Ph.D. in sociology and anthropology.

Bryce is only a few years into his career as a private practitioner, but has already developed a bit of a specialization in what he calls identity and trauma work, primarily with LGBTQI2S+ populations, which developed out of an internship in Toronto. He described his approach to his work as eclectic, drawing on narrative and feminist principles and anti-oppressive practice, and centred on shame and innocence: "for me, it's about trying to understand the pervasiveness of shame in people's lives and how they've internalized that through their marginalization or their trauma or whatever".

When working with male trauma survivors, Bryce often starts with some open discussion about dominant cultural norms related to masculinity. He describes these norms as constantly shifting and changing, but ultimately still problematic for many, most notably he believed in restrictive emotional expression. Bryce spoke of male survivors who struggle to process trauma in their therapy sessions in very specific ways:

They are skilled at talking around it, but never about it. They have all these defense mechanisms: humor and laughing and all these things that kept it [the trauma] and their feelings always at a distance.

Bryce spoke of survivors struggling to come to terms with sexual identity issues and wondering if experiences of sexual abuse have somehow "made them gay". Many survivors also struggled with the belief that sexual violence only happens to women and girls. Therefore, because they "perceive the abuse as outside of the male experience, they are now left outside of it, feeling like maybe they're no longer masculine".

When asked to define masculinity in his own terms, Bryce said his "hopeful answer" was that he believed the concept of masculinity was changing in the Western world, and in particular in what he described as "progressive urban spaces". He saw that evolution in terms of "greater flexibility – flexibility around roles, gender expression, and range of emotional expression." However, Bryce described a very personal way that masculinity plays out in his life. Since moving to Halifax a few years ago, he's described making numerous female friends but struggles in making new male friendships. He believes part of that reason is that one of the ways he would frequently relate to other men – through sharing alcohol – is denied to him now:

I think — I think somehow it's the difference between 'do you want to go for coffee?' and 'do you want to go for a beer?' That's where masculinity is...I don't know where that is exactly, but this one's masculine and that was not...that's all I know. And it really reveals itself in that moment where — because of my history, and being in recovery — where I just don't have the capacity to ask a dude if he wants to go for a beer. So whatever that reveals about whatever stereotypes roles men are supposed to play - if that's activity and sports and beer and watching and doing things, and not talking about things and feeling things and having meaningful conversations on that level — then that's what masculinity means to me.

Bryce described how often male survivors are marginalized by wider society. When considering societal expectations around masculinity, he drew connections to capitalist society and notions of productivity, and how struggles with traumatic histories might impact:

If you're a guy on the margins in any way in society, then chances are you're there, in some ways, because you weren't producing in the same way that everyone else seems to be able to do with ease...So when I [as a survivor] end up on the margins, whether from the trauma itself or from my identity or orientation or race or, you know, the struggles ']that I have had...when I'm pushed to the margins, I misrecognize that as something fundamental or essential to what I am. All I I know how to do is to describe why I don't contribute to society in the ways that society is demanding of me.

Bryce believes the shame and blame that survivors express is related directly to three areas — worry related to sexual orientation, false beliefs that only women and girls are assaulted, and fear over being perceived as not masculine. Much of his work, as a result, is around unpacking and addressing these "untruths'. Bryce says he likes to spend time exploring the impracticality of self-blame, because he believes the shame and self-blame that male survivors experience contributes to reasons for non-disclosure, but furthermore to the invisibility of the phenomenon, as it "forces people to not talk about it and in that way removes the conversation from public discourse."

Also crucial to trauma work for Bryce is breaking down the stereotypes by exploring masculinity norms and considering how male socialization potentially interferes in the healing process. For some survivors, he says:

Masculinity itself is therefore part of the marginalization they are experiencing - in the sense of how they are subscribing to these traditional norms. And for those survivors, I

want to be sure to highlight this as a social problem. 'We're all in it, struggling with these competing messages. It's not just you. You're not alone'.

Peter

Peter and I met at his office in a busy high-rise downtown at the end of an even busier workday. Peter is originally from the UK, and has been a psychiatrist for more than 20 years, and prior to that worked for many years as a family doctor/general practitioner. Peter attributes his overall holistic approach today to those early years practicing medicine. He has always believed strongly in helping his patients build connections, noting that "people get their sense of worth from things that they find rewarding and in- in having some sense of belonging". The challenge, he says, is that people with mental health problems, "by virtue of either anxiety or depression or psychosis often don't easily kind of approach joining in or feel safe".

Trauma survivors he has worked with often suffer from substance abuse issues and have problems with anger and emotional regulation, and help seeking is almost always externally driven: "it's interfering with relationships with family, friends, work colleagues. Life is pretty bad, things are at their lowest, and people are pointing towards...you need to kind of seek some help or else".

Peter believed that men are generally not very well attuned to help seeking, and he says he observed that often during his time as a general practitioner. He says it was relatively rare to see a man enter his office unless it was a matter of some urgency:

It's women and children and elderly people. That you don't see a man between 20 and – like, maybe 60 years of age unless it's something severe or something significant has happened, like an accident...that to me really speaks to challenges in help seeking for men.

By nature of his role today as a psychiatrist, his patients are often those that are facing significant struggle. He works with a large cross section of people from different walks of life, all presenting with unique and individual needs. He says some are seeking to reflect and process their trauma or their struggles with addictions, while others are less comfortable exploring their past and are seeking more practical skills to use today, or are seeking assistance with medication efficacy. As a result, he says, he is constantly adapting his methods to meet their needs and their unique styles of interacting.

He sees experiences of being sexually abused or sexually traumatized for some young men as "a kind of interruption in their own natural or expected development of their sexual interest – something they have been cheated out of". Despite working with a fairly significant number of male survivors over his career, Peter has never had a patient disclose being victim to a female perpetrator. In all cases, survivors were abused by other males, and in most, males that were significantly older and often in a position of power or authority or trust.

Peter believed providing psychoeducation around physiological responses helps shift some of the critical self-thinking and helps "normalize in a sense or at least reduce the intensity of feelings around it [the body's response]". He described using a combination of cognitive behavioural therapy and acceptance and commitment therapy with a focus on strengths and resiliencies, while still paying attention to the areas that they feel are damaged or uncertain:

Different people seem to be able to kind of heal differently, or I guess- I guess maybe resiliencies are different. Other people, for a number of factors are often very unforgiving of themselves, and hold on to self-doubt and self-criticism. They have no self-confidence, they are left angry and they feel like they are damaged and nothing will fix them. And all of that – to me, it just serves to undermine people's full on level of participation in life.

Reflecting on how he perceives masculinity in the world today, Peter likened it to his own struggles, starting back in early childhood and raised in what he described as a "judgemental society". He recalled growing up in a strict, authoritarian family with a father who was a police officer who represented, in a way, 'this idealized form of dominant masculinity – strong, powerful, silent." He also went to school in Europe during at a time when sexes were segregated, and it was a generation, he said, where stereotyped roles of masculinity and femininity were expected:

It's 'what does a man do, what does a woman do?' It's when you are at work... at home... when you're out. It is about - what do you wear? As a man, how do you stand? What do you sound like? And how do you talk? With people, you know what's interesting is that it – that it is all so ingrained.

Intellectually he had always rejected those stereotypes, but at the same time battled with how ingrained these ideas were and how it shaped his own sense of self through the years. He described having two grown children today, and how their experiences in school were light years removed from his in terms of how diversity and cultural differences are more accepted. "Everything is OK, ever shade and color of person, every place on the sexual continuum, any identity, any interest, any preference – to them, none of it is ever a big deal."

Peter said he is always mindful of new and innovative ways to practice therapy. He saw the benefits, for example, in terms of engaging certain men in outdoor therapy techniques and experiential approaches as trauma treatment, but he thought it was most essential when working with men and boys that a focus be placed on talk therapy: I really think you do have to learn how to translate experiences into words ultimately, I really do see that as necessary to healing. So I do think that piece is required, even though it doesn't come easily to most men.

Joanna

Joanna works as a clinical therapist and clinical supervisor in a service area that primarily works with men who use violence. Joanna is in her early fifties and became a therapist ten years ago after graduating from Acadia with a Master of Education in counseling degree. We meet an early weekday morning in her office, and spend the first part of the interview talking about her "previous life" as an artist and teacher, and her strong interests in art history and art therapy:

I discovered early on a lot of research that connects domestic violence and sexual violence to art therapy because those are difficult things to talk about, so drawing about them and creating art was a way to express it- and I just loved the whole notion of that, of beauty and art coming out of it and bridging communication.

She says she brings her passion around art into her work nowadays by always being "flexible and creative and innovative, so that together with the client we can figure out what works and always be open to try new things".

Joanna is a strong advocate for narrative therapy approaches in her work. She said she believes narrative therapy offers an open approach that knows no cultural boundaries: "Narrative allows survivors to understand through their stories that if we can be telling ourselves a different story or seeing things in a different way, then that can really help change people's lives."

She says most men she works with have some form of trauma history, and that the processing of sexual trauma with male survivors often unfolds in a particular way:

What happens here is they start to put together the idea that they been sexually assaulted and usually even abused before that. That's what tends to make young men vulnerable – so many are in an abusive family situation and they have low self-esteem and are, you know, not connecting with kids at school. And so, they are a little out on the edges and someone picks up on that and takes advantage, and then they are sexually assaulted and all this adds up to having a lot of difficulty in relationships in a variety of ways. And then they either start to make sense of what's happened to them and...stop blaming themselves around what happened or [they are] just being left completely confused by the whole mess.

Common emotions that men will express around the event include guilt, shame, and confusion. She said men will internalize societal messages around stoicism and masculinity and will not speak up about their past trauma and "hide it away inside, but there comes a time when it flashes back before them and they get triggered back to the times they thought they had long buried". Joanna described adolescence as a confusing time for anyone, and notes how those challenges are compounded that much further by experiences of abuse: "any bodily response leads to a lot of confusion, and they have no one to – we are in a culture where you're not supposed to tell anyone, and so you have to process this all on your own."

Joanna talked about how expectations around masculinity can intersect with many other aspects of a man's identity and have a detrimental effect. Although she was clear to state she was not speaking specifically to experiences of sexual trauma, Joanna was able to connect a propensity for violent behaviour to past trauma history:

Many of the men that I know are able to stand up to some of those things they face, but poverty and disadvantage has left men - it must be difficult for many to think that you are

– for example – if you are this privileged entitled white man and yet you have no job and have no power and, you know, no agency in the world? And it's a precarious position to be in. So perhaps you get the idea that you will be powerful at home and overpower the people there. And so, you know, when you use your anger and rage against not having all the privilege that was supposed to be at your feet...well, you end up pretty pissed off.

Because many clients can struggle in that precarious position, Joanna works to draw connections to masculinity and patriarchal society in exploring some of the harmful messages men contend with in culture related to expectations around privilege, particularly when faced with harsh realities and limited resources.

Joanna says she is always interested in hearing people's stories and looking for hidden strengths or resiliencies:

I'm always listening carefully for people's skills and knowledge and ways that they've been managing to help develop a bigger story about that. I'm always listening for the effects of assault in their lives so that we can re- author those stories in a way that might be more helpful to them.

Joanna said that for survivors their resiliencies are not always clear from their current perspective, but when they are able to really reflect on what happened and change their perspective from an adult to a vulnerable child at that time, they can see for themselves how skilled and resourceful they may have been:

people often tell me that there was someone- if not at home, there was someone at school, a sports coach or a teacher or a neighbor or a community person who they respected and looked up to and who saw them as this interesting or good person that, you know, that they aren't able to see. So I work to help them understand how resourceful it was for them

to even have found that person, to attach to that person and to help them see themselves through that person's eyes — 'oh look, it was - they saw in you that, they, you know, reached out to help you or they let you know you were safe. That was a safe place for you to go.' To see themselves through those eyes is incredibly empowering.

Daniel

I met Daniel when he was days away from changing jobs and entering a new role, moving from team lead and clinical social worker in a community mental health clinic to that of manager in another division of mental health and addictions. Daniel identifies as a gay male, one of the few out therapists in his clinical area, and he is married to a fellow mental health therapist. He engages in a lot of advocacy work in the LGBTQI2S+ community, with a particular focus on connecting the queer community to mental health supports, a connection that he sees as significantly lacking. Daniel critiqued how so many of societal systems are highly heteronormative and cis gendered, and therefore not always perceived as welcoming by the queer community. For those reasons, he says it is important to him to ensure that "the community feels welcomed and accepted and are able to recognize aspects of services they are seeking that resonate with them- that they see in themselves -and part of that is working as an out, identifiable queer therapist".

Working primarily in addictions, Daniel sees a strong relationship between harmful substance use and traumatic histories. He is a strong proponent of trauma informed approaches, which to him means "creating safe, open spaces where people felt comfortable in processing and unpacking past traumatic experiences". He believes that as a society we are now having more open dialogue and discussion around trauma and its impacts, and he believes "in general our systems are moving in the right direction towards trauma informed treatment approaches."

Daniel described approaching therapeutic work in addictions as sorting through many different layers of complex problems and issues. While people generally present with some form of harmful substance use problem or gambling addiction, he said as the therapeutic relationship builds, they get to explore "all the things attached to that – the risky behaviours, the aggression, the complicated family or personal relationships" before eventually exploring how that all relates to trauma.

While he believes that many of the clients who come through addictions services have some form of trauma history, he feels most if not all LGBTQI2S+ clients he sees struggle with trauma in one form or another, often due to experiences with marginalization. He notices in particular a disturbing trend in working with trans males who struggle with significant substance misuse that often serves as a coping mechanism due to past trauma. He also believes LGBTQI2S+ men and boys are more vulnerable and susceptible to perpetrators and abuse, and that victimization can affect them in significant ways:

If we look at trauma as a continuum [from one to ten], and ten as the most severe or most impactful, I would say most of the gay, bi or trans males I've worked with would put themselves in that upper limit, just based on all those things they've been through.

Daniel has worked with youth in other roles, including the Youth Project, a local organization devoted to the needs of LGBTQI2S+ youth, and he works at times with teens in a small private practice he runs. His addictions work has primarily been with those 19 and older, and what is constantly surprising to him is the number of men, ranging from their twenties to their sixties, who disclose childhood sexual trauma for the first time during their addictions counseling. "My hope was that statistic was changing - that people were coming forward earlier - but that doesn't really seem to be the case".

Daniel says that something he feels is very unfortunate yet common with male survivors who concurrently struggle with substance issues is that they often seem very disconnected from their experiences of abuse:

They disclose this horrific experience and...it [mentioning the trauma history] was kind of just a...it was a passing comment. And what I mean by that is that they have been through so much in their lives, that was just another thing on the list and they weren't emotionally affected. It was down so far that they just glossed it over and it was just a normal part of conversation and I- they just weren't even in tune with this terrible thing that happened.

Daniel says while clients are open to hearing his recommendations, it is important again to always be respectful of their goals:

It is intertwined - hugely intertwined - it's a hard - it's a hard sell when someone is just trying to keep their shit together. So that's why sometimes it's just a matter of getting the foundation of addictions work down. Maybe learning coping behaviour but then perhaps there's another door [in terms of accessing therapy] at another time that's better for them to take care of the other piece, the trauma work. So, you're always dancing around it, and it's always in the room.

Daniel says that when most men he has worked with initially start to disclose, they have a limited emotional vocabulary to work with, so part of his work is to help expose them to more expressive language and explore more complex emotional states, "because they're happy, they're sad, they're angry – they can only identify these base emotions". While he works with women who struggle with accessing and identifying emotions in similar ways, he sees this much more commonly in men, and therefore the need for more slow and intentional guidance in developing

that vocabulary. He has also observed some unique differences in general between heterosexual and queer clients. He believes that gay male clients are generally "more in tune with their emotions and how they connect to experience, even if they can't always identify them right away", and he relates this to their rejection of masculine norms:

Straight men, by and large, they are still raised in this culture where men don't cry, men don't show emotion...gay men tend to access those feelings a bit more easily, but with straight men, I can – I can really see the struggle and the toll that it takes at times.

Susan

Susan was arguably the participant with the most direct experience working with young boys and adolescents' males who have experienced sexual trauma. She has worked for over thirty years as a trauma therapist and social worker and developed great expertise in working with young males who have experienced physical, sexual, and emotional trauma. We met in a small therapy room at the clinic she works, a room with soft lighting and shelves stuffed with books, puzzles, and toys and games that might appeal to all ages.

Susan said she was initially inspired to work with young men and boys through the experience of being a mother herself to two young sons. Being a mother is one of the greatest joys of her life, and as mother to two boys she says she felt a sense of responsibility to raise them a particular way on their journey to becoming men: "It was important that they learn to be compassionate and in touch with their feelings and willing to express those feelings, and teaching them that it is okay to cry". And so they were, she said, until one day they started school, and all of that began to change. Her boys were always bright and curious and expressive, but as she watched them interact more and more with their school peers they became more closed off, withdrawn, and at times sullen. This caused her to think more about the gendered

nature of the school environment and the messages that her boys received from the adults that cared for them and the other children that surrounded them about what was appropriate for boys and what was meant for girls: "And so...this experience with my boys...it really afforded me a window into how culture and socialization teaches males to withdraw or get angry, and so watching that - that was simply just fascinating to me."

Another strong influence was her work in justice and corrections, where she met many men with histories of early childhood trauma, something that was usually unrecognized and as a result more often than not left untreated. The appeal of her current role as a trauma therapist was to work to "interrupt these trajectories and bring healing and recovery to these young men before their life course seemed kind of set".

Susan described her philosophy and foundational approach as grounded in the work of Carl Rogers, focused on person in environment with the belief that a person can self-actualize to heal and recover and reach their full potential. She believes in a therapeutic relationship between client and therapist that is based on trust, curiosity and compassion, and one that deviates from the therapist as all-knowing to the client as expert in their own lives. She said a good trauma therapist would "focus their energy on what happened to you, and not pathologize you by seeking what is wrong."

Susan notes few boys and young males make connections between experiences of early abuse and how it may impact their lives at present. She sees the impacts of their current struggles with behaviour and emotional regulation as devastating at times as they often leave survivors "disconnected and isolated away from everyone". She says because of this, psychoeducation around trauma is so important and an integral early step to treatment as it demonstrates to a

survivor that sexual trauma happens, that it can impact their lives in many ways, that they are not alone in their experience, and that they have the opportunity to heal.

Susan says she is conscious from the start about the environment she works in and how these young men might first perceive the therapeutic setting "because oftentimes two people in a room where confidentiality is maintained doors are closed- that itself can replicate the abuse experience". One way she alleviates that stress is to offer as much choice as possible to the client. She talked about how even the smallest choice can be empowering to a young survivor, from choosing where to sit to deciding how bright they prefer the lighting in the room. This also helps in terms of building relationships and trust and just helping someone relax into the process.

Susan says she feels it is important to link male survivors with other survivors in terms of having that shared experience and coming through the other side of it. She also sees that help build connections that perhaps are lost, and teaches ways to relate to others. She believes it is very empowering helping survivors to find their voice as the voice is often silenced by stigma and shame. She says that she seeks to:

identify that it's not about just surviving but really getting to a place in that person's life where...you're thriving, not just surviving...that you feel whole and complete and, you know, while this is always a part of you, it doesn't need to define.

Susan says the most important thing with therapy is to go with the person's pace and to consider whether the trauma was a one-time experience or of a long duration:

If it was multiple events, then that impacts the healing work. So, some, you know, may come and do a piece of work and go away and they may be fine for a couple of years and then something will bubble up and may need to go back and do a little bit more.

When asked about the future of this work with male survivors, Susan says these are her best hopes for the therapeutic relationship is helping people get in touch with their gifts, their talents, their strengths, and ultimately their ability to heal. She sees this as crucial in combatting stigma and overcoming it:

It's about always kind of reminding people to be kind and courageous and kind of overcome the stigma. Because the stigma really mimics the abuse, because stigma wants to maintain the secrecy, silence the voice, heighten the fear. [Stigma tells the survivor] 'You can't recover. You can't - all this will be too hard'.

Kate

Kate was born and raised in the United States and completed her social work education there before moving to Canada. She has practiced for more than twenty years and has worked in various roles, including in child protection, as policy advocate and developer, and as a mental health clinician. She says she is passionate about anti-poverty work due to experiences working in child protection services and at an inpatient mental health unit and noticing a strong correlation between poverty and mental health struggles. Kate notes that belonging to a low socio-economic background often leaves one vulnerable to a host of other challenges in terms of their physical and mental health, and by addressing poverty on the macro level some of these issues become "fixable". Kate says she feels a similar way about the trauma work she does today: "it appeals to me for the same reason - if you heal trauma, it improves a whole other host of areas of people's lives- with their mental health and in their social relationships".

Kate says her central role currently is to provide immediate crisis support to families and children during times where a child has experienced physical or sexual assault. The clinic she works in serves as a bit of a nexus point for the various agencies and systems that are involved in

the investigative process following an allegation of abuse, including police, child protection, pediatrics, mental health, and the justice system:

Before we had a place like this, what they found was that it could be very traumatic for kids and families to go through that investigative process...to go from place to place, person to person, to not know what's happening, feeling afraid to ask questions.

Kate says the families she sees often struggle with so much guilt and shame "it's the worst nightmare. They failed and in terms of their primary job as a parent. That's how they see it - I failed my child". Part of her work is designed to address some of those things parents might struggle with, and involves providing psychoeducation workshops and education related to trauma. She said the inspiration to provide free workshops to families came from families asking the same questions over and over – such as "tell me the signs and symptoms to look for after the trauma". She says that "I think our natural instinct is to be like, here's the name of the therapist, bring them to therapy and it's not really the question that they asked" and so the workshops became an attempt to answer those basic questions.

Kate subscribes to a neurosequential model of trauma and believes that "trauma is a natural side effect of a healthy brain trying to take care of itself". At the same time, she also recognizes the role of social and cultural forces and how they impact on survivors, and how they might contribute to resiliency. She feels that trauma often goes unrecognized and untreated, and sees the medical model, problem-solving approach we often take as problematic:

Trauma doesn't fall neatly into that category. It's not something wrong that's happening in your brain, it's something right, and so I work to promote the idea that a trauma response is a natural consequence of a healthy brain. It's our body and our brain's way of protecting us from serious harm.

Kate acknowledges that most of her experience treating survivors have been with women in a role as clinical therapist, which has provided her good insight into how gender constructs can impact and impede women's healing. She says she tries to approach working with male trauma survivors with a similar gendered lens:

My approach to males is to ask the question – do you feel your gender has impact on the way you are experiencing this thing – and to ask in a way that they don't feel like there's judgment...It's an approach as a way to be real about some of the things that people have to deal with in when faced with these traumatic experiences.

Kate believes the vast majority of sexual trauma survivors, both male and female, do not seek help. In her experience as a crisis intervener directly following a traumatic experience, few male victims of sexual assault go on to seek therapeutic interventions. She says they often only seek help if they "absolutely cannot avoid it – like, if it's now an ultimatum to live in one's home, continue to attend school or work, or stay in relationships."

Kate says it is imperative we change the public narrative around childhood sexual trauma. She sees the ways boys are socialized comes with some inherent danger in the sense boys are encouraged or allowed generally to have more independence, which can translate to this notion of never having to rely on anyone else:

I think that messaging is so - that's dangerous. Because there's an underlying message there, that no matter what, you [as a male] can handle this thing that happened, this terrible thing....that you that you alone need to manage this.

Nathan

I met Nathan late on a Monday evening at his private practice in the downtown core.

Mondays are the exception to the rule in Nathan's practice, as it is the only day he offers

traditional office hours. He rents a small room out of a doctor's clinic, but generally he takes advantage of an open lounge area that is available after hours to meet with clients. Initial sessions take place there and center on goal setting, but from that point "how and where we want to work together is up to them". A typical week might involve hiking through Point Pleasant Park, meeting at a diner over bacon and eggs some early morning, or a game of racquetball at a gym. He described a family session he had coming up that Friday at a walking trail off a nearby beach, chosen by the family deliberately "so that we're in a different space and not feeling so literally and figuratively confined."

Nathan started our chat by reflecting on his past as a bit of a self-described adventurer and thrill seeker when he was younger. He left his hometown in Ontario at an early age and lived and worked in various countries all around the world before settling in Nova Scotia. He spent most of his career as a teacher in the public school system before moving to a hospital based mental health care environment, but he soon found the hospital work quite limiting and restricting. During that period, he also returned to school to work towards a Masters of Education in Counselling degree, where an internship in a violence intervention program for men proved quite inspirational. It allowed him to develop what he called a dream project – an outdoor therapy group for men who were mandated into mental health treatment by the justice system. "It gave me a platform to work with these guys who society sometimes had almost written off as unreachable....it was a ways and a means for them to access the treatment they needed". Returning to his old job was difficult, and following the freedom and creativity he experienced in his placement, it proved much too constraining, and eventually he sought a leave of absence for his own mental health and wellbeing:

There were a number of things that were going on at the hospital that I found very stressful [in terms of expectations and limitations on therapeutic work] and knowing at the same time - this is around the time that my Masters did finish -I really needed to find a way out, because I knew it wasn't the place for me.

Fortunately, he found an "escape" when he was awarded grant funding that enabled him to embark on a two year journey promoting adventure and outdoor therapy at low or no cost, focussed on men and boys who use violence or who have traumatic histories. After great success with this venture, he was able to establish and build a private practice that he now runs full time, with a continuing focus on working with men and boys.

Nathan says he is thankful for his master's program, his internship and his supervisor as they all allowed him to be "a little looser in terms of how I approach this...this framework with my clients and really encouraging this idea of being comfortable, not having a specific label in terms of the way I work". Nathan describes his approach as "part narrative, motivational interviewing, person centered, existential while calling on arts-based, experiential kind of modalities." He critiqued formulaic approaches to therapy because they veered from being client centred: "being formulaic brings along a sense of professional arrogance that [the therapist] automatically know what they [the client] need, and of course that's not the case".

Nathan says he is mindful of not moving too quickly into history taking, because that can be re-traumatizing. He says for that reason he is very intentional about discussing consent and what it means during the therapeutic process. Nathan wants clients to be comfortable sharing what they can and using self-advocacy to slow down or stop when necessary. He said part of his initial work is an assessment piece around how much or how well they might have processed their past trauma:

Some people, you know, have already come in having worked on stuff and – well, just because I find out that someone has been abused in the past, for example, doesn't mean I need to hone in on that like a cruise missile right? Because they might feel like they're not ready to deal with it or – or they've already dealt with it in some way, shape or form. And that's where the invitation for them to be open and honest with me comes in through in where they're at with that.

Nathan says that if one of the key premises of mental health work is to support a client's sense of self confidence or esteem, then there has to be more openness to allow people the space to be who they are and to enter therapy on their own terms:

I like that idea is that it's not just about the readiness and in the ability [of the client] to share but the willingness to go there and that we [as therapist] need to be okay with that and whatever they decide, regardless if we think we know what might be best for them. And I think – I think that empowers the individual to possess a greater sense of self efficacy and self-advocacy. Now those are skills I can get behind, and I feel like they almost trump the idea of the stigma card. Stigma is harmful, and the more we as a society talk about things the less stigmatized they become, it's true. So yes, stigma needs to be addressed as a societal construct in the realm of mental well-being, but not at the cost of pushing people past their edges.

Gillian

Gillian described herself as a married mom of two grown children who works to balance her work and home life by challenging herself to take up new hobbies, such as learning to surf for the first time at age fifty. We met on an early Tuesday morning at her office situated in an intensive mental health treatment program for adolescents. A proud graduate of the Maritime School of Social Work, she has spent most of her career in mental health, starting closer to home in Ottawa where she grew up before making the journey to Nova Scotia. In Ottawa, she began her career at a rape crisis centre, working primarily with women, and learned many therapeutic skills that she carries with her today:

It was so important in that work to be attuned to not replicating anything that starts to feel overwhelming or overpowering in that therapeutic dynamic. And so I learned from that to constantly do a lot of kind of checking in around – 'Is it okay if we talk about this, you know, can we talk a little further? I need you to correct me if I'm not getting something right.' And then also helping the client see that well for one, it was an issue of power and its sexual expression might have been the weapon but it wasn't- it's different- It was different than sex. It was about power.

Gillian described now working predominantly with adolescents who are struggling with issues related to mood and anxiety, typically teens 13 to 19 years old whose life is "a little more off track than where they would like it to be and figuring out together how things like mood and anxiety might contribute to that." She works from an anti-oppressive, strengths based social justice perspective and uses emotionally focused therapy and dialectical behaviour therapy often in her day to day work.

She said what really guides her work is operating from "a place of success instead of a deficit model or...illness model or disability model...while maintaining an awareness of sometimes where folks are at is through no fault of their own but due to larger societal forces at play." She says it is important to always keep an awareness of the "axis of oppression and where and how they may intersect in someone's life and bring that into some of their understanding of self and what their journey has been".

Gillian said many male adolescent trauma survivors have not sought help previously because adolescents at this age often go hand in hand with avoidance:

Because it's tough work right and there's not - they are still developing a sense of self and an understanding of who they are. So, most often when I hear about the past work [they might say] 'it was all right but I didn't really get much out of that.' And so the work comes sometimes in slow pieces that I hope layer...layer on, and then start to form a bit more solid piece that someone can sink into. Yeah, so there's lots and lots of ambivalence, and to that I try to give empathy.

Many young males Gillian works have witnessed familial violence and sexual assault within their own homes, which in turn has had a profound impact on them:

Often the struggle is that where they get stuck is feeling like they somehow weren't good enough to protect someone, or to step in. They weren't strong enough. Or they should have done something at, you know, whatever tender age or so it was...and I do think males experience and carry that differently due to how their socialized than females who have witnessed domestic violence, that might see mom being hurt or hit.

Youth typically spend up to four months in Gillian's program, which really only allows time to develop some skills around distress tolerance so they might manage trauma work in the future:

mostly foundational stuff to hopefully get someone to a place where they're grounded enough and have enough skills to even look at the implications of what they have been through. And again, reminding them, reminding myself...it's a slow, slow journey."

Gillian believes it is important to address masculinity constructs and stigma related to help seeking directly, and often explores it early on in therapy. She believes stigma that many men and boys struggle with serves to perpetuate the silence that surrounds male sexual trauma:

I think, just to name it [the stigma around help seeking] and acknowledge it and recognize it and helps break the silence. Helps someone label those mixed messages they receive as men and boys around help seeking and – and just to put it out there and calls that stigma out.

In terms of myths around sexual trauma, the physiological response to sexual trauma as connected with enjoyment is a common one that she feels needs to be debunked for both male and female. She uses an analogy with survivors to how one experiences a burning sensation from touching a hot stove: "if you put your hand on a hot stove…it's going to burn whether you want your hand there or not. So understanding that arousal response…that's a big one that we still need to address."

Gillian says there are many things we could do differently to support all survivors of trauma. She believes it is important to let survivors explore trauma at their own pace and when they feel most comfortable and safe to process their experiences. She says it is important to be flexible in the therapeutic approach and to always recognize when someone is wanting to move forward and respectful if they need to pull back:

We need to be compassionate, patient, we need to give them time and space. We also need to give people wiggle room so to speak. And once you've said something it doesn't mean it's always in the room and it stands forever. It is always adjustable.

Gillian describes spending a good deal of time focusing on anger as a response to trauma with the youth she works with, because "it is a justifiable and valid response [and] it's also very accepted within the fabric of masculinity". She says the problem with many survivors is that anger can easily overwhelm a survivor and cut them off from an ability to express and

experience a fuller range of emotions, which has significant impact on the survivor and those around him:

And so it's amputating for them, but it also cuts people in their circle of care that they love and are attached to off from being able to see a fuller sense of a person as well. So, it it's not just damaging -that kind of notion of toxic masculinity isn't just damaging to men. It's also robbing everyone of a way to be fully expressive.

Ben

I met Ben at his home on an early Saturday morning at his invitation. Ben is a midthirties man who shared he was recently out of a long term relationship with a woman and was
"now adjusting to a new normal". He runs a private practice centred around working with mostly
men and boys with trauma history that he established following graduating from Acadia with a
Master's of Education in Counselling. Prior to that, he was a high school teacher and guidance
counselor for over twelve years. He had reached a point where teaching was not as meaningful as
it once was and he felt overwhelmed by "bureaucratic nonsense that came with the job" so he
abruptly quit and pursued a Masters of Counseling Degree, so that "I might be in a position to
help more people in the ways I wanted to help them".

Ben said many of his clients struggle with post-traumatic stress disorder (PTSD), borderline personality disorder (BPD), or significant addiction issues. "They often come in with so many unhelpful labels attached, so some of that initial work is just unpacking that and externalizing it and letting it go". He credits his narrative therapy training in helping clients with meaning making around experience. Ben says that building trust and a sense of safety are his first priorities in the therapeutic relationship, and that starts with listening and enjoying getting to know their stories.

If you can develop a safe connection and trust – a trusting relationship with your client, they will open up, and if you cannot do that, then it doesn't matter if you throw a thousand different psychological modalities at someone and are trained in all of them - without the connection, they're not going to open up to you, and you're still stuck at square one.

Ben shared that he was a survivor of childhood sexual abuse, and that history was instrumental in him becoming the sort of therapist he is today. As a youth, he faced various struggles, and was labelled wrongly by various professionals as having attention deficit/hyperactivity disorder (ADHD), issues with anxiety, and possible bipolar disorder: "so I was forced to see all these professionals, but no one asked about early trauma stuff...it just never came up in the room."

Ben says it is important to be intuitive and really understand the limitations of your clients who have experienced trauma in terms of processing their experiences:

Not everyone's going to go at the same pace so knowing when to push down the gas pedal just hard enough and when to ease off and understand, you know, this is where this person's at sometimes and it's nothing that can be rushed. And over time, you can learn when someone's ready to go to the next step, and to challenge them, and other times when to kind of back off - and that's this kind of a big learning curve you go through as a as a therapist.

He says a commonality he sees in the male survivors he works with are relational difficulties, whether that be with romantic partners, family friends, employers or teachers. Attachment and intimacy issues are common: "nobody wants to get hurt again and you're dealing with a lot of triggers and of course if you want to be in an intimate relationship you really need to be vulnerable...and being vulnerable has become scary". He says survivors often do not recognize

their trauma triggers that bring on dysregulated states, so Ben does focused work to help them be more attuned. He promotes this through things like mindfulness, yoga, and physical exercise:

So, just focusing and taking the time to listen to your body, see where it's at, figure out if really something is a threat or is it imagined – so if my body's still experiencing [a triggering event], how do I proceed with my next move behaviourally based on that? Is there evidence that this is happening again, or is this my body responding to trauma because it's happened in the past? So that's something to be very aware of, and starts with developing that awareness of your mind and your body.

Ben has a strong interest in working with young males and men who experience sex addiction, something he said modern discourses generally are misinformed about or get wrong. He blames that on traditional masculine norms: "with sex addiction...it's an affliction. It can be as harmful as the worst drug addiction. But with young guys in particular, sex with girls is something you pursue, you want, it's constantly something you are supposed to chase".

Ben says that he believes male on male sexual violence is pervasive and yet as a society we do not talk. He feels this is related to living in predominantly a heteronormative society, and while acceptance of LGBTQI2S+ people has grown in recent decades, homophobia is still rampant and more stigmatized when it comes to sexual trauma, "It's another layer for males that makes it harder to disclose".

He says it is still relatively rare that a young client will disclose sexual trauma, and that it is much more common to work with men in their thirties and forties: "this happened to them, you know, at age 10 or 13, but due to the fact that they just didn't trust, that they could not open up about that subject to their parents – that really is quite common". He spends a good deal of time in the beginning of treatment around psycho-education related to sexual trauma in males,

including reviewing the one in six statistic: They often feel alone, and this makes them feel less alone. You can't take for granted what they know or don't know...what's truth and what's a myth". Ben says many of the male clients he has worked with have reached out in some form or another and had a less than empathetic response or the experience minimized or dismissed: "I think that stays with them. If I tell I get slapped in the face. Or I don't know what might happen so I better not open any wounds".

He said in combating stigma we have to be more open and transparent about the trauma that happens to young men, or otherwise survivors will continue to hide their pain and suffer in silence:

If they felt they could be comfortable enough to reach out and talk to family and friends about it, I think men and boys would be doing it. Why are they not? Because they are still stuck in a place where it's just not safe to talk about that. So really normalizing that, you know, that this happens in our world today and it has now happened to you.

Ben says it is really important for caregivers, parents and teaches to believe survivors when they first disclose, and to support them and not judge them:

We need to get them resources as soon as possible so that they can let that healing process begin – so they can get back to their lives and kind heal and move past that, as opposed to just learning to deal with it. Otherwise, all sorts of things can happen and combines to affect your life in a very adverse way...it's important to believe them, to support them, not dismiss or downplay, and from there – from there get them to the proper helpful and responsive resources as soon as possible.

Ben says it is important to instill hope for the future in survivors:

I think with sexual trauma, you don't ever really 'get over it', but you can learn to manage it, and you can understand it, and you can live a very fulfilling life. It's not a life-long sentence. You can heal...you can recover. I'm living proof of that.

Richard

Richard has practiced for over thirty years and describes himself as a clinical social work and a mental health forensic specialist. He has run a highly successful private practice for the past eight years that offers a therapeutic group specifically for men who are victims of sexual assault, the first of its kind in this province, and something which has spun into all different kinds of specialty work related to the lives of men and boys. He has had a long and varied career, and served in many different clinical, front line, managerial and consultative roles. He does a great deal of policy and advocacy work related to the needs of the black community and the LGBTQI2S+ communities, and as a gay black male, Richard is a strong proponent of promoting more culturally sensitive models of treatment and care.

Richard completed most of his studies in New England, and began his career there in child protection before eventually transitioning to working within the prison system, both of which gave him his first exposure to men and boys struggling with significant trauma histories. He says as a result he became a "deep student of trauma there because I found correctional mental health often focuses on the criminogenic issues that these guys had, while doing nothing to address the terrible underlying histories." He centred much of his future research and studies on addressing trauma symptomology among prisoners in protective custody and isolation, and he had success working with men who were "written off as impossible to reach…by virtue of addressing their trauma, not by addressing their criminogenic issues".

Richard says he is a person of faith and that his faith helps guide his work in the sense of what he calls the "interconnectedness of the human family". He says that is very meaningful to him because "we don't have the luxury of ignoring each other, and so if people are suffering, that's my business. It is all of our business".

He says what he hopes to achieve is a greater equality in acknowledging and accepting the stories of men and boys and their victimization, and he talks of need for society to arrive at a more unified discourse not bounded by gender, "so that we can see the interconnectedness in all stories of trauma and victimization, and we can promote healing in a way that – that is therefore much less segregated."

Richard believes there are many similarities, regardless of gender, in working with a sexual trauma survivor, with believing disclosures as the most important part. He says the role of therapist is to:

lower the bar of skepticism and recognizing the difference between clinical memory and forensic memory – and by that I mean that our interest is not in proving a story beyond a reasonable doubt, but rather in hearing stories, knowing what happened and in what ways it has hurt or affected someone.

Richard states that exploring the concept of confidentiality is a crucial early piece in the therapeutic relationship. He sees some tension with a survivors reporting historical cases of trauma and a duty to report by a therapist allegations of abuse and harm to the Department of Community Services. He believes some therapists currently "get it wrong" in their rush to disclose to authorities:

I clarify to a client that – well, if you were to tell me a story that currently a child is being abused than, right there, I have a duty to report. But those historical stories of childhood

abuse that you have suffered, that you've seen other people suffer? As long as those people are no longer children, I'm not required to report, which gives people a lot of confidence in telling historical stories and helps them to guide what they will share with me. I think it is interesting how sometimes we almost coach our clients not to tell us certain stories...because that has such an impact on our clinical relationship. The moment I have to slap my forehead and say, oh, I have to call somebody, then right there your healing work stops.

Richard feels as though some men "literally cannot attach emotion to their abuse. And that's one of the significant consequences of this kind of abuses that it separates one from one's feelings and ones capacity to feel". However, he has also worked with young men who experience high emotions and highly disinhibited, emotionally chaotic states. He said many dismiss the experience, suppress or try to forget it, and struggle through life with some dysfunctional aspect because they've adopted as protection. At times he sees survivors triggered and then suddenly they are:

flooded with memories of having been abused... So these guys have to kind of come to an awareness that they have been abused and then they have to deal with the chaos, the emotional chaos that that unleashes in their lives and that their brain has been protecting them from for so long.

Richard sees survivors as blaming the abuser, blaming oneself, but something of interest he notes is "a real hesitancy to lay any blame on the caregivers who failed to protect". He says he has seen the most defensive and volatile responses in men he has treated when he asks about the role of their parents. He says he has seen this commonly in families where incest has been tolerated, where parents "discover it has happened, slap the offending child in the head and say 'never do

that' and yet the children go right back to the same sleeping arrangements. Nothing changes". He says this is often an indicator of a much deeper family dysfunction, which "of course needs to be explored and unpacked before... healing can begin".

Richard said in considering the ubiquity of abuse in both young males and females, it is important to remember that: "when something happens to that frequency, it's not healthy or natural, and we have to address sexual abuse as a health issue, rather than a criminal issue." By that, he does not mean sexual abuse should be decriminalized, and that while he sees the criminal justice system as an important intervention in protecting children, "the hard reality is that it does not do much to address the problem of child sexual abuse."

Richard proposes a system of "health surveillance" around sexual health and wellness as part of child wellness programs: "that starting as young children, during routine medical screenings, we are assessed around the state of [that child's] sexual health." He also believed clinicians need to be better prepared and have what he describes as the "internal fortitude" to do the work. He believes this is a huge disservice in our systems today that there are therapists who are not comfortable working with trauma, and so part of not addressing that trauma through therapy is potentially therapists' avoidance of tacking the issue head-on:

So we need more clinicians who aren't freaked out by this [childhood sexual trauma] and who able to see it for what it is - another mental health dilemma, for both survivors and perpetrators. And - and I think that's part of the mythology, the mythology around this – that essentially the idea that that a sexual assault should primarily be responded to by the criminal justice system and I reject that. And I'm not saying that that doesn't mean that there's there isn't a spot for adjudicating offenders. Absolutely. But I think that that if

we're not doing the healing work, then accessing criminal justice services as our primary approach is a primary approach that is ultimately failing.

Adam

I met Adam at his private practice office in the north end of the city. Adam is in his early thirties and lives with his long term girlfriend. He has a great passion for the outdoors and spends most weekends exploring hiking trails throughout the province. Adam works part time in a private clinic offering general counselling, and two days a week runs a private practice where he focuses primarily on working with men and boys.

Adam initially had plans on becoming a police officer, something he says had dreamed of much of his life. But as he progressed in the program, he started to realize "how dominant approaches to managing crime were not very effective, and that lead to a re-evaluation of my whole career path." He wanted to explore what a rehabilitative or preventative approach might more fully entail, so he took time away from studies to gain some work experience in places like men's shelters and group homes, where he was able to provide support around life skills and offer some informal counselling. That inspired him to consider graduate programs in psychology before finally deciding on the Master of Education in Counseling through Acadia University.

Adam completed a thesis as part of his program that focused on outdoor therapy as a treatment option for men who commit intimate partner violence. Adam's main participant grew up in a rural community, had a significant history of physical abuse at the hands of father, and as an adult became abusive to a series of partners. He had been through many different anger management programs and saw numerous therapists but nothing seemed to connect with him until he became involved in an outdoor therapy program that, as Adam describes, "allowed him

to connect with nature, something he did to feel safe as a boy, and then begin to engage in therapy on a much more authentic level."

An internship at the East Coast Forensic Hospital led to him developing an interest in working with males with significant mental health diagnoses such as schizophrenia and bipolar disorder. Because these men often struggle with aggression in their daily lives, Adam says there were often "written off" as too dangerous. "But once I started to have a conversation with a guy, I very quickly became aware that he didn't want to be doing what he was doing [acting out aggressively] - he just didn't know how *not* to do it".

Adam does not believe there is a "one size fits all approach" to therapy. He says he focuses on social determinants of health, with an understanding that the issue is not located within the individual, and that there are multiple systems – family, community, broader society - that influence someone's thoughts, feelings and actions, and that therapy needs to be responsive to those influences. He works with survivors to "expand their emotional vocabulary" along with their capacity to experience those emotions: "The biggest emotions tend to be anger, rage along with sadness, and feelings of loss, although initially they might not name it as such and usually focus solely on their anger." He has worked with survivors who disclosed at a young age, and often it can seem as though their world falls apart in the aftermath, which they then often internalize as blame and shame around the sexual trauma, and interpret it as the "consequence for doing that thing they weren't supposed to do".

Confidentiality is also key to the therapeutic relationship, something he sees as a critical tool for therapist to use "but something that is sometimes forgotten". Confidentially allows the space to have these therapeutic conversations and for some survivors the chance to disclose secrets long kept. Adam says he reiterates confidentiality often throughout the therapeutic

process. Another major piece of the work for Adam centres around consent, especially when exploring past trauma: "It's almost - it's asking a question to ask a question, if you know what I mean...I've got some hints of what happened, and now I am asking permission to ask the question [around sexual trauma]".

Adam feels the stigma is particularly strong around mental health and help seeking:

There is just that overarching stigma around, you know, not having your shit together or not being able to snap out of it, which is just like such a - it's just such an ignorant position to be in - or ignorant notion to hold - that if you're dealing with depression, you can just snap out of it right? Or if you're dealing with anxiety, you know, if you're dealing with trauma…like 'yeah, but that happened a long time ago, so like, just move on.' Get over - get over it. Like the absolute least helpful words in the English language you could ever say to someone.

He sees young survivors of childhood sexual abuse suffer from unresolved trauma which ultimately results in "a worse off or poorer quality of life". He believes the most devastating impact of childhood trauma for males is the emotional and social disconnection that he often sees experienced:

Relationships and their ability to be in a relationship, it all suffers...and as social creatures we need - we're humans and we need social connection. Which is another major thing a lot of men are lacking. And I think is - again it goes along with the sort of this masculine script [that] influences our behaviour - our tendency to isolate, and we need - we need social connection to be healthy. That hesitancy to seek help will just fuel that sense of isolation.

Adam talked about young men who struggle with aggression and aggressive acts that in turn can sometimes be traumatizing, and leads to further impacts on them and those closest to them:

The cycle of violence is real. So many of these guys experienced or witnessed trauma as a kid and they are now acting out as adults – they're harming themselves and their harming others. So without getting help in between it just keeps on happening. And now the kids are experiencing it, and if they don't get help, they might grow up to do the same thing. And while, honestly, I hate - I hate how deterministic that sounds, but it real. The cycle is very real.

Shane

I met Shane over a lunch hour at my office on a busy weekday. Shane is a 34-year-old male employed as a community outreach worker/addictions counselor with NSHA, and holds a Bachelor of Social Work Degree from the University of Manitoba. Shane identifies as queer and is married to a same sex partner who also works in the field of mental health and addictions. Shane's career path is similar to mine in that he worked for many years as a child and youth care worker with vulnerable youth before returning to academic pursuits as a mature student in his BSW studies with plans to eventually pursue an MSW. He has also done significant policy work in the past around child and youth care practice and trauma informed care. In his current job, he works primarily with adults diagnosed with a concurrent disorder, meaning they are presenting with a mental health problem combined with a substance use issue.

Shane was excited to tell me about the recent launch of a "Seeking Safety" group, which is a group therapy offered through NSHA aimed at survivors of physical, emotional, or sexual trauma. He described the group as one centred on developing coping skills while exploring the linkage between substance use and trauma and how previous trauma might impact someone in

present day. This particular group, he says, is unique because it was the first one offered through his program site that was solely focused on men with sexual trauma history, something he had advocated for since starting in his role. While he knew from his work with vulnerable youth that trauma histories in men and boys were fairly common, he was surprised at the sheer number of men who were struggling with sexual trauma history:

So, I've seen men coming in acting super hyper-masculine and presenting as very agitated or confrontational until the sexual assault dialogue has kind of surfaced within the therapy and then they become a little bit more softened and more authentic. They really have a really hard time articulating around what that means as a male being sexually assaulted either as a child or as an adult. So, it's about exploring this guilt and shame and how that relates to self-compassion and healing and the trauma. So, if we're looking at safety as one of the first stages of trauma, it is about helping them understand what that means to them and how to get that safety

Shane focuses in individual work around dialectical behavioural therapy (DBT), because he likes how it teaches "good coping skills and tools around emotional regulation" but he says a starting place is always exploring with a client what approach might be most relatable to them. He also relies on motivational interviewing as a way to "explore and kind of elicit change talk."

Shane spoke at length about boys and men who use anger and aggression as a means to manage and cope with the stresses that are in their lives that are often related to or a direct result of their significant trauma history. He sees it again as means to protect themselves, to keep them from being vulnerable, but what become problematic is that it served to disconnect them from the resources they need to improve their situations.

He reflected on past work with young men and boys who presented with over sexualized behaviours and says that related discourse in mental health is problematic in that:

We have this big reaction around preventative stuff. Protecting the other youth or children around them, labelling them as sexually aggressive when really what we need to do is help the child or youth understand what occurred to them, and help them learn new schemas of how to appropriately work with people or even be around people. And as a last step, helping them develop a way – a means where they can articulate what happened to them.

Shane emphasized again how establishing safety was key to starting work with male survivors.

After a baseline of safety occurs, Shane focuses on remembering or recalling the traumatic experience, using prolonged exposure to gauge distress tolerance, as it allows a means of managing the discomfort of taking about trauma and not engaging in maladaptive or problematic behaviours as a means to cope.

Shane spends a lot of time with male clients exploring what gender means to them as a means of exploring normative expectations of masculinity. "It's important to explore as well how these two things can exist in the same space - because masculinity and femininity is part of all of us, so recognizing and accepting that femininity is a part of being male." He believes as a culture we need to teach even really young children more explicitly about constructs of gender and help develop a better understanding of gender identity and sexual orientation at an earlier age. "I feel if young people don't have a good understanding of those things, they develop their own understanding and different ways to cope, and sometimes their attachments and their relationships get damaged or disrupted because of that". Shane sees the way young children are socialized in terms of gender roles as still problematic, and has worked with parents to address

some of those tensions in the past. He says he is often concerned when working with aggressive boy that they internalize messages about being "in charge" at home:

I always cringe when I'm with, say, a mom of a 4 year old who's acting out significantly and she'll say 'well, he's the little man of the house', but then we don't necessarily say to our girls 'well, you're the woman in this household'. No, he's a little boy. He's just a little boy.

Although he says he generally presents as cis gender male in his outward appearance and dress, Shane more readily identifies as gender fluid, "somewhere on a continuum between masculine and feminine, and I can display myself however I feel in the moment". He says he notices a conscious shift when around people he has met for the first time or does not know well, as he tends to act more of what he called stereotypically masculine: "how I carry myself, how I walk and how I sound, how I dress – I'm conscious of all that". He says as a society we continue to "place people's gender based on their genitalia and not necessarily on how they identify"

Shane believes exploring vulnerability and the way aggression is sometimes used as a means of protection is crucial to how we can help male survivors. He says that many struggle with guilt and shame as it relates to their anger, and that often times it is misplaced:

I'm recognizing I'm angry right now, but I'm not angry at you, but I am going to display that anger at you because I'm uncomfortable because I'm having these trauma symptoms or triggers or flashback. And then relationships get damaged because I am acting out.

And then guilt comes from that and then shame from allowing themselves to feel vulnerable, which can feel like that same vulnerability that led to their assault.

Conclusion

These narrative accounts allowed me as researcher to explore and analyze each narrative "temporally, spatially and in terms of the personal and the social" (Clandinin & Connelly, 2000, pg. 89) while attempting to understand and retain the participants intended meanings within the context of their experiences. The rich, descriptive data also allowed for interpretive themes to emerge which go on to form the basis of Chapter Five. Through including them, I attended to Clandinin's (2013) belief in the participant's voice as most influential in narrative inquiry and my hope is that this serves as a means of honouring those voices included here.

CHAPTER FIVE: INTERPRETIVE FINDINGS

The following chapter includes findings of participants' narratives that were interpreted from the rich textual data provided during the semi structured interviews. Following Braun and Clarke's (2006) analytic process, data was summarized according to patterns which allowed for interpreted themes to emerge, with focus placed on significance of the patterns and themes that emerged and their broader meanings and implications. Braun and Clarke also suggest the use of vivid or compelling examples from participants' narratives that would support themes. The rich, detailed content and resulting themes were considered from a postmodern lens in order to "hear the operation of broader social discourses shaping that person's story of their experience" (Clandinin & Rosiek, 2007, p. 55). Overall there were seven themes that I chose to focus on: Invisible Victims/Hidden Survivors (examines the silence that surrounds the phenomenon that renders its victims invisible); Some Moral Distress (describes challenges and constraints participants as therapists face within a medical model of care); Dangerous Disclosures (details some of the challenges faced by survivors in disclosing their trauma); Challenging and Renegotiating Masculinity (considers how participants as therapists work with male survivors to deconstruct and reconstruct different versions of masculinity and address issues in toxic masculine culture); Misogyny/Fear of the Feminine (examines how participants challenge dominant discourses around women as weak and narratives of femininity as bad within the context of patriarchal and misogynistic society); *Homophobia: Am I a Fag?* (considers myths that perpetuate homophobia following male on male abuse within the context of heteronormative and homophobic culture); and Re-Storying Trauma (details how participants as therapists work with clients to understand resiliencies and vulnerabilities and being the healing work).

Invisible Victims / Hidden Survivors

Cultural myths surrounding the sexual trauma experiences of boys and men pose serious obstacles to understanding the impact on male survivors and serves to derail healing processes.

All participants described a sense of profound silence around the phenomena of male childhood sexual abuse within society, which in turn has the effect of hiding or making invisible survivors. Participants believed this occurred in several ways: through a lack of public discourse around the prevalence of male childhood sexual trauma; through a lack of available male centred supports and resources, and through a professional bias in clinicians that fails to recognize the traumatic histories in male clients.

When describing their experiences, participants noted that few young males actually present to their services with sexual abuse histories as a primary presentation and reason for seeking treatment. Yet while participants widely agreed few men and boys enter mental health treatment with childhood sexual abuse as a primary concern, they all spoke of their belief in the ubiquity of male childhood sexual abuse due to their experiences with men who disclose later in life, and often quoted the one in six statistic (Fisher et al, 2007). Daniel noted a significant number of men he works with in addictions services experienced some form of childhood trauma, estimating "at least 75-80% of the men have some physical abuse history, and many of those include sexual trauma history". Susan and Richard described sexual trauma histories among incarcerated men as an even higher statistic, with Richard noting the last prison he worked with in the US "over 80% of men incarcerated had experienced at least one unwanted sexual encounter in childhood". Shane reported that the statistics around males impacted by sexual abuse are similar to those suffering from prostate cancer but noted "the silence that

surrounds that number - I mean, think about it. Only one of those things has fund-raising and an awareness campaign behind it."

Adam believes that male childhood sexual trauma is very prevalent, but something rarely spoken of: "It's a scary thing to think about children being sexually abused. So, I think we prefer *not* to think about it". Kate and Richard, with their backgrounds first in child welfare services, were adamant that disclosures and investigations of male childhood sexual trauma were higher than what common cultural discourse would suggest, and more specifically, what mental health care services would report.

In her role in helping families navigate complex systems following a child's traumatic experiences Kate says,

People don't want to see it until they have no choice...until they are forced to deal with it [the disclosure of sexual trauma]. We need to see it more because people are getting away with it, and that perpetuates the abuse. So, to make ourselves see it, to open ourselves up to these hard things to see, we need as a society to talk about it.

Not recognizing or acknowledging the significant issue of male childhood sexual abuse is problematic in that it serves to minimize the experiences of survivors and therefore divert them from the services they might otherwise need. It makes them invisible to the very mental health care systems that should be readily able to support them. Joanna spoke of the need for mental health professions to bring discourse around male childhood sexual trauma more readily into the public consciousness to raise awareness and reduce stigma. She believed through opening the discourse in society, survivors will be better positioned to access the help they need. By continuing to not recognize the significance of male sexual trauma, we inadvertently silence its hidden survivors.

It is the silence I think that is difficult -that if men could feel that they would not be judged, if they were to speak about these things openly and readily. And of course, you can do that in counselling, but then there's – the shame about that, there's stigma around that as well [in acknowledging the need and seeking mental health support] you know, so there's a lot of work to be done. A long way to go in recognizing the problem. We are sadly pretty underground so far in this area [of male sexual trauma work].

Peter described the phenomena of male childhood sexual abuse as a "mostly silent area" in trauma work and mental health, and while reflecting on his diverse career in mental health, expressed surprise around the lack of visibility of male trauma work.

It is definitely not something that surfaces a lot, you know, day to day here in the clinic or - or in the media. There does not seem to be any kind of advocacy rights groups banging the drum. I just know that there really isn't a lot of focus - *no* focus, really.

All participants described themselves as advocates for trauma work and innovative ways to help trauma survivors, but when it came to male childhood sexual trauma that advocacy work comes from a small minority and is conducted, as Joanna noted, almost entirely underground.

All participants were challenged to describe immediately available resources for men and boys who experience sexualized trauma. As clinicians, they claimed they were often highly versed out of necessity in finding available community resources for a wide range of issues impacting their clients, and yet with the phenomenon of male sexual abuse there was no, as Peter described, "obvious fit or ready referral source for male survivors available". Not knowing where to access specific male-centred services in dealing with sexual abuse issues was highly problematic to clients. As Adam described it:

All of this has much bigger implication on healing processes, because if a survivor does not have an entryway into help seeking or does not feel safe or comfortable when forced to engage, then it comes impossible to do the healing work.

Gillian echoed a very similar sentiment, noting male adolescents who disclose and are open to ongoing therapeutic work have very few options available to them:

So, once you get to that place of trust willingness, who's there? And it's not like they have their own financial means, especially if they're not - if they haven't disclosed to their family and they're trying to seek services on their own. They can't just go and find a great private therapist who works in trauma. They are left with few to no options.

Participants also noted a professional bias in public health care that reinforces the belief that men and boys cannot or are not sexually abused. Shane described a need to challenge some colleagues and their formulation and approach to treatment with male sexual trauma survivors. He recalled working in an intensive treatment program with a fifteen-year-old male who had been displaying some sexually provocative behaviour towards his male peers. Shane noted this youth had disclosed to his primary therapist his history as a sexual trauma survivor and his experience as a victim of sex trafficking. However, this disclosure was met with a great deal of trepidation and in some cases disbelief by the clinical team:

This was a group of smart professionals. Caring professionals. And I found myself having to constantly point out the fact that – you know what? Young males can be victimized, they can be trafficked into the sex trade. It happens. And yet it's like – it's almost like they're confused by that notion that young males can be victimized and - and preyed upon. They dismissed his acting out as 'he's confused about his sexuality, he's gay, he's not comfortable with it, so we need to focus on more relationship skills'. Like –

dating skills? For a kid who's been trafficked...can you imagine? So, I think there's a very different focus and...maybe a different empathy for boys when they've had these experiences.

Participants described working with young survivors who had been in therapy in the past and had not disclosed, something they attributed to the multitude of challenges to disclosure that I will discuss in a further section. They also expressed concern over screening processes for experiences of childhood sexual trauma in men and boys, and even more concerning, clinicians' ability and comfort level to identify, explore, and treat the issue. Richard said it is crucial that therapists be prepared to hear these stories and act upon them: "we need clinicians that are competent and comfortable in hearing these difficult stories and can put their biases aside". Richard also believed this had to start before the point of seeking mental health services, with the idea that doctors should be willing and able to screen all men and boys for past history of sexual trauma, and from there have access to resources to refer their patients on for further support.

Along with acknowledging a general lack of expertise, participants were also quick to note there are few if any training opportunities specific to men and boys and sexual trauma available, therefore making it difficult to build a competent practice to address the issue. Participants noted the training and workshops they attended were all specific to female experiences with sexual trauma, and while many of the same principles applied to treatment (such as establishing safety and helping to build tolerance) other aspects, such as challenges to male identity and questions around sexual orientation, were gender specific to the needs of men and boys. Ben stated, "it's not like that [specific work with men and boys] was part of any modules at school". Shane said even when male sexual trauma is addressed in workshops he has

attended on sexual abuse, "the focus is on women and girls.... IF men and boy are mentioned it – it gets tacked on...almost like an afterthought".

Nathan states that while there is a great deal of research that look at poor help seeking behaviours of men and boys and seeks ways to address that deficit, he believes "for the most part I think the focus of research is all wrong". He believes it is much more important to focus on the goodness of fit or the general lack of fit between types of therapy offered and how that relates to the way men and boys engage with the world around them:

Traditional sit, talk based office based time bound, all that kind of stuff. And how men and boys tend to show up – that needs to be the focus. And if we're going to have one iota of a chance, not only to shift the conversation on masculinity, but in giving men and boys an opportunity to voice some of that distress that's going on - sexualized trauma or otherwise - it will happen on the edges of how we offer that therapeutic work, work that is more than just sitting on our asses and talking. So whether that is sitting over coffee, going to a music store going on a walk, walking through a cemetery, going rock climbing...whatever form it takes. Otherwise we'll continue to struggle in the stigma and in the relative voiceless of these men and boys because it's really not about...it's not about how gender affects men and poor help seeking behaviour. But yet we keep blaming it on gender constructs, when really it is about what the fuck we're doing or what we're not doing to support the boys.

It seems abundantly clear that the silence around the male childhood sexual trauma stifles the conversation and leads to a lack of public discourse. The lack of discourse leads to a lack of disclosure and a lack of knowledge related to supports and resources centred on the needs of men and boys. And then ultimately, the lack of disclosure and lack of supports leads to sexually

unanimously advocated for the need to bring experiences of male survivors out of the shadows and into the light. They challenged the cultural beliefs that men and boys cannot be victims and pushed the need for more public discourse and more focused attention in health care settings. They also described the significant lack of resources and supports available to survivors, and how that not only impacted their ability as an individual clinician to provide effective treatment, but also their faith in the larger systems they work in to address the needs of men and boys with sexual trauma histories, something that will be explored further in the next section.

Some Moral Distress

Participants in this study were open about the impact of perceived barriers and limitations in public mental health system in addressing the needs of male sexual trauma survivors and the moral constraints those systemic issues places upon them as therapist in their therapeutic work. I argue here that this results in experiences of moral distress, which is defined within a health care realm as a helping professional making decisions and judgements about the right course of action to take in any given situation in the best interest of their client, but being unable to carry out that action (McCarthy & Deady, 2008). In other words, the therapist knows what the right thing is to do to help their client, but they are unable to do it for a variety of reasons, or they do what they believe is wrong because they lack other discourses.

Something that came up in all conversations and weighed heavily on participants was the lengthy wait times for service in seeking mental health supports. Participants who worked primarily in adult community mental health (Daniel, Shane, and Peter at present, and Ben and Nathan by history) believed wait times were generally between six months and a year after point of referral to see a therapist for an initial appointment. Wait times for children and adolescents

were generally better, according to Shane and Gillian, with those times anywhere between six weeks and three months for a non-urgent referral. Priority referrals typically were based on risk of harm to self or others or acute psychiatric presentation (e.g., symptoms of psychosis such as hallucinations or paranoia combined with a significant decline in functioning; daily panic attacks; significant compulsions or obsessions). These initial appointments were typically booked within seven days, but the wait time for an ongoing therapist for follow up was also lengthy. Gillian also noted the wait time for a specialized more intensive program like hers was generally at least a year or more. By nature of the program she works with, Kate tended to see boys and young men immediately following a disclosure around trauma, so while an exception in terms of how quickly she provided care and support, she was also limited in her ability to refer people on to other services due to lengthy waits. In her role as a trauma therapist, Susan was constantly triaging children and adolescents referred to her service, so depending on urgency they could be seen quickly, but as a triage system that meant that others that presented less urgent (for example, someone that witnessed a terrible car crash versus someone who suffered a sexual assault) would be pushed down the list to wait further. Either way, this presented to Susan a terrible choice.

Ben was highly critical of the public mental health system in general, and believed the system currently:

broken and beyond repair in its current state. I mean, the ridiculous wait times, the lack of services available... stories of people where they go down to emergency room when they are dealing with something very acute or long term and can't get support....when you have suicidal thoughts or behaviour and you cannot get a bed, and you just get turned away.

Other participants described similar challenges and stressors within mental health care systems. Peter believed strongly that treatment for trauma always involves a slow and steady approach, designed to build confidence with the process in the survivor, and with time to develop a comfortable rapport with the therapist. He said, however, "with a brief sort of intervention approach, we might fool ourselves into thinking we are doing something when in fact we haven't done much by virtue of the timeframes allotted for treatment". He then contrasted that belief with the constraints he saw inherent to the public mental health system:

It's kind of pointless just reassuring people or being flippant. And I don't mean flippant as in...well, you know, without ever meaning, it can come across as very shallow if you just adopt certain approaches or make certain assumptions. So, I think you really do need to get to know the person. Which often can take many visits, which is challenging in a system that is not necessarily open to that. We are very action oriented – like, assess, check, treatment plan, check, initiate, check, discharge, check.

Susan, Ben, Nathan, and Bryce expressed similar thoughts on the way mental health systems are structured and how they do not allow for that degree of intense therapy and focus needed for survivors to progress through successful treatment processes. As a result, they believed a hurried approach that often barely touches the surface of the trauma experienced and be in fact traumatic to client in itself.

Bryce critiqued brief focused therapy approaches, and says that most clients he sees reported "traumatizing experiences" with other therapists in public mental health systems where they feel uncomfortable disclosing past history and engaging in treatment:

Just the fact that my approach is ... at the very least, built on validation and is trauma oriented and anti-oppression oriented. I hear a lot about why that approach is so different

than walking into an office and sitting across from a psychiatrist and being diagnosed in twenty minutes. Nobody feels heard in that situation. Nobody feels like...you know, the problem here is 'I still don't feel safe to say these things, and I don't feel safe to say these things precisely because you haven't given me either the time or the space to do so.'

Gillian critiqued current systems and responses that she felt did not provide adequate service to trauma survivors in their healing processes:

I don't think our system really sets itself up or lends itself to help people get there [recovery from trauma], because that's slow gentle work and we keep trying to piecemeal our response to people in distress instead of recognizing that 'you know what? This is a long slow journey. It's a long slow journey here.

All participants in private practice (Nathan, Adam, Bryce, Ben and Richard) described offering treatment pro bono, at reduced rates or on a sliding scale based on what the individual accessing treatment could pay. But while the intention of offering reduced or no fees is admirable, younger less established clinicians like Adam and Ben are often faced with harsh economic realities of trying to sustain a viable practice and cover rent and administrative costs. Nathan was able to offer no cost services while accessing grant monies in the past, but since starting his private practice "I've tried to offer a sliding scale and lower costs of treatment, but the sad reality is to be self-employed and stay afloat, I'll never be able to offer the free or low cost service I used to".

Kate questioned how often symptoms of sexual trauma were not identified or missed in the histories of young men and boys, and finds it distressing to think of how many agencies and services (school, clubs, organized sports) likely interacted with them and did not identify problematic behaviour as a possible indicator of trauma:

What's so striking to me is.... well, I see male victims at the time they are victims. There's a lot, right? But you don't see a lot later represented in therapeutic interventions, in other services in mental health. Like, they're here, and I hear their histories, and yet they've never sought help before, but – but it's never been offered. And I think you see a disproportionate representation in criminal justice systems [of males with sexual trauma histories]. And even then, if you asked, they might not disclose. But they're there. So why aren't we more attuned to that?

Gillian admits that in her present work in an intensive treatment unit for adolescents she has worked with few that have gotten to that point of disclosing their past abuse history to her:

I think my – my much more common experience has been working with kids who were just really kind of behaviourally breaking the rules and non-compliant and you know maybe...you know, finding out years down the road and this piece [around childhood sexual trauma] comes to light. And in those cases, I've been – well, at times I've closed the door and cried, and others I just- I want to bang my head against the wall and say 'damn, what's wrong with me? I missed that. How the hell did I miss that?

Kate, Richard and Joanna all criticized a health care system with research readily available that illustrates the struggles males have with help seeking, and yet with no consistent approach to addressing the issue. As Kate says:

If we as a collective society know this, if we know this history and we know these struggles with help seeking and we don't react - then it's like we've basically just given up on this entire site segment of population.

Many participants also spoke of unique stressors in their work with male survivors in providing, as Daniel described it, "therapy as a consequence." For most men and boys they have worked

with, engaging in therapy was almost always externally driven, and something survivors are forced to take part in. For Adam in his private practice, this was particularly troubling:

It's almost always other people either recommending or suggesting or forcing them to go. Especially with the younger guys where it's like, typically a parent you know, or someone at school - someone is saying 'okay you've done this thing and you need to now go for counseling. So, I can often be seen as sort of the *stick*, which is a difficult position to start with, when really [as a therapist] that's not my goal – I don't want therapy to be seen as the punishment.

Gillian also struggled with the idea of these men and boys forced into therapy. She saw help seeking in the adolescent population she serves mostly externally driven, in that other people have pushed them into help seeking:

It's often a caring well-intentioned family member or teacher - or via justice or police or school suspensions, and often coming out of concern about behaviour, aggression and anger. But the youth are often scared, suffering.... they don't see it that way. And so, it's challenging as a therapist who wants to help- I don't want to be seen as this – as some form of punishment or consequence for them due to their lashing out.

Ben spoke of often seeing young men and boys not ready to commit to therapy but forced into it, usually by a loved one who insists. He said therapy is often presented to his client as "do this or else" and he also struggles with the idea of therapy as consequence:

It puts us both in an uncomfortable place as therapist and client. I find it's not helpful because generally if you're not willing to go on your own...without that insight or if haven't....well, sadly, if you haven't had enough consequences in your life to get to that breaking point, it generally doesn't last long term for these people. It just becomes about

readiness, about the readiness to do the work. And it's not fair that – that me or someone else gets to say when that readiness comes.

Echoing what he mentioned earlier regarding a goodness of fit to therapy, Nathan stated he realized over the course of his career as a therapist that it was just as important that the work he did resonate with him as much as it did with the client:

So as much as it is about the client and what works for them, it also needs to be about us and where we feel most resonant, because if we're not in our most organic states, if we're not doing our most organic work, then we're not being our most organic authentic selves.

And if we're not authentic, how do we expect to be a support and help to our clients? Despite concerns with current service delivery and experiences of what I have labelled "moral distress" in their efforts to provide effective therapeutic work, participants believed that treatment for sexual trauma needs to be a public resource available to everyone and increasing the number of clinicians and developing more specific gendered training related to male sexual trauma would go far in accommodating some of those needs. They were frequently challenged with finding appropriate supports and resources in the community and believed this population of male survivors was under-represented and under served in our systems of care. Finally, they struggled with the idea of therapy as a consequence and the therapist as "the stick" and looked at ways of addressing the challenging introductory client/therapist relationship.

Dangerous Disclosures

Participants talked about challenges centered on acts of disclosure that for many survivors can feel tenuous, uncertain, and dangerous. Some participants, like Joanna, Peter, and Daniel acknowledged having limited experience working with young boys and youth who disclose histories or experiences of sexual abuse and described a more frequent occurrence of

working with men in their twenties, thirties, forties, and even fifties who disclose for the first time. Literature around male sexual trauma (Alaggia, 2005; Allagia, 2010; Gartner, 2000; Lisak, 1999) demonstrate that later in life disclosures were extremely common for male survivors. Although Ben had experience working with both younger and older males who disclose, he more commonly worked with men who had experiences with disclosure as young boys or teens that were met with some level of resistance or disbelief:

I have worked with men and boys who have reached out at an early age - and I think that those initial negative responses, whether they were patronized, or it was blown off or not taken seriously, whatever - I think that *really* stays with them. They've learned – 'you know what? If I do that and tell I am left deeply ashamed while everyone else is uncomfortable...or no one believes me, or I get my face slapped. So now I'm just going to stay away from that –I'm just not willing to go there.'

Ben spoke of how crucial it was to restory that early experience of opening up and telling their story in his young male clients in a way that illustrated "he was right to disclose…the problem was those he disclosed *to* weren't ready for the telling".

Several participants challenged what they described as the dominant discourse in how society responds to allegation of childhood sexual abuse. Some, like Gillian and Susan, described the experience as the equivalent to a bomb going off in a family's life and then struggling to make sense of the destruction and fallout that follows. Others, like Kate, noted how parents of young children could feel betrayed or judged for not protecting their child, and can internalize anger and upset, which then spills over to the entire family". Kate described parents being overcome with "an immobilizing sense of guilt and shame", something that also contributes to

the silence around it: "It is their absolute worst nightmare. The ultimate betrayal. They failed in terms of their primary job as a parent. That's how they see it - I have now failed my child".

Richard believed the way society reacts to claims of child abuse is problematic, and can serve to put children at further risk:

When kids do make disclosures, literally the universe falls on top of them. Think about it. What happens when a kid says, "someone touched me"? The daycare or the school goes wild, and they call Child Welfare. Child Welfare calls the police, and together they investigate to see what happened and who is responsible. And so, the kid's world completely falls apart – and remember, the kid [from his perspective] is just telling a story. And the next thing they know there are uniformed police officers and child social workers and their parents are crying or angry or distressed and the kid learns – he learns very, very quickly, not to talk about that stuff.

By hitting a "panic button", he argues, we run the risk of silencing children because they may then internalize responsibility for the negative reactions of caregivers and others around him. However, by not responding, there is a risk of creating a discourse that lessens the problem and diminishes the significant and harmful impacts on young survivors.

Bryce believed that "part of how deeply stigmatized male sexual abuse is actually in a weird way contributes to how I [as the male victim] experience the thing itself". He described a commonly held belief that child abusers are punished and possibly murdered in prison by fellow inmates for their crimes. Bryce believes this underscores what is "potentially one of the worst things in society, and therefore that means [as a victim of childhood sexual trauma] I'm experiencing the worst thing there is that could happen. The worst thing ever. And now I cannot say it to anybody, I can't tell anyone...because it is that bad."

Several participants critiqued common discourses around mental health and trauma that impact men and boys' disclosures. Shane says the current medical model tends to pathologize men and boys when they are seeking help and support for trauma by nature of the way they present:

I think when we see men who are seeking help around trauma, one of the ways that they experience their emotions.... their array of emotions is as labeling them as anger or aggression. It comes out that way, and our clients get labeled as hard to work with or difficult or aggressive when really they're trying to manage and cope with the stresses in their lives that result around significant trauma history. So, they get labeled as someone who's -someone who's going to be harmful or even dangerous to work with, which makes it difficult for them to seek help because people aren't willing to work with them.

Other participants described similar sentiments in their work with survivors. Joanna noted how many of the men she sees struggle with emotional dysregulation which makes disclosure that much more challenging and puts focus on their problematic anger or acting out and attention away from the triggering traumatic event(s):

they struggle to regulate themselves because of it [their past trauma] - they're getting triggered all over the place and don't understand what's happening to them, and that becomes so limiting...so staying in school, holding down a job staying in a relationship, being able to manage the difficulties of parenting, you know, all of those things become hard...just basically getting along in the world is so difficult.

Gillian and Kate described working with young male adolescent survivors that were fearful of disclosing to family because of their potential reactions. Gillian said "these youth are often so vulnerable, and they have no other means of financial support. So, if you are assaulted by

another male live with parents who either do not believe this phenomenon even happens or cannot accept it or have homophobic beliefs or religious beliefs, it silences you.... you simply cannot speak of it".

Kate believed that perpetrators of young males use that fear of disclosing:

I think that there's a certain amount of- I think our natural tendency [when victimized] is to go to a shame and guilt response, especially if you are a kid, because you don't understand the full context of the situation. If you look at perpetrators, they use that against kids to -they use it in that they – somehow, 'this is your fault' or you know, that they manipulate their victim...that they scare them into silence or into submitting.

Ben acknowledges struggling with disclosing his own sexual trauma history, and noted he was well into his twenties before he first sought out treatment, at that time encouraged by his then-girlfriend. He says his struggles around disclosure were related to feeling unsafe in a closed off environment:

I grew up playing hockey. I'm still very immersed in the typical Canadian hockey culture which to this day.... you know, the entire sports world...it- it really hasn't moved much in terms of being a safe environment to share - due to the fear of judgment. And I'm not suggesting that there aren't people in these dressing rooms that have not been through it [i.e., experienced sexual trauma], but I can tell you that no one's talking about it. And if I were to speculate it is because they would not feel comfortable about the responses they would get - and those response will be quite negative humiliating and certainly not compassionate or empathetic in any way.

Bryce saw self-blame intertwined with both fear of reprisal of abuse from the perpetrator and fear of disruptions in families as a reason for many to not disclose:

To say it out loud and threaten the family to say it out loud meant to invite further abuse to say it out loud meant to make your mom feel guilty, whatever, right, you just you just never had any capacity to have this thing exists outside of you in a valid way so it just turns back in and turns back in. I've just learned to be silent. I've learned to be invulnerable in these ways and now I don't ever ask for help or speak up or use my feelings because those are dangerous.

Participants reflected once more on the need for more public narratives around experiences of male childhood sexual abuse and recovery as a means of promoting disclosure. They queried whether having a major public face to the struggle – such as an actor, musician, or an athlete – that might champion the cause would make someone more likely to disclose or seek help through inspiration from that public figure's story. Regardless, participants believed that more public conversations about the reality of childhood sexual abuse and the ways in which young men and boys may be impacted were crucial to lifting some of the stigma and raising awareness of the issue. At the same time, they recognized that the landscape of disclosure was fraught with dangers and perils in the form of retaliation and rejection which inhibit the survivor from reaching out and contribute to their silent voices. Referencing a theme related to masculine identity that will be discussed in further detail in the following section, Kate summed up the experiences of survivors and their challenges with disclosure in this way:

There's something about the identity piece that is harder - because if I [as a male survivor] feel like 'I don't know how to be a man or I am less of a person than I need to be', then it really does impact every single aspect of your whole life. It's so much harder to seek help. It's so much harder.

Challenging and Renegotiating Masculinity

Participants described an initial stage of therapy in developing specific awareness of masculine norms and the ways in which they play out in an individual survivor's daily life. Understanding how those norms influence men and boys behaviour on an individual and contextual/societal is a crucial starting block for therapeutic interventions with men and boys (Allagia, 2010; Gartner, 2017b; Lisak, 1994). From there, it was often a matter of challenging and unpacking those norms and helping a survivor separate thoughts and ideals that were positive their influence, and those potentially harmful.

Some participants, like Adam and Nathan, said it was important to first help outline what a healthy version of masculinity might look like. And part of that work involved exploring gender constructs and gender expectations for both males and females and how they may present as problematic. As Adam notes:

...in that sense, I like the idea of masculinities as plural. And so, for me...my starting place is... So, my idea of a healthy masculinity or healthier masculinity is one that allows for I'll say, for starters, the expression the full expression of our range of emotions or human emotions, because we regardless of gender we all have the capacity to experience and express all of the same emotions. What I think gender concepts do in general - like masculinity and femininity - is channel all of those emotions into very specific ones and that's not the healthiest way of being. Not for anyone.

Nathan says he believes it is first helpful to illustrate what masculinity does not have to mean:

We need to sort of shake the traditional constructs of masculinity around, you know, needing to be adventure seekers and highly sexually exploitive, or doing no wrong and not needing help being self-assured etc. etc. That's the first part...but in doing so, the

invitation then is to explore a whole range of possibilities along the spectrum of masculinity or even the spectrum of - even better said is along the spectrum of masculinity and femininity and I think one thing that this work has taught me is that is to be fluid with my understanding of even what masculinity is and I think the greater the stronger invitation is to a) ask and be curious and b) find ways to be okay with however we're showing up

Accepting the way a client presents in therapy sometimes meant recognizing some of the limitation placed on male survivors who ascribe to traditional masculine norms, especially those related to stoicism and restricted emotions. Joanna described it this way:

I just see how unhelpful it is in so many ways to -to be told that 'you're supposed to know everything' and 'you're not supposed to ask for help' and 'you're not supposed to have any emotions and...'. No, it's just- just bizarre. What it does is...it just shuts men down. It shuts them down and they live on this very surface level.

Adam says it is broader society where "masculine scripts", what he described as norms of expected masculine behaviour, plays a significant role. He believed that often those impacts were insidious and that men and boys were rarely consciously aware of the impacts: "we're likely for the most part to go on about our daily lives, unaware of the influences that, you know these, really deep rooted beliefs come from and how they're influencing us as men and boys."

Gillian noted similar concerns with male survivors she works with, and sees impacts not only on the survivors, but on those that surround them:

I think it's just really tragically limiting. In terms of the range and depth of themselves...

I'm thinking, particularly around emotional expression and how so much of it gets

channeled predominantly down a flow of, you know, anger or and cut off from an ability

to express you know a fuller range of emotion that includes you know sadness or hurt or joy. And so it's amputating for them, but it also cuts people in their circle of care - that they love and are attached to - off from being able to see a fuller sense of a person as well. So it -it's not just damaging -that kind of notion of masculinity- that toxic way of subscribing to masculinity – it isn't just damaging to men. It's also robbing everyone of a way to be fully expressive.

Peter and Richard, both men and therapists in their late fifties, believed that traditional masculine norms related to power, strength and dominance were still quite pervasive and impacted survivors' ability to seek out and access support. Richard stated that acceptable enactments of masculinity were based on those norms: "the lines are still fairly well drawn. They are well drawn particularly for people of African descent, Hispanic descent...but I think for most men, those boundaries are still quite powerfully and enforced."

Participants saw those messages of what it means to be a man and what the limitations or boundaries might be as influenced by many sources – among those family, friends, school, workplaces, community, social media, and popular culture. Peter sees those influences embedded everywhere in modern society, with expectations to comply with them all around:

I think about it sometimes -like, what does a man do? What is he supposed to do, as a man? What does a woman do? When you are at work at home... you know, when you're out...what do you wear? How do you stand? How do you talk? And what's interesting is that it is all so ingrained, I think. I mean, I think the culture- the culture around us has a huge influence in how we present ourselves. And I guess most times we don't even realize it.

Daniel says he works initially to understand how survivors define their sense of masculinity identity without first making assumptions. He subscribes to the notion of multiple masculinities, with "people coming from all walks of life, and our constructs are based on where we come from, the families we grew up with, the communities, the media and its messages that surrounds us". He feels the way masculinity is traditionally played in the media and popular culture is problematic "these whole extremes of masculinity and femininity and we see ourselves in comparison to that. And it's impossible for anyone...these standards for anyone to meet".

Although he acknowledges presenting as male in his outward appearance and dress, Shane feels he lies "somewhere on a continuum between masculine and feminine, and I can display myself however I feel in the moment". He says he notices a conscious shift when he is around people he has met for the first time or does not know well, as he tends to act more stereotypically masculine, and is aware of "how I carry myself, how I walk and how I sound, how I dress. It's like...sometimes depending on the crowd I'm around, I'll hear my voice deepen automatically, or I cut loose and hear myself laugh just a bit louder". He says as a society we continue to "place people's gender based on their genitalia and not necessarily on how they identify – and that's a huge problem".

Participants described trauma survivors sometimes use aggressive or assertive behaviour as a coping mechanism. Ben noted how some survivors might "wield toxic masculinity like a weapon", but how it was important to look at their behaviours in context and consider what might constitute survival skills. Susan echoed similar sentiments:

It actually makes perfect sense why a person might display toxic masculinity beliefs when they've been sexually assaulted because it's a way to protect themselves from not having the same thing happen. And it serves as a way to almost re-establish my manhood,

because the messages I've received are that these things – these things don't happen to men. So, I think that normalizing what that actually means about you – that this show of toxic masculinity is really a way to cope and protect yourself, and then being able to say if we're going move forward successfully here, we need to look at new ways to cope.

Participants noted how having a good understanding of societal constructs of masculinity and expectations around dominant masculinity were key to treatment. They viewed a significant part of their role was to help challenge some of the more damaging or harmful standards while working to separate and preserve the positive aspects, and help survivors form a new and healthier understanding of their masculine identity.

Misogyny/Fear of the Feminine

Most participants relayed stories of male survivors receiving messages throughout their lives that sexual assault was something that only happened to women and girls and therefore outside the experience of men and boys. Men could be perpetrators, they might say, but not victims. As Joanna noted, when sexual trauma,

becomes understood as a woman's issue....it shuts men down and they live on this very surface level, and anything that's remotely to do with women or femininity is just it's a horrible curse – it's every curse word. To have this happen is to have done something feminine.

Ben notes how many survivors in his practice have struggled with their trauma as what they described as "this woman thing that to me", and so his focus is on psychoeducation and reinforcing that it was rather a human experience:

Whether it's, you know, a man that's been victimized by sexual violence or it's a woman. It shouldn't matter. I think because of some societal norms – that's what holds back a lot

of men from reaching out for support is because this culture that says, 'you know well that's a woman thing that happened to you'. I mean...what the hell does that even mean? You are now more like a woman? This idea that no man could let that happen to him...it gets in the way of healing. So being mindful of that and certainly realizing that this is someone - it's a human being, who's been victimized, and gender doesn't really matter. But I think for a lot of men, it really *is* harder. It's more difficult for them to disclose and open and so normalizing that experience in - in helping them understand [it could happen to anyone]. Because I find that many clients tend to be dismissive or they minimize or just shut it out....and so, you're helping them understand that this is - these are severe violations that have happened to them as a person – and that doesn't make you any less of a man for having it happen to you.

Many participants reported having to address issues with misogyny, hatred or mistrust towards women as part of the therapeutic process. Part of that work involved critically exploring the influence of patriarchal society that calls for the positioning and policing of women as inferior or subordinate to men (Manne, 2015). It also means considering how this positioning and policing extends to men and others who exhibit feminine like characteristics that challenge masculine ideals. As Ben, Kate, Shane, Joanna, Susan and others noted, important to the work was addressing that fact that sexual assault was not something that is solely a female experience - it can happen to anyone, regardless of gender, sexual orientation or any other defining characteristic.

Kate and Susan believe how we socialize males towards females and how ideas of gender develop over childhood has a huge impact on how survivors process their experiences. Kate

describes a fear or hatred of anything feminine as central to the problem, something she believes starts when boys are very young:

If you look at insults - the worst insults you can make relate to like female genitalia. Men learn, starting when they are very little, it's somehow less good to have feminine characteristics and that somehow, it's more feminine to be the victim of violence and sexual violence, that's part of...I mean, it sounds terrible but it's really true, that somehow if you're a victim of a sex crime that - that makes you more female in the perception of it.....and that somehow to be female is somehow not – that it's somehow bad.

Shane states that many male survivors he works with struggle with feeling weak or helpless, and that "by admitting this [trauma happened], they're showing some femininity in some way. And that's problematic...relating femininity to weakness or even vulnerability, but it really is ingrained through how it's understood...through the way we construct it." Susan agrees that experiences of weakness and vulnerability for many survivors go against the male discourse and lends itself to their struggle:

So, the traditional belief about masculinity, I think, often focuses on dominance, independence, and an orientation to the world that's kind of active, assertive, valuing competitiveness, turning away from intimacy and achieving esteem kind of through force. And I think at the heart of that, you know, the fear is of emotion and if men see or express emotion, then somehow that compares them to females who are believed to be weaker or more dependent.

Bryce describes many survivors he works with equating feelings of weakness as emasculation as they struggled with accepting their past history of trauma:

So, they will.... essentially they will emasculate themselves. 'I should have done more. I should have said something to somebody. I can't believe I'm such a pussy'. I've even heard of stories like 'I was made to feel like a fucking girl'. Like all kinds of ways of ... emasculating themselves. Just all kinds of condemnation. Self-condemnation.

Bryce notes that female survivors struggle with similar self-condemnation and questioning things like why they did not fight back more, but for male survivors this is more intricately connected to their male identity and their claim to masculinity.

Kate believes there are cultural expectations and pressures on men not to see themselves as victims, and to admit to being victimized poses a significant threat to one's sense of self and identity.

It does something to the ego and how we identify ourselves. You think of yourself as less of a man, where with women it's a different thing. So, as a woman I might be a victim of a similar thing, but I can see that's not my fault, and that it doesn't necessarily shift my whole view of myself [as a woman or as feminine versus a male's experience].

Joanna reports that most men she works with struggle with cultural expectations around masculine behaviour, and she says for many male survivors of trauma, what that translates into is that:

Literally anything that's remotely to do with women or femininity is just a horrible curse. It [femininity] is literally every curse word. To have this happen is to have done something feminine and so it's, you know, horrible...it's a horrible thing, and so no one wants to get in touch with that. And these male messages in the world -this entitlement and this disrespect for women. I mean, how can you grow up in the world with disrespect for half the population? Like good luck with that!

Managing misogynistic thoughts, beliefs, and attitudes was very common to participants work with male sexual trauma survivors. Participants described how their clients often equated victimhood with femininity and as a result seemed to at times denigrate or demean anything associated with women and girls and tended to internalize anger and frustration towards all things considered feminine. In order to establish their masculinity, men and boys would cast out anything remotely to do with femininity. This finding was interesting because while there is a clear linkage in the literature between misogyny and anti-femininity and men's violence against women, there does not appear to be an established connection between misogyny and anti-femininity and the experience of male sexual trauma, other than the rejection by survivors of emotions related to vulnerability, which is arguably described as a more typical feminine characteristic. This will be unpacked and explored further in the following chapter.

Homophobia: Am I a Fag?

Participants described survivors' struggles to understand why their perpetrator selected them, and how they were often left overwhelmed with feelings of shame, guilt, and self-blame relating to their victimization. Participants all believe this was most shameful and difficult to navigate when the perpetrator was male, and many equated that to a heteronormative culture that continues to perpetuate homophobic beliefs and stereotypes that are harmful to male survivors. Survivors also struggled more intensely if they experienced a bodily response and arousal to the abuse, which they often equated to signifying they either were gay and therefore attracted the perpetrator unwanted attention, or they would now become classified as gay due to the experience. Participants noted there was little pattern to why boys were selected, except that those more marginalized were likely most at risk. Examples of marginalization were given as

young survivors who were isolated from peers, or who grew up in homes with intimate partner violence, or with caregivers struggling with substance abuse or mental health issues.

Survivors who questioned, "what was it about me that made this person select me?" were often left with a silent, inward struggle to understand the experience, which can lead to an undermined sense of self. Peter described working with survivors who struggled where that questioning "has definitely interrupted and interfered with just the kind of stable sense of self, with sometimes a feeling like they are one person to society, and then another person to themselves."

Homophobia serves as an organizing principle of a cultural definition of manhood, and one that some men struggle to position themselves away from (Kimmel, 2008). Susan says many of the survivors she works with struggle with sexual identity, and she describes "their worry that the experience made them gay or means they are destined to be gay." She described working in corrections and dealing with homophobia among survivors that impacted their ability to access:

Many of these men would adamantly refuse to work with male therapists – because it was a male who perpetrated on them - and so they needed the 'safety or another gender',

but unfortunately sometimes a female therapist just wasn't an option in those settings.

Kate believes experiences of sexual trauma can leave young men and boys struggling with questions around their sexual orientation and the influence and impact of the sexual act itself:

Many don't even realize they were victims. Like it didn't even occur to them, even though it had impacted huge areas of their intimate relationships for years following, even though there was no consent. Because their body right responded in a way they somehow take as their own fault, so therefore they couldn't be a victim. And if a person isn't gay or bisexual or anything and it's a male perpetrator, very clearly in their mind

that am I now gay. Does this make me gay? Because of their body's responding in a way that you know that they don't want it to. It's so much more confusion in an already confusing time. Because they can get an erection, they actually they feel like that somehow, they wanted it. So it's almost like it couldn't have happened. It couldn't happen unless you got an erection, so therefore you participated. You are complicit in it.

Richard also sees how male on male abuse sets up some confusion around sexual orientation for many survivors. He says part of the early work in therapy is exploring the concept of "body betrayal", the experience of sexual arousal during the abusive encounter:

'I was abused by a man and my body responded positively to that and so I must have liked it'. I think especially when you achieve orgasm while being sexually abused. It creates real problem for men, and something that you often have to work - the fact that your, your body has its own its own neurologically and physiologically way of responding to things that's completely separate from your beliefs and cognitions.

Bryce worries for some that challenging their questioning around sexual orientation leads to further confusion and pain for some survivors, and does not necessarily need to be directly addressed:

My thoughts are that if trauma has something to do with why somebody ends up preferring men to women, so what? I don't need homosexuality to be a centralized as I was born that born that way because like I think that's something that we think - something that queer communities have had to use to try to secure themselves in some way like "we can't help it." But what if you could? You should be able to... you should be able to choose this too, right? So, if trauma has something to do with why [a survivor]

ended up feeling safer with men or women or identifying gay or straight, I don't have any problem with that. If that's somewhere in the storyline...I don't think that makes you worse or wrong or bad or anything. There doesn't have to be the right reason to be gay, trans or anything...

On the other hand, Daniel believes that young males who experience sexual trauma in early adolescence and during a time when they are starting to form their sexual identity as particularly harmful and that it needs to be addressed as part of the therapy process:

I've worked with men who have experienced abuse twenty years before in their early teens, and up until now they've never spoken about it. And they express a lot of sadness and confusion, anger of not being able to make those choices or not coming to that decision in terms of their sexual identity in their mind by themselves because of the confusion added so early in their lives by the abuse. And it leaves them with shame and so many what if questions. So, it wasn't the... process of somebody coming into their own, of identifying how they wanted. And so that's a lot for them to take in, and something that needs to be unpacked and explored.

Richard says that, in general, queer men more often than straight men are more likely to intentionally seek out his group for male survivors:

It's like they are coming out of the closet as victims of sexual assault, which is like their second or third coming out whereas with straight men – it's a big deal. And I think that a lot of straight men have difficulty with, particularly if they were abused by men, have a hard time squaring that off with their heterosexuality. You know, men will say things like, did he abuse me because he saw something in me? Which is - it's almost a fearfulness of the possibility of queerness within.

Richard says he firmly believes that for heterosexual men, reporting male on male abuse is problematic because "we live in a society where homophobia is still a real thing. And so, to say that this thing happened is not only shameful, but it also has the added dimension of calling into question your sexual orientation". Susan also echoed this as a complicating factor for many survivors:

Because then there's the whole issue with sexual identity and how that impacts intimate relationships going forward. And again, creating a safe place to talk about that [experiences of sexual trauma] because that's just not easy for human beings to talk about, period, and let alone, you know, male survivors to be able to kind of talk about that. But I feel what is crucial is to help a survivor not to view every other male person through the lens of the abuse and to really illustrate that there are healthy respecting men that- that can be kind and loving and that men can be with in a relationship.

A heterosexist society is harmful and damaging to male sexual trauma victims of male perpetrators because it stigmatizes what for the survivor is already a shameful act. Fear of a homophobic response makes young male survivors reluctant to disclose, because they are afraid of the resulting stigma and stereotypes. Sexually abused boys often feel deep shame about their abuse but are even more ashamed to disclose due to the insidious effects of a heterosexist and homophobic culture.

Re-Storying Trauma

Meaning making in trauma work requires survivors knowing their larger stories, something that is often challenging, both because their stories are so difficult and because of dissociation and fragmentation among the parts of that hold different aspects of the story (Grossman et al., 2017). Participants felt that a significant part of their work was helping

survivors build a coherent narrative and that a more integrated self was key to the recovery process. They described approaches that I will present under seven distinct subheadings which will help illustrate and explore some of the treatment processes they describe adapting and using with male survivors.

Psychoeducation

Participants talked about the importance of psychoeducation around the prevalence of male early childhood sexual experiences in the early part of treatment and, as Ben described it, the need to "normalize what isn't normal and reassure people they aren't alone in this journey". Gillian stated that "to just sit with them and help them understand, sadly, the harm that we can bring to each other, to know this was an abusive thing that happened to me and it was wrong...there is power in that". Susan saw psychoeducation "as providing a foundation that all of treatment is based upon" and said that:

to give knowledge and understanding that this is a shared human experience, that others have walked this path and that not only have they survived, but today they are now thriving – that in itself is incredibly uplifting and powerful.

Dealing with Uncertainties

Ben acknowledged a need to correct misbeliefs or untruths around the phenomena of male childhood sexual abuse, particularly in situations where families either did not believe a disclosure or dismissed or minimized the impact of the experience. Through that form of response, a male survivor often questioned the event with their own degree of uncertainty: "if you as a parent are telling me it didn't happen or it's not a big deal, my only choice is to walk away thinking the same".

Kate and Daniel both described working with survivors who did not realize the abusive acts in the accounts they were describing. Kate says "they can recount this terrible story where they have been manipulated, used, threatened, and the story – the story trails off to a question. 'Was that trauma – was that what I think maybe it was?'"

Richard described uncertainty and denial as playing a significant role in his work with survivors of incest perpetrated by older brothers, and said:

invariably there will be stories of the abuse having been caught or discovered, where a parent will slap the older child in the head, say 'don't do that!', and still they go right back to sharing a bedroom, the same sleeping arrangements, and the same ongoing cycle of abuse with no one calling it that.

He noted how the incestuous experiences became almost normalized to some survivors, "something that probably happens in lots of families, and so therefore he [the survivor] says 'maybe then it doesn't affect me that much'".

Working through Shame

Julia spoke of how prevalent shame was in a survivors' beliefs and attributions about himself, and something that had to be tread carefully around. She talked about approaching the work with:

relentless empathy and validation and curiosity. I think that's pivotal to kind of...to help someone even be able to come out and begin to peek out under from that veil of shame, to begin to show themselves and to show up in a way that's authentic and true and maybe different.

Shane says the shame around these experiences can be overwhelming, much of which he attributes to heteronormative assumptions in society that make it challenging to process sexual traumatic experiences for men and boys:

There's so much shame, so much stigma...tied to the sexual abuse, and all the stuff that makes you question around your sexuality, your orientation. 'Did I invite this?' But there are also the other things that are shameful like I mentioned before, like the substance use or the fighting and the constantly trying to prove yourself as – as a man. There's shame that comes with all of that, too.

Susan spoke to the silencing aspects of shame and how it "almost colludes with the abuse", while Bryce approached the issue as "shame is not letting me [the survivor] be vulnerable, and by not being vulnerable, I'm not able to heal".

Attaching New Meaning to the Aftermath

Several participants talked about helping people re-story their lived experience because they were not only dealing with shame and guilt due to the abuse experience itself but sometimes also the aftermath. As Shane described earlier, an important tenet of his work and something he describes as feeling very passionate about is:

helping people understand the reasons why they — why they...connected with some certain people and not with others, maybe made choices around relationships or substance misuse they weren't proud of.... I feel like it is crucial to redefine some of those things done in the past as was a way and a means to survive, as a means of coping with stuff that was pretty hard to take, and, yeah, that they had to do these things as a means of survival.

Joanna noted how it was important to reconcile the survivor narrative as part of your overall identity, but to always keep in mind:

there's so much more that goes into what you, as a human being, represent with your many skills and your talents and your wisdoms. If anything, this experience shows that you are strong, and that you can overcome most anything, and that is such an important message to take away from it all.

Susan stated she believed that many people who primarily trauma work do so as a result of their own histories with trauma as a means of contributing and helping others. She believed it was important to work with survivors as they heal "to give back to others, to think of ways they can use their gifts and be the instrument in someone else's healing".

Honouring Survival

Ben described a crucial part of that work involved "empowering that...that ability of being able to survive until now; regardless of how they survived, despite some truly terrible circumstances". Ben labelled this "highlighting hidden resiliencies". Resilience is defined as an engagement in behaviours that helps the individual [the trauma survivor] navigate their way to the resources they need to sustain and flourish (Ungar, 2011). Processes of resilience only occur, however, when the individual's social ecology (i.e., their formal and informal social networks) has the capacity to provide resources in ways that are culturally meaningful (Ungar, 2011). Most participants believed that helping survivors locate these hidden resiliencies was helpful in allowing the survivor to uncover more successful ways of engaging with the world.

Others, however, thought it much more relevant to point to actual capabilities a survivor has in that moment, considering their age, their environment or their circumstance. Richard said

it was "important to put yourself [as a survivor] back in those shoes of that little boy, not the man you are today". As Joanna described it, that work revolves around helping a survivor:

realize what a vulnerable little boy he was, because often they put that all that maturity and experience on to that incident but when they sit here, I can remind them over and over: 'How old were you? Do you know any 8-year-old boys that could...?' I feel like it's important to look to change that perspective, to take away the maturity and the experience they infuse in it, and highlight instead the vulnerability

Re-imagining Identity

Gillian spoke of how being a survivor of sexual assault can, for some, become allconsuming to one's identity. She described working to foster a sense of a more multidimensional identity within survivors:

So, yes, a sexual assault survivor is part of who that person is, but maybe they also do a kick ass job of creating art or maybe they are an incredible baker.... to not see them solely as a survivor, because they really are so much more. Helping them see that, and to rework or rewrite that story. And yeah, sometimes the survivor piece shows up in the other areas and that is where some work comes in to understand, like 'oh you know, it was this part that maybe peeked out and popped up there and maybe this was the trigger'. I do a lot of parts work that helps people manage conflicting emotions as well - there could be a part of them that feels something and another part that is pensive or scared, and that is OK. And it gives people wiggle room as well. Once you've said something it doesn't mean it's always in the room and it stands forever. It is always adjustable.

Other participants talked about helping survivors manage conflicting feelings about their histories. Joanna noted that this didn't always involve their history of abuse but rather:

other stories, like how they are relating now. Often people will tell me things about a time when they lost control and.... hurt or lashed out at their parent or their friend or their partner - something happened, something triggers them back...they're unsafe and they have that trauma response to fight. So, you can ask them, you know, what was going on in that moment? So, we go back and help them realize that what was happening in that moment before they acted or reacted? So that they're just really slowing down what's happening to them, so they can get some understanding of what's going on, how they're getting triggered to past emotions and how they can help ground themselves in a way that assures them that they're safe – and that those around them are safe – right now in this moment.

Richard shared his belief that, through healing, a survivor needs to include their abuse narrative as part of their overall narrative, because "at that point you no longer need to operate from a place of shame". This becomes complicated for survivors because it forces them to evaluate all relationships in their lives, and these relationships may have implications that are shameful or harmful to survivors. If they operate from an assumption they were a dysfunctional person prior to healing work, he says, and now see themselves as a more functional person in the present, they might question whether the relationships they have today are the relationships they need to be healthy and whole. For Richard, it is not only about re-storying the past but also looking to reshape the future:

So, your world actually changes as you heal and you actually have to prepare to live at that next level because there is- well, with growth comes loss...and so, you gain this new level of health and functioning, and you actually have to integrate yourself at this new level. But the thing about growth is that you're going to grow for the rest of your life. So,

once you've attained some kind of stability at that level, guess what happens? You begin to experience more distress, which is confusing because you're already, in your mind, 'healed enough'. Suddenly, new needs and new opportunities come and then you say, 'Oh, my goodness, I'm seeing things that I've never even seen before.' And now I realize that because of my abuse, I'm not ready for this stuff, so I must go through his healing cycle again for the next level. So, this helps to see that healing from abuse – that it's a cycle, a never-ending cycle, and in that sense, it becomes – it becomes almost a never-ending chance to re-story one's life

Participants all believed that telling and re-storying the trauma narrative were important parts of the healing processes for men and boys. But due to the narrow discourse on experiences of sexual trauma and men and boys, they were often forced to pull together an eclectic mix of approaches to find an effective pathway towards healing work. They all believed it was important to look past the initial presentation of things such as anger issues or substance abuse problems and set aside pathologizing diagnoses and past failed approaches to treatment so that they might look at survivors with a new perspective, and consider how for many their challenging way of being in the world could be a secondary response to some terrible life circumstance they may have tried and failed to successfully overcome.

This chapter explored seven themes interpreted from the rich and detailed data obtained through semi structured interviews with the study's participants. Themes included: Hidden Survivors; Some Moral Distress; Dangerous Disclosures; Challenging and Renegotiating Masculinity; Misogyny/Fear of the Feminine; Homophobia: Am I a Fag? and Re-Storying Trauma. These themes will be more fully unpacked and analyzed in relation to current literate around experiences of male sexual trauma in the following chapter.

CHAPTER SIX: ANALYSIS AND DISCUSSION

The following chapter will provide an in-depth analysis of the research findings as well as connect the identified themes back to the current literature. The analysis will help further an understanding of how toxic masculine culture impacts the healing process for male survivors of early sexual trauma. I will begin with a brief overview of how this culture intersects with experiences of male childhood sexual trauma. I will then review current therapeutic practices by participants and provide a critique of the medical model of care. I will then discuss implications of silence around this phenomenon and how it renders survivors invisible and examine some of the myths and assumptions that perpetuate stigma related to male childhood sexual trauma. I will critically examine the impacts of sexualized trauma on male identity and explore ways that survivors might work to re-story these experiences. I will also explore ways we might address and combat perceived stigma within health professions around male childhood sexual abuse. Finally, I will consider some possible male-centred responses and alternatives to traditional office-based talk therapy with male survivors.

Toxic Masculine Culture and Challenges to Healing for Survivors

The findings of this research are consistent with literature that describes strong adherence to traditional masculinity as having negative impacts on the ability of young male sexual trauma survivors to access and receive help (Allagia et al, 2007; Gartner, 2000; Gartner, 2017a; Kimmel, 2008; Sorsoli et al., 2008). Furthermore, stigma related to homophobia and misogyny was demonstrated throughout this study as a problematic to male survivors and a significant impediment to disclosure. This was something that was highlighted by all participants in this study, as they detailed difficult feelings held by male survivor relating to fear, shame, guilt, self-loathing, and a sense of isolation. While the impacts of homophobia on male sexual trauma

survivors within the context of membership in heteronormative culture were well detailed in the literature (Allagia, 2010; Gartner, 2000; Kimmel, 2012), issues relating to male childhood sexual trauma specifically to misogyny or anti-femininity are not clearly described. There does exist a large body of research relating misogyny and male violence against women (Augusta Scott et al, 2007; Katz, 2006; Kimmel, 2012) but no real evidence of male sexual trauma survivors and a propensity for violence or hatred towards women as an impact of their abusive experience. But part of the challenge in drawing conclusions or ruling out connections between misogyny and male childhood sexual abuse is the scarcity of research on male survivors, and more specifically the lack of knowledge around male children and youth and their experiences of disclosure, and if, within that disclosure experience, exists a propensity towards misogynistic thinking and a potential for violence towards women. While the research that is available does suggest boys do adopt some anti-femininity rhetoric or position themselves as masculine away from all things feminine (Allagia et al., 2005; Kia-Keating, 2007, Kimmel, 2012), it would be unreasonable to suggest a certain connection between male sexual trauma experiences and misogynistic violence without focused research in the area. In reality, most male survivors do not engage in violence towards women or in homophobic acts, but they do struggle at times in a heterosexist society that condones both misogyny and homophobia and equates femininity with weakness and submission.

However, in the narratives presented here, participants described survivors who conflated characteristics of strength as aggressive acts, which in turn were problematic and often caused harm to survivors and to those closest to them. We have discussed in earlier chapters how men tend to die earlier, suffer more chronic illnesses and experience undiagnosed and untreated mood disorders. Boys and men are often socialized to keep their emotions in check and present as

strong and silent. Strength can be very positive, and not just physical strength, but strength of character, something that might give voice to calling out injustice and racism and misogyny and homophobia and transphobia in society. But when a man cannot and will not show or express his feelings or emotions, or when he cannot ask for help or seek assistance, he will likely suffer for it. When masculinity impacts men in that way, it is then it becomes toxic.

Participants believed the silence that surrounds the phenomenon of male childhood sexual abuse clearly perpetuates stigma, and the stigma related to mental health issues and help seeking behaviour for men and boys compounds the issue even further and contributes to the hidden victims of childhood sexual trauma. Participants also highlighted some profound effects on masculine identity formation, and how this was highly gendered experience for male survivors, something that is not always apparent in therapeutic approaches with boys and men. This was consistent with literature that called for a review of therapeutic approaches under a more male centered lens that might address issues for men and boys in ways that were more culturally meaningful and responsive, and served to engage them more fully in treatment processes (Allagia, 2005; Fisher & Goodwin, 2008; Gartner, 2000; Kia-Keating et al., 2005; Kimmel, 2012; Lisak, 2017).

Aside from the few in this study who work in care areas centered mainly around trauma treatment, participants by and large did not have a great deal of experience working with male children and youth at the point of disclosure, as male childhood sexual trauma was rarely a presenting issue in their mental health clinics and settings. Other factors that participants described, consistent with the literature, include a fear of loss of one's masculinity, a fear of being seen as "feminine" or womanly, and a fear of being perceived as homosexual as delaying or preventing disclosure (Alaggia, 2005, Alaggia & Millington, 2018; Sivagurunathan et al,

2019). The literature also references some significant deep-rooted fears of survivors becoming sexual abusers themselves (Allagia, 2005; Gartner, 2000; Gartner, 2017a; Lisak, 1994; Lew, 2005). However, this was not reported or discussed by any participants in our interviews, and when explicitly asked about reports of survivors presenting fears of becoming potential perpetrators on others, participants did not disclose any real experience with this phenomenon. Participants like Joanna, Shane, Richard, and Adam did describe fears reported in some survivors in managing their anger and emotional regulation and the potential in harm to others in relation to that dysregulation. I would surmise, however, that because there is such strong evidence in the literature that fears of perpetration on others is a significant concern for survivors upon disclosure, participants lack of experience with young males at the point of disclosure might have precluded their encounter with this phenomenon. More importantly, however, I would argue that for a survivor to admit fears of one day becoming a perpetrator of sexual trauma is a very dangerous sort of disclosure, one that would likely be met with considerable stigma, rejection, controversy and fear upon discovery. And so rather than risk vulnerability in admitting such a fear, the survivor is silenced by a community that would not welcome such disclosures.

Participants noted a lack of training opportunities and workshops devoted to treating mental health or trauma issues specific to men and boys, something that was also reflected widely in the literature (Alaggia, 2005; Allagia, 2010; Teram et al., 2006; Sivagurunthan et al, 2019). As a result, therapists may feel under trained, have a lack of confidence or feel unsupported in treating male childhood sexual trauma (Gruenfeld, Willis, & Easton, 2017). Again, this was common in reports from participants who critiqued the lack of available training specific to male childhood sexual trauma experiences, or like Shane who criticized how issues specific to men and boys were minimally attached to training around women and girls and sexual

violence as an afterthought. Participants also critiqued the lack of readily available supports and services, which as Adam and Gillian noted, left survivors who were prepared and motivated to access help without an avenue to receive it.

By treating victimhood as outside of cultural norms of masculinity for men and boys, young male survivors are therefore excluded as victims and constrained by social expectations (Hlvaka, 2017). This study supported the notion that rigid adherence to traditional or hegemonic masculinity – or toxic masculinity – can have a negative impact on the help seeking processes for young male survivors. Male sexual trauma survivors are therefore subverted and separated from the resources they need to process and heal from their traumatic experiences.

Break the Silence/Into the Light

Foucault (1991) suggests that it is through discourse that our view of reality is both created and sustained; in other words, language and the ways and means in which we communicate not only reflect our reality, they create it. As Foucault points out, silence is necessary for the construction of language. Because silence forms as absence through exclusion, what Foucault defines as 'madness', an opposing discourse based on inclusion and representation can be constructed. However, those things that threaten the dominant disclosure are often excluded, devalued or ignored. In this case, experiences of male childhood sexual trauma that position men and boys as far removed from tenets of hegemonic masculinity and therefore they are dismissed, because their very existence poses a threat to that dominant ideals and normative values of hegemonic masculinity. And yet, the 'not said' – those things excluded from dominant discourse – cannot be fully erased or extinguished, and so in their continued existence they also serve to inform what is said. Silence therefore is not necessarily the total absence of voice but rather an inability to be heard, something that can have many causes. As Foucault (1991) tells us,

"there are not one but many silences, and they are an integral part of the strategies that underlie and permeate discourses" (p. 27). Foucault views all discourse as made of up of structures and elements which join together to make a whole. While also an agent of power, silence is one of those discursive elements, and discourse emerges from things said and unsaid, those that are allowed and those that are forbidden. Discourse as a vehicle for power can manifest as the domination of one discourse by another, in an oppositional relationship (Fairclough, 1989) – in this case, that men and boys cannot be victimized – and therefore the , dominated discourse (childhood sexual trauma of men and boys) is silenced, suppressed, and rendered invisible. Discourse around childhood sexual trauma can therefore silence and suppress experiences of men and boys because it goes against dominant discourses that say men cannot be victims and that sexual abuse is something that happens only to women and girls.

In researching and in reviewing the current literature, I found the rampant silences and secrecies that surround the phenomena of male early childhood sexual abuse to be astonishing in its scope and reaches. As a survivor from decades ago, it was disheartening to discover first-hand how little had changed in terms of knowledge related to sexual trauma of males, in the lack of expertise within mental health, and in the available pathways to support and healing over time. Silence around male childhood sexual trauma was central to every conversation I had during this study. Participants noted little capacity to work with male sexual trauma within their own workplaces due to a disparity of skilled knowledge and focused attention, noting, as Peter described, "there currently is no specific centre of expertise or establishment of readily available resources and supports related to male childhood trauma within our systems."

The few and sparse resources available within Halifax Regional Municipality to help address the needs of sexual assault survivors are designed and intended for women, based on research related to specific needs and approaches with female survivors of sexual trauma.

Breaking the Silence: A Coordinated Response to Sexual Violence in Nova Scotia (Nova Scotia Government, 2016) was released in 2017 as the first provincial sexual violence strategy, intended as a foundation for coordinating services to help sexual trauma survivors. However, while it does acknowledge that male survivors do exist and that gendered male-centric services would be appropriate in treating their unique needs, it does nothing to define explicitly what those services might be, and does nothing to further address treatment and support for men and boys. As Adam noted when considering available resources for male survivors: "it's not even that men and boys need help navigating the system pathway...it's that there is NO pathway. The pathway simply doesn't exist".

While the literature says that one in six males experience early childhood sexual trauma (Fisher & Goodwin, 2008), it is important again to reiterate about half of the participants in this study described less experience and overall few opportunities to work with boys and young men directly at the point of disclosure. Rather, a more common occurrence was to work with adult men who were externally driven to engage in therapy, sometimes by life events such as the deterioration of an intimate relationship or the threatened loss of employment and livelihood, and from that point hear them disclose early childhood sexual trauma experiences. Haunted perhaps by troubles with addictive tendencies or explosive bouts of anger or tattered relationships, participants report that these survivors found themselves encumbered by pathologizing diagnoses and treatment pathways that did little to address likely the core issue at hand - the aftermath of their early betrayal. At times, successful adult development is difficult for male survivors

because they are hesitant to trust others or express their personal needs and vulnerabilities, which in turn can make them prone to "shutting down", with a desire to isolate themselves in order to feel safe (Lisak, 1994; Kia-Keating et al, 2005). Although acts of isolation and withdrawal could have served as important for protection in childhood and as means of resistance, these behaviours can become highly counterproductive to positive adult development and healthy relational attachments (Kia-Keating et al., 2009; Schuetze & Eiden, 2005).

In her work with female survivors, Brown (2013; 2018; 2020) speaks of the dangers of trauma talk, meaning that patriarchal assumptions and dominant discourses around violence and trauma create problematic stories that are often unhelpful and blaming and in turn work to suppress the voice of survivors. Female survivors try to make sense of their experiences within dominant social narratives promote dominant oppressive discourses, including disbelief and blame. I argue that this explanation has salient meaning within the context of male survivors and their experiences, because much like the women described in these studies (Brown 2013: Brown 2018), men's stories are constrained by the limited discourses available that actually describe or fit the experience of male childhood sexual trauma (McKenzie-Mohr & Lafrance, 2011). Foucault's (1991) notion of technologies of self is useful here in considering male silence as it relates to their experiences of male childhood sexual abuse. As men, we construct our social identity through discourses that include silence. Silence shields portions of life from view, and in turn men can engage in technologies of the self by choosing to reveal different aspects of their identity dependent on their social location, their circumstance and the context of power relations they find themselves within (Sirotich et al., 2012). Silence then becomes best understood in terms of a plurality of silences which are fluid, permeable, changeable, and with boundaries that can be broken (Sirotich et al., 2012). When survivors silence themselves and their experiences in

order to be compatible or congruent with the messages they receive socially, they can become confused and constrained in relation to their masculine identity.

Breaking the silence and changing that discourse therefore involves risk to one's masculine identity. "In the end, silence seems to be a strategy employed to address a yearning for acceptance, or to ensure survival, but its adoption, ironically, often leads to alienation" (Sirotich et al., 2012, p. 59). Part of the work in supporting survivors is to critically examine ways to lift these veils of silence within our system and bring the sequelae of male childhood sexual abuse experiences out into the open and the light in a more mainstream way, perhaps by first acknowledging and recognizing the reality of the phenomenon and its devastating and silencing impacts, and then seeking ways through consultation and working with male sexual trauma survivors to offer and provide appropriate treatment services and resources to aid in healing and recovery.

Stigma in Mental Health Systems

While the long-term sequelae of childhood sexual abuse for men and women have historically been considered similar (Allagia, 1 2005; Teram et al., 2009), research has identified some important differences, such as the greater propensity for presenting externalizing, acting out behaviours among boys and for internalizing behaviours among girls (Allagia, 2005; Allagia & Millington, 2018; Teram et al., 2009). Researchers also recognize these differences in what appear as presenting externalizing versus internalizing behaviours are also part of a gender bias in health care, where boys are more likely to be identified as aggressive, hostile, and emotionally dysregulated, and girls are more likely to be perceived as sad, anxious, or depressed. Girls are also challenged and encumbered with the many trappings of a cultural dominant discourse where, and as Marion Brown (2011) tells us:

girls grow up in a hostile cultural climate that is oppressive to females, sexually charged, and dangerous, circumstances that break them of their pre-teen confidence and splinter their authentic selves into subservient, depressed and alienated versions of the self (p. 111).

Another major difference, and arguably the most impactful with possible consequences on male health, is the incongruence between male role expectation and the experience of victimization, and it is this disconnect that creates barriers to the acknowledgment of sexual abuse by male survivors themselves and by the health care professionals who work with them (Allagia, 2005; Allagia & Millington, 2018; Gartner, 2017a; Lisak, 1994; Teram et al., 2009). Findings of this study also indicated that mental health professionals may use some skepticism and engage in minimizing and take the experience of male survivors less seriously than that of their female counterparts. Participants like Ben, Gillian, Daniel, Richard, and Adam all spoke of disclosures that were dismissed or minimized in a way that caused survivors to retreat from help seeking. Daniel described a disclosure as something that befalls the entire family, and so for a survivor "to name the abuse meant to risk shaming and alienating everyone around him, and so once that happens and you deal with the ripple effect, you learn not to do it again".

While the many barriers to disclosure of sexual trauma in men and boys is described in the literature and voiced by this study's participants, it is important to reiterate that men's social power and their dominant role in patriarchal society likely contributes to the lack of discourse related to male childhood sexual trauma (Allagia 2005; Gartner, 2000; Teram et al., 2009). Ironically, therefore, it is "this privileged position that makes it more difficult for male survivors to acknowledge themselves as victims and for health professionals to view them as such" (Teram et al., 2009, p. 514). While male survivors do not seek out therapeutic support as often as their

female counterparts, the literature suggests that the number of men and boys seeking help is in fact slightly increasing (Allagia, 2005; Fisher & Goodwin, 2008; Gartner, 2000; Teram et al., 2009). These findings from the literature are not consistent with experiences of participants in this study, who saw men and boys vastly underrepresented in sexual trauma help seeking discourse. However, there could be many explanations for this discrepancy in findings. Many of the studies reviewed were conducted in larger metropolitan areas like New York, Los Angeles, Ottawa, Vancouver, and Toronto, so statistically with the sample size under consideration, more instances of male childhood sexual trauma history would likely appear. Furthermore, these same metropolitan areas house centres of excellence that provide male centric treatment options, which may mean they are more likely sought out by professionals and survivors – for example, one of this country's most recognized leading agencies in male centred approaches to working with men with sexual trauma histories, *Men & Healing: Psychotherapy for Men*, is located in the heart of Ottawa.

However, findings of increased demand for access to services could also be an indicator that eventually more and more male childhood sexual trauma survivors will enter our systems of care, and as health care professionals there is both a need and an onus to build more capacity in male centred approaches and to be better prepared to help male survivors manage their unique trauma experiences. But due to the lack of research specific to health professionals' attitudes and practices towards male survivors, it is challenging to determine the extent to which mental health care systems are prepared (or not prepared) to face this demand. One of the few studies suggested that health professionals (including samples of doctors, nurses, social workers, and psychologists) were unlikely to ask male patients about abuse if it was not indicated (Lab et al., 2000), despite evidence pointing to the prevalence of histories of abuse in those with psychotic

disorders (Gartner, 2000; Gartner, 2017a; Lisak, 1994). This arguably indicates that some professionals are not fully aware of the correlations between sexual abuse and mental health and its many impacts. The literature indicates that most professionals have little specific training or experience assessing and treating male sexual abuse survivors (Allagia, 2005; Gartner, 2000. 2017b; Teram et al, 2009), which would also be consistent with the experiences of this study's participants. Furthermore, studies show that mental health professionals do not consistently screen for early childhood sexual trauma with boys and young men, and that there exists a tendency to underestimate the prevalence of male childhood sexual trauma and its long-term effects (Allagia, 2005; Gartner, 2000; Gartner, 2017b; Lisak, 1994).

Moral Distress/Moral Courage

Participants worked in what they themselves described as "pathologizing" and problem based medical models of mental health care, overly focused on assessment, diagnosis, and labelling. Those in community mental health clinics described an almost endless cycle for trauma survivors, regardless of gender, of seemingly inhumane intake processes where they must repeat their traumatic histories over and over. From that point, if allowed entry into service, they were offered a short-term intervention that barely scratches the surface of the challenges they faced before being discharged, then only to start the process over again a short time later with some newly arisen crisis. Participants critiqued current mental health systems for their over reliance on brief intervention therapies, processes that were often, as Bryce noted earlier, traumatizing in terms of their exposure, and problematic approaches that served as what Ben described as "band aid solutions to longer term problems". They described a tension sometimes in working with survivors and treating things like substance abuse issues, as Daniel noted, "as a disease versus a symptom and perhaps a maladaptive way or means of coping" with some larger unresolved

trauma. As Brown and Stewart (2007) note, when treating addictions as a primary disease, the possibility that substance misuse is a secondary response to trauma, depression or anxiety is often totally discounted.

Despite the many challenges' participants faced in reconciling these discourses in mental health work, they described active ways they resisted and advocated for positive change in their work with trauma survivors. Participants presented complex case formulations to team leaders and managers and pushed for longer term sessions and shorter time between visits. They also collaborated through forms of clinical trauma supervision where they present challenging cases and discuss treatment formulation amongst peers and colleagues. Meanwhile, those in private practice, as they described earlier, often offered sessions at reduced rates or on a sliding scale, recognizing how the financial constraints of paying for therapy may place it out of reach for many.

Participants' experiences were consistent in their description of current mental health systems' propensity to conceptualize extreme behaviours and distress as symptoms of mental illnesses, rather than view them as occurring within the context of past or current trauma (Sweeney et al., 2016). As Herman (2015) tells us, most survivors of prolonged, repeated childhood trauma never come to psychiatric attention and those who do recover are often left to do so on their own. Survivors who become patients can present with a bewildering array of symptoms including depression, anxiety, problems with anger, extreme phobias, disordered eating, insomnia, sexual dysfunction, self-harm, suicidality, and substance addiction which at times appears at a higher level of distress than "typical" patients (Briere & Scott, 2015; Herman, 2015). Participants also endorsed what Herman (2015) describes as a "disguised presentation" – symptoms such as difficulty in interpersonal relationships, troubles with sexual intimacy, or

repeated victimization where neither the patient or therapist recognize potential linkages between presenting problems and a history of traumatic experiences. Survivors are therefore "frequently misdiagnosed and mistreated. Because of the number and complexity of their symptoms, their treatment is often fragmented and incomplete" (Herman, 2015, p. 123). This corresponded with findings in this study from participants such as Peter and Gillian, who used words to critique approaches such as "piecemeal", "hurried", and "fragmented".

Participants also acknowledged another persistent theme in the literature in that sexual trauma survivors can accumulate many different psychiatric diagnoses before underlying root causes of complex trauma recognized, identified, and treated (Briere & Scott, 2015; Herman, 2015; Van der Kolk, 2014). Borderline Personality Disorder (BPD) was the one most commonly cited diagnosis by participants, something Herman (2015) says "is frequently used within the mental health professions as little more than a sophisticated insult" (p. 125). BPD is usually characterized as challenges with emotional mood swings, impulsive and self-destructive behaviour, difficult relationships and attachments, fear of abandonment and isolation, self-harm, suspicion or paranoia, and feelings of disassociation. Clinical observations articulated decades ago by Judith Herman and Bessel van der Kolk note that BPD and its identifiable traits are much more representative of the complex sequelae of early childhood interpersonal trauma in which challenges with emotional dysregulation plays such a central role (Ford et al, 2018; Herman, 2015; Van der Kolk, 2014). As Brown (2020) notes, women who receive a borderline diagnosis are seen as "angry, non-compliant, resistant, and attention seeking, experience ongoing suicidality and as such are too often written off as being beyond help" (2020, p. 84). She also notes that little effort is put into understanding why they function this way in the world, and little psychotherapy is ultimately offered. The voices of those diagnosed with "borderline" have not

been heard, and more importantly their relationships have been let down, and most of their interactions with professionals now reinforce their thinking that people cannot be trusted or are incapable of understanding them and will ultimately abandon them (Brown, 2020). Ben says he believes BPD:

is a highly controversial diagnosis that needs to be retired, because it is stigmatizing and – and I believe serves as code for someone's history with complex trauma. So, if a client walks in with that diagnosis, already I know what to look for in working with them.

Eclectic approaches are not uncommon among social workers (Coady & Lehmann, 2016), and participants in this study described their philosophical approach as eclectic and drawn from various sources. As discussed earlier by participants, a significant part of what restricts and constrains a guiding therapeutic approach for some is the limits and confines of the medical model and its problem/disease centered focus. With the deeply embedded brief solution focused emphasis of current models of care and treatment, the ability to slowly build rapport and unpack the trauma narrative becomes potentially constrained and restricted, which then has harmful implications for the healing process of trauma survivors.

Further critique of current models of care by participants was consistent with literature that centred around the lack of training and guidelines for practice with male survivors, and that available treatment services were designed primarily to meet the unique needs of female survivors (Easton, 2012; Gartner, 2018). Participants also identified a gender bias in trauma work that acted as a barrier at times to identifying, assessing and treating child sexual abuse in male clients (Easton, Saltzman & Willis, 2014). Specifically, therapists and health care professionals did not initially screen for past sexual trauma with male clients, partly due to the gendered nature

of the discourse around early childhood sexual abuse being something that happens mainly to girls (Allagia, 2005; Gartner, 2017a, 2017b).

Masculine Identity and Male Privilege

Patriarchal society demands that men live up to some ideal of traditional masculinity often far removed from their realities, and this divide is particularly true for male survivors of sexual trauma (Kia-Keating et al., 2005; Sorsoli et al., 2017). This dominant discourse around expectations of masculinity and the ways men encounter it in the sociocultural domain may be one of the major differences between male and female survivors in terms of disclosure (Kia-Keating et al., 2005; Sorsoli et al, 2017). For men and boys, sexual trauma experienced against dominant discourses of hegemonic/dominant masculinity are understood by many survivors as weakness and victimization, equate the experience as feminine, and thereby signifying their loss of masculine identity.

There was a gendered difference in this study in how participants approach the work of tackling male gender constructs and helping clients unpack their meaning. Specifically, female participants like Kate, Susan, Gillian, and Joanna were more likely to work with clients to discover and explore incongruities between traditional forms of masculine and their own lived experiences and offer empathy and a great deal of validation for those who had endured or managed a stricter enforcement or enactment of masculine codes. They approached the work with curiosity, and often used terms in our discussions like "tragically limiting" and "an impossible situation to be in" when describing the challenges of surviving sexualized trauma while navigating traditional masculine norms. Most of the female participants described working with clients with a very limited range of emotional expression when detailing their lives – as Joanna described it, "things were very black and white, you could be angry or you could be

happy, and that was it....it was like there was no recognition of the subtleties or of all the greys in between".

On the other hand, male participants, while recognizing the legitimacy of the discourse around more dominant forms of traditional masculine behaviour, were quick to defend certain constructs of masculinity, and declare what masculinity could be, or perhaps more accurately define what it need *not* be. For the male participants in the study, specifically and carefully unpacking a multitude of masculinity types was deemed a much more helpful starting place in the therapeutic process. Adam and Nathan, for instance, were often much more hesitant to label certain forms of masculine behaviour as "toxic", but they did acknowledge and recognize the impact stringent adherence to norms might have on clients, including how this rigidity might exclude some from being able to actively seek out and participate in treatment. Instead, they advocated for promoting and unpacking positive aspects of traditional masculinity such as self-reliance, strength of character, with focus on recognizing a sense of vulnerability, and a greater range of emotional expression.

An interesting finding came from three of the participants that identified in their personal lives as members of the LGBTQI2S+ community. Their positioning straddled a border between those described above, by strongly identifying and accepting those limitations around normative expectations of hegemonic masculinities, while also acknowledging the space to resist and renegotiate new forms of masculinity. I would argue that the difference in insight, as opposed specifically to their male heterosexual counterparts in this study, is likely related to challenges gay men face navigating, as part of their membership in a culture of subordinated masculinity within the context of a heteronormative and heterosexist society.

While participants expressed beliefs that while many men might not engage in sexist or

homophobic/transphobic beliefs, the indelible influence of those beliefs on masculinity constructs was undeniable. Male privilege, in the sense of the societal power allotted to most men and boys over women, and the cultural bias towards a heteronormative, cisgender idealized masculine norms stood in stark contrast to the experience of victimization of sexual trauma for male survivors. This was also consistent with research that demonstrates how male privilege is intertwined complexly with the maintenance of strict gender codes of conduct or masculine scripts of behaviour, and how that can then serve to restrict and constrain healthy functioning in men and boys (Allagia, 2005; Dorais, 2002; Gartner, 2017a; Gartner, 2017b). All participants spoke to the problematic tensions that arose from strict adherence to gender norms or masculine scripts which often resulted in men and boys being challenged to express vulnerability and the need for support. All participants used terms like "masculine scripts", "masculine norms", or "the man box" to describe rigid characteristics of masculinity that require stoicism, a denial of pain, and an ability to ask for help, and noted, as Adam and Bryce both stated, they are "so subversive" men and boys do not even recognize their influence or their harm. The findings of this study would therefore underscore the need to help survivors address and, at times, challenge patriarchal sexist and heteronormative beliefs that remain, for many, deeply ingrained in our culture.

Participants described working with survivors who struggled to reassert or re-establish their sense of masculinity in what survivors described as hyper masculine ways, which is consistent with toxic masculine culture under examination in this study. These included observations like engaging in violence or disruptive behaviours, remaining emotionally aloof and detached, and promoting misogynistic and homophobic beliefs. The literature here suggests that young male survivors intuitively recognize some of the societal stigma associated with same-sex

acts, and as a result seek to disassociate themselves from anything that symbolizes femininity or homosexuality because it serves as a threat to the dominant discourse of masculinity (Allagia, 2005; Gartner, 2000; Hlavka, 2017). Some survivors than go on to internalize some of these more hyper-masculine beliefs and act them out, possibly as a means of avoiding further stigma or self-ridicule (McGueffy, 2008). Furthermore, boys that are perceived as gay are stigmatized, while an acceptable masculinity (one that conforms as "masculine enough") can be performed and achieved through homophobic behaviour (Pascoe, 2007).

Much like female survivors, male survivors experience personal and interpersonal factors that impact their ability to disclose and engage in treatment, such as feelings of deep shame or guilt, sharing a sense of blame, or feeling they will not be believed (Allagia, 2005, Kia-Keating et al, 2009). Being labelled a victim for male and female survivors can be understood as shameful and stigmatizing. But what becomes much more damaging to male survivors are those sociocultural factors that dare to wrongly equate victimhood to femininity or impose the question of whether this abusive experience means they can now no longer stake claim to both their masculine and heterosexual identities (Gartner, 2000; Lisak, 1995). Research clearly shows that male survivors struggle with the expectations of traditional hegemonic masculinity, particularly in regard to toughness, stoicism and sexual prowess, and they can also find ways to successfully renegotiate these experiences while keeping their masculine identity intact (Allagia, 2005; Gartner, 2017b).

Victimization disrupts masculine ideologies related to power and control and uncovers unexplored terrain related to victimhood and vulnerability. But the challenge here is that those struggles lie not only within the male survivor but also within the hegemonic and heteronormative culture that responds to his needs. A good deal of research on male rape describes male survivors as routinely judged for having failed in their masculine duty to protect

themselves (Hlvaka, 2017; Stermac et al., 2004). Males are also viewed to generally be less traumatized by rape then female survivors, and their sexual orientation is often called into question (Denov, 2004; Hlvaka, 2017; Sternac et al., 2004). If a female is a perpetrator, claims of rape are often dismissed or minimized due to cultural myths that men are the sexual aggressor and sex is always welcome (Hlvaka, 2017). Male victimhood, therefore, becomes nearly incomprehensible. I argue that part of the key therapeutic work with male trauma survivors involves going beyond unpacking gender constructs and actively challenging these stigmatizing cultural narratives of sexist and heteronormative beliefs that facilitate feelings of guilt and shame that serve to hinder and interrupt help seeking and healing processes.

This study is consistent with research that demonstrates that part of healing for men involves successfully engaging with traditional expectations of masculinity while allowing for the experience of feelings and attributes that oppose certain masculine ideals; therefore, part of the work for survivors is to both contain and resist traditional roles in order to heal (Allagia, 2005). When working with boys and men, mental health professionals should seek to address issues of privilege and power related to misogyny, homophobia and transphobia. Male privilege is often invisible to most men, yet they can become aware of it through a variety of means, such as therapy, mentorship, school-based education, and personal experience (Kilmartin et al., 2013; Kimmel, 2008; O'Neil, 2015). Men who understand their privilege and power may be less apt to rely on power, control, and violence in their relationships (McDermott et al., 2012; Schwartz et al., 2004). Some research shows that men tend to overestimate the degree to which other men hold sexist or homophobic beliefs, and that developing awareness of this discrepancy may reduce that inaccuracy (Kilmartin et al., 2013; Kimmel, 2008). Mental health professionals can also work to help men and boys develop awareness of systems that assume cisgender heteronormative

masculinity expression is the expected norm and identify how they may have potentially been harmed by societal or cultural discrimination or oppression. Finally, I would also argue that it is crucial for mental health professionals explore their own perceptions and biases and to understand that, although not all boys or men hold sexist or homophobic beliefs, these discourses are largely ingrained into our society and experienced each and every day, and as a result are imperative to address.

Challenging Misogyny and Homophobia

Studies on male sexual assault demonstrate how men "confront a set of stigmatizing cultural narratives that contribute to a unique sense of shame. Male sexual victimhood is incomprehensible because it contradicts cultural ideas of what it means to be a man – strong, powerful, self-sufficient and impenetrable" (Hlavka, 2017, p. 483). Heteronormative and toxic masculine discourses equate male sexuality with dominance, aggression and desire, while female sexuality is associated with passivity, vulnerability, and submissiveness (Butler, 2015; Hlakva, 2017). Men and boys are not socialized to see themselves as at risk or as particularly vulnerable to sexual assault (Hlvaka, 2017; Lew, 2000). Myths and assumptions around men's pursuit of sex and the sexual act itself as always being welcome impact a male victim's legitimacy and make the experience altogether invisible (Hlvaka, 2017). These statements are all consistent with findings of this study. Participants spoke of survivors raised in a society where sexual abuse seemed outside of the male experience – it was "just a woman thing that happened". These and other myths continue to permeate society and exacerbate the difficulties that survivors have in disclosing their experiences, which can serve to increase stigma, hinder the development of appropriate services and impact further research on the phenomenon (Stermac et al., 2004).

Anti-femininity, or men's avoidance of all things related to femininity, refers to men's desire to avoid being perceived as feminine by abstaining from any actions, thoughts, and feelings that are commonly associated with femininity (e.g., avoiding the colour pink, equating crying or vulnerability as weakness). Anti-femininity also involves a fear of traditional feminine values and behaviours (e.g., appearing weak or docile), and therefore encourages restricting one's emotions and portraying a façade of toughness (O'Neil, 2013; Zurbriggen, 2010).

Internalization of anti-feminine norms and attitudes serves as a necessary component of male socialization in order to achieve perceived masculine dominance in society (Murnen et al., 2015; O'Neil, 2013). As a result, adherence to anti-femininity norms are associated with the devaluation of women because femininity as a whole is seen as inferior and less desirable (Murnen et al., 2002; O'Neil, 2013; Zurbriggen, 2010). There is a significant amount of literature that suggests the internalization of antifemininity has been associated with men's perpetration of sexual violence against women (Lippa, 2008; Thompson & Cracco, 2007; Young et al, 2020).

Philosopher and writer Kate Manne (2015) proposes a definition of the term misogyny that provides a conceptual distinction between what she calls a naive conception and a feminist account of misogyny. The former refers mainly to individual agents – typically, but not necessarily, men – who may feel hatred or hostility towards women generally, "simply because they are women" (Manne, 2015, p.1). A feminist account of misogyny denotes the "system which operates within a patriarchal social order to police and enforce women's subordination, and to uphold men's dominance' (p. 2). Instead of considering the role of the individual, this definition emphasises the role of social structures in the production of misogynistic attitudes that serve a patriarchal ideology. Writer and activist Gillian Serano (2007), in her exploration of the experiences of transwomen, describes misogyny as steeped in assumptions that femaleness and

femininity are both inferior to and exist primarily for the benefit of maleness and masculinity, while also insisting that male and female roles are rigid and fundamentally different, and feminine traits weak and inferior to masculine characteristics.

As discussed earlier, literature that speaks to myths and assumptions on male sexual assault in adult populations' show that male victims are sometimes judged because they are seen as failing in their masculine duty to defend themselves from attack (Hvlaka, 2017; Stermac et al., 2004). Men are therefore seen as culpable, and for this reason their experiences perceived as less traumatizing than female counterparts, and are often assumed to be gay (Hvlaka, 2017; Stermac et al., 2004). These myths contribute to a cultural acceptance of sexual violence and a victim blaming narrative that serves to dismiss male victimization while hiding the very real effects of that victimization, such as shame, stigma, depression, anxiety, substance abuse and suicidal ideation (Dube et al, 2005; Gartner, 2000, 2017a; Hvlaka, 2017; Lisak, 1999).

Myths associated with male childhood sexual abuse, such as survivors are or will become gay, further marginalizes young men who are already struggling to process these experiences (Easton, 2012) and removes them further from capacities to heal. Homophobia – the culturally produced fear of and prejudice against gay people - is a central organizing principle of a cultural definition of manhood (Kimmel, 2015). Homophobia is more than the irrational fear of gay men, and more than the fear that that one might be perceived as gay. "The word 'faggot' has nothing to do with homosexual experience or even with fears of homosexuals. It comes out of the depths of manhood: a label of ultimate contempt for anyone who seems sissy, untough, and uncool" (Leverenz, 1986, p. 455). Homophobia is the fear that others will unmask and emasculate men and reveal that they do not measure up to manhood. Fear leads to shame, because in acknowledging that fear of not measuring up, men find proof of the limits of their masculinity

(Kimmel, 2008). Men can therefore weaponize homophobia, wielding it against any perceived threat of humiliation and the risk of emasculation in the eyes of other men.

Many studies mention the fear of survivors of childhood sexual abuse in their likelihood to engage in and perpetuate a cycle of abuse as being a significant barrier to disclosure. This is despite the fact that only a small minority of male survivors of sexual abuse go on to abuse children themselves (Easton, 2012; Lisak, 1999). This is another issue men and boys often must tackle as opposed to female survivors, making it more challenging for them to come forward and disclose, because societal discourses appear more willing to believe male victims will go on to perpetrate, even though research on perpetration does not see a significant correlation between victim becoming future abuser (Allagia et al., 2005; Fisher & Goodwin, 2008; Gartner, 2000, Lisak, 1994).

Another common myth or assumption cited by participants and discussed extensively in the literature is what Fisher and Goodwin (2008) refer to as the myth of complicity, wherein survivors struggle with deep feelings of shame centred around the physiological response and sexual arousal they may have experienced as part of the abuse. Developing an erection or experiencing ejaculation during a sexually abusive experience serves as one of the most confusing and distressing aspects of sexual abuse for men and boys. Trauma, confusion, and arousal can leave men and boys with feelings of shame and disgust at themselves and their body response. Richard spoke of this as "body betrayal" with the survivors he works with and how he provides psychoeducation around physiological responses. For many, Richard says, "the fear of touch by another man is a real thing, and homophobia is a huge part of that whole dynamic".

Sexual arousal is a powerful, involuntary physiological sensations, and when experienced can lead men to have questions about sexuality. Nathan and Kate spoke of how they carefully

unpack arousal states survivors might have experienced, and that the visible indication of an erection did not mean they invited or welcomed the abusive experience. Among many gay and bisexual male survivors, however, a common discourse persists that the experience is responsible for influencing or determining their sexual orientation (Allagia et al., 2007; Gartner, 2018; Lisak, 1999). This is described as an area of some controversy in the literature due to the high prevalence of childhood sexual trauma among gay and bisexual men, which would seem to indicate some correlation (Cassese, 2000; Dorais, 2000; Fisher and Goodwin, 2008; Gartner, 2017b). However, Gartner (2018) notes that most sex researchers believe that predominant sexual orientation is established before early to mid-childhood, while most sexual abuse of males occurs more commonly after this period (Fisher & Goodwin, 2008; Gartner, 2000; Lisak, 1994). As some study participants like Shane and Gillian noted, I would argue that gay and bisexual boys, and those questioning their sexual identity, are likely much more vulnerable to being abused by a potential perpetrator as they are already a marginalized and oppressed group. Five of the participants spoke of their belief that many of their LGBTQI2S+ clients had experienced some form of early childhood sexual abuse. Shane and Daniel believed that "almost all have experienced some form of sexual trauma", while Gillian spoke of queer children as "fairly frequently marginalized, often growing up in homophobic environments, and often very vulnerable and susceptible to harm from others". Gillian saw this particularly with young gay males who were "at times exploited by older men, and unaware of the power differential at play". Research shows that sexual predators are more likely able to detect and exploit a sense of vulnerability, while an isolated and distressed sexually questioning youth may welcome some aspects of that attention and interest to overcome their sense of aloneness (Fisher and Goodwin,

2007). However, Richard had strong feelings that the reason LGBTQI2S+ youth were actually overrepresented in the data was because they were much more likely to initiate disclosures:

I simply believe queer men are more likely to disclose. They are coming out of the closet as a sexual assault survivor, which is like their second or third time coming out – whereas straight men, they have much more of a harder time squaring that off with their heterosexuality.

The literature demonstrates how male victimization can come to equate a loss of a masculine and heteronormative identity for some men and boys (Allagia, 2005; Hvlaka, 2017).

Fisher and Goodwin (2008) tell us that myths and cultural delusions about male sexual victimization are:

generated in the intersection between the traditional male code and the reality of male sexual victimization. Because the latter is utterly incompatible with the former, the delusions act either to deny or minimize the abuse, or else portray it as a failure of masculinity (p. 56).

Overall, participants believed these myths and assumptions were damaging to healing processes, and psychoeducation to inform and help correct some of these contradictory beliefs was crucial in early phases of treatment. Misogyny and homophobia can serve to silence male survivors. It is a silence centred on a fear of reprisal from other men that allows men to walk past a woman being harassed in the street and not call it out, or to listen to a racist or sexist rant and not challenge it, or ignore a gay bashing joke. These fears serve as the sources of men's silences, and men's silence is what keeps the patriarchal systems running and dominates the cultural definition of manhood.

Re-storying Trauma/Reconstructing Masculinity

A crucial dimension of survivors' recovery "is finding a way to 'make sense' of what happened to them in the past, and to make some kind of meaning of the place the abuse has in their current lives" (Grossman et al., 2006, p. 443). However, despite being understood as a marker of resilience and recovery, understanding ways male survivors make sense of their experience is very limited, as is understanding of how that meaning making relates to recovery and ultimately inform practice and treatment approaches (Grossman et al., 2006).

Overall, very little data exist on how men make meaning of childhood sexual abuse, but the research that does address it suggests that gender socialization strongly influences how males construct meaning from their abuse (Gartner, 1999, 2018; Kia-Keating et al., 2005; Grossman et al., 2006). As research literature and popular media indicate, men in modern culture are often socialized to appear emotionally stoic, invulnerable, and physically forceful and aggressive, while often preoccupied with sex and sexuality (Grossman et al., 2006; Mahalik et al., 2003). Taken together, these pressures make it much harder for them to acknowledge their abuse and victimization, gain support for themselves, be open to supportive therapeutic work, or develop a framework of meaning around their trauma experience (Addis & Mahalik, 2003; Grossman et al, 2006; Lew, 2000). When men and boys attempt to process their trauma, they are walking a fine line between a need as a survivor to be vulnerable and share feelings such as shame, guilt, fear, and anger around the abuse, while at the same time navigating pain or discomfort they might feel through their violation of perceived heteronormative and hegemonic/dominant masculine norms.

Cognitively, generating meaning involves using various frameworks of understanding available to us, whether they be psychological, philosophical, or spiritual, to make sense of our experiences (Grossman et al, 2006). Research has emphasized the importance of trauma

survivors developing a personal account of the trauma narrative, sometimes referred to in the literature as account making or healing stories (Grossman et al 2006; Sorsoli, et al., 2008). Herman (2015) believes the active construction of a truthful narrative is crucial to healing as it helps survivors recall the detailed traumatic memory, process and transform their recollection, and mourn their traumatic losses. Findings in this study also placed an emphasis on processing the trauma narrative for similar reasons and noted the importance of, as Susan describes, "helping children and adolescents distinguish between innocuous stimuli that could trigger a traumatic response versus actual real-life experiences of clear and present danger". Kate, Joanna, and Shane also spoke of doing focused work around identifying triggers that might upset and finding ways to use strategies like mindfulness and journaling to remain present and focused on the here and now. Furthermore, they believed it was crucial to identify unhelpful and inaccurate thoughts or distortions related to the traumatic event that may then benefit from further work in correcting or clarifying later in the therapeutic process.

As Brown and Augusta-Scot (2007) tell us, all stories about social life and subjective experience involve interpretation and are reflective of the social processes of meaning making, which in turn are impacted by cultural and historical influences. For survivors, therapy serves "as a site for the deconstruction of experience and, specifically, the meaning of life experiences" (p. 3). If we do not question or unpack these stories through the therapeutic process, we can inadvertently reproduce unchallenged problematic existing stories and reinforce damaging dominant discourses (Brown, 2007). In working with male survivors of sexual abuse, it is important to note, however, that disclosure and a subsequent narrative for male survivors in particular often comes many years after their initial abuse experiences, and can therefore be challenged by the passing of time and the trouble with memory (Sorsoli et al., 2008). Trauma

experiences disrupt time, as survivors can appear numb, disassociated and removed from their experiences. They can live for long periods in an almost paralyzed memory overwhelmed by their traumatic experience (Herman, 2015; Van der Kolk, 2014). As participants noted in this study, many survivors minimize their experience and, while knowing they were somehow harmed, lack awareness that that the experience was an abusive display of power. I would argue that while female survivors face similar challenges, this is a particularly salient point as many studies show (Allagia, 2013; Kia-Keating, 2013; Kimmel, 2008) the length of time between event and closure is often significant for men and boys, and can further cloud a male survivor's recollection and make the entire process that more complicated and uncertain as a result.

Brown's (2018) work regarding women's narratives of trauma has meaning here as well, as all survivors may try to make sense of their experiences within dominant social narratives available to them, which often provide inadequate or inaccurate accounts of their experiences while reifying oppressive dominant discourses. Much like female survivors, the stories men and boys are often left with are unhelpful and full of self-blame and uncertainty. Stories of experience are a temporal phenomenon: previous experiences shape the present, which in turn shapes and influences the future to come (Clandinin & Connelly, 2004). These stories are also part of the landscape in which one lives and its discursive forces, and therefore not separate from larger social stories that are accepted as truth and remain largely unquestioned (Brown, 2007). The space and time between the actual traumatic event and the act of disclosure is often long and convoluted for male survivors, but as both the research and findings of this study suggest, the process of meaning making and re-storying the trauma narrative remain crucial components in promoting healing, recovery, and future wellness.

Foucault (1991) tells us that discourse "transmits and produces power; it reinforces it, but also undermines and exposes it, renders it fragile, and makes possible to thwarts it" (p. 100). In other words, it establishes power relations and imposes limits within society, with a focus here around normative expectations of dominant masculinity, and men and boys learn to recognize those limits and those power relations, even if they in turn limit or restrict them in different ways. Michael White (1995) reminds us that people are involved in the subjugation of their own lives through processes of power, and by way of narrative approaches, one can resist practices of power, such as internalized problem stories. For male sexual trauma survivors, these problem stories manifest as a perceived failure in masculinity due to dominant discourses which can leave men and boys feeling powerless and without agency. In re-storying trauma and reconstructing their unique version of masculinity as one of many forms, a male survivor can take back agency and move towards a more fulfilling and actualized way of being in the world.

Pathways to Recovery

While men's health problems are generally related to a highly complex interplay between basic physiology, environment, and socio-cultural factors such as race, economic status, and sexual orientation as they relate to power and privilege, the literature highlights patterns in gender health disparity that are connected to heightened risk behaviours for men often first observed during early adolescence (Mahalik et al., 2013). As discussed earlier, male gender role socialization often encourages men to adopt masculine belief systems that can serve to promote high risk behaviours involving substance misuse, violence and aggression, sexual compulsivity, self-harm, or eating disorders (Fisher & Goodwin, 2008; Gartner, 2018; Van der Kolk, 2014). Furthermore, the literature demonstrates a clear link between these same resulting high-risk behaviours and long-term physical health consequences like sexually transmitted diseases,

cancer, heart disease, and obesity (Cohen et al., 2017; Currie & Spatz-Widom, 2010; Herman, 2015; Van der Kolk, 2014).

Other studies also illustrated how men and boy are less willing to consult medical and mental health care providers (Addis & Mahalik, 2003), and are much less likely to proactively utilize preventive health care (Courtenay, 2011). However, the more men perceive that their male friends were seeking help either in the form of talking to someone about a troubling problem or getting an annual physical in the last year, the more likely men report having done the same (Mahalik & Coady, 2007; Mahalik et al., 2007). This is consistent with some of the findings of this study, where participants like Adam spoke of informal yet therapeutic spaces where young men and boys are often more open and comfortable to "share vulnerable thoughts and feelings that then lead to disclosing helpful thoughts and feelings which moved through the group like wildfire".

Participants noted how it as not uncommon for survivors to minimize or understate their challenges and present at times as muted and not overly expressive, or to the other extreme, present themselves through acting out and aggressive behaviours so dysregulated it makes it difficult to determine treatment formulation. When treating problematic acting out behaviour of boys and men, such as hyperactivity, aggression, or issues related to substance abuse, mental health professionals tend to focus on addressing deficits rather than looking at potential strengths (Kiselica et al., 2008). Research also suggests that medication may be relied upon over mental health interventions with boys, particularly in regard to Attention Deficit/Hyperactivity Disorder (ADHD) diagnoses (Kapalka, 2008). These findings in the literature were also supported by participants in this study, who expressed further concern that traumatic responses in young survivors could sometimes be misinterpreted and treated as symptoms of ADHD.

Participants strongly felt that services can and should be more adaptive and integrated to the ways men have been socialized (Englar-Carlson, 2014). Furthermore, they believed that mental health professionals should possess a solid awareness of traditional masculine gender role characteristics that could serve to disguise or inhibit the assessment of underlying mental health issues. For example, participants noted the importance in clinical sessions to probe with questions related to mood and anxiety with boys and men and to ask more extensive or clarifying questions when faced with brief or minimal responses. Mental health professionals need to be aware of the relationship between mental health stigma and traditional masculine ideology which could have certain impacts on how men and boys cope with traditionally stigmatized and gendered mental health issues such as depression or anxiety and how they report it. Participants were quick to note that while many men and boys struggle with mood disorders, a common discourse in their settings is that anxiety and depression are far less common as primary concerns in males, a belief they saw as harmful and in need of correcting

While the literature does support some gender-based adaptations for work with men and boys of diverse backgrounds (Englar-Carlson, 2014; Kiselica et al., 2008; Rabinowitz & Cochran, 2002), except in a few cases, participants were unaware of specific work in this regard in their care areas. The participants did much of what the literature suggests for boys and male adolescents - shorter therapeutic sessions that potentially incorporate some movement breaks; the use of art or musical instruments as components of therapy; and the possibility of moving to more informal settings outside the typical office environment with a focus on natural settings like playgrounds, parks, and hiking trails (Addis et al, 2005; Kiselica & Englar-Carlson, 2008).

Clinical engagement with male survivors begins from the moment of first contact and creating a welcoming process at intake is critical, not only because intake sets the tone of

experience for men and boys abused in childhood due to mental health stigma and challenges in help seeking (Fisher & Goodwin, 2008). This again speaks to the crucial need for mental health professionals to be aware of common discourses and myths and assumptions around male childhood sexual trauma, or otherwise they risk missing out on subtle imitations of past trauma a survivor may have kept long hidden (Fisher & Goodwin, 2008; Gartner, 2000; Lisak, 1994).

When struggling with uncertainty and apprehension around addressing their trauma, and after struggling alone for some time, male survivors can present with a deep sense of hopelessness (Gartner, 2017b). Providing a narrative around recovery to them can help alleviate such fears, such as simply providing psychoeducation in a compassionate way and using statements like "many guys do get better" can alleviate doubts and inspire hope (Gartner, 2017a). The mental health professional's level of comfort with discussing male sexual trauma and recovery is crucial throughout the therapeutic process. As Fisher and Goodwin (2008) note, "having a knowledgeable and trustworthy individual encouraging them to disclose their abuse history represents a significant departure from the isolated way that many of them have previously dealt with their trauma" (p. 177).

The literature describes the benefit from some male survivors to participate in group therapy sessions as a means to counter the negative impacts of male socialization that could potentially impede treatment progress (Fisher & Goodwin, 2008; Gartner, 2000). Some benefits described in the literature include: providing a welcoming, safe space to express anger and pain; countering stereotypes regarding male trauma survivors; allowing for discussion related to a variety of difficult sexual issues and experiences, including potential confusion re orientation; receiving validation from other men and boys in their shared experience; and working together to

redefine definitions of masculinity (Dorais, 2000; Fisher & Goodwin, 2008; Gartner, 2000; Lisak, 1994). This was consistent with findings from this study, but participants encouraged starting much earlier than at a point of intervention regarding sexual trauma, through engaging boys at school and boys and men in extra-curricular group activity settings. They favoured and in some cases worked to create what they described as "small't' therapeutic spaces" in these areas where boys and men could be encouraged to be more open in terms of vulnerability and work on deconstructing any problematic or harmful messages they received or carried related to traditional masculine norms. However, to recognize the relative advantages of group therapy is not to privilege it exclusively over individual therapy (Fisher & Goodwin, 2008) and ultimately what matters most for the survivor is allowing him to discover the best fit.

Four of the participants described aspects of outdoor therapy as an important and at times essential focus to their practice. Outdoor therapies have been shown to provide many positive outcomes in youth dealing with mental health disorders (Ungar et al., 2005). Techniques that are often implemented in order to foster positive change within clients including the use of small groups, natural environment settings, creating positive and supportive atmospheres, employing healthy risk-taking, and offering relational supports (Ungar et al., 2005). The challenging but supportive environment of outdoor group therapy in particular seems to suggest an opportunity for survivors of traumatic experiences to reduce symptoms of trauma and provide opportunities for healing, growth, and the fostering of resilience through their connection with others (Ungar et al., 2005). Providing aspects of outdoor therapy that foster skills connected to strengthening resilience has some potential therefore of providing therapeutic benefit to young men and boys, even those not actively engaged or actively seeking treatment for trauma. A key benefit to any outdoor therapy program is the natural environment and the benefits that are inherent to it.

Therapists might often encourage meaning making by utilizing the outdoors as a metaphor for challenges clients may face in life (Taylor et al., 2010). For example, Nathan described working with a client who compared a wounded animal they came across on a hike to his challenges with his traumatic past, which then served as a "bit of a watershed moment in terms of his understanding what happened wasn't his fault".

Participants noted how traditional methods may at times prove problematic with an emphasis on what the literature describes as feelings, vulnerability and owning up to possible dependency (Gartner, 2018; Rochlen & Rabinowitz, 2014). However, as noted earlier, sometimes therapists can make harmful assumptions that men and boys are unable to express emotions fully which may then serve to preclude them from therapy (Mahalik et al., 2012). Mental health professionals can also misdiagnose boys and men because they do not interpret acting out or externalizing behaviour with depression and anxiety - symptoms that are potentially related to or secondary to past trauma (Addis, 2008; Gartner, 2000; Lisak, 2017) This is further compounded by constraints participants have already described regarding medical models of care which may serve to limit or inhibit a full and complete assessment.

Something that resonated throughout these discussions related to treatment options was the emphasis on choice being offered to male survivors, with a recognition that what worked for one might not work for another, and alternatives were therefore key to a more successful outcome. Participants did not advocate for one treatment modality over another, nor did they debate challenges between office based or outdoor therapy models. What they did emphasize was goodness of fit in terms of treatment options and offering alternatives for male childhood sexual trauma survivors. The creation of open and inclusive therapeutic gender responsive

approaches were considered imperative for all those who continue to suffer or be impacted by early childhood traumatic sexual experience.

Chapter Summary

This chapter provided an in-depth analysis of the research findings and connected identified themes back to current literature relating to masculinity and male childhood sexual trauma. It considered the intersection between toxic masculine culture and experiences of male childhood sexual trauma. It also explored the many implications surrounding the invisibility of male survivors, and examined some of the myths and assumptions that perpetuate stigma related to male childhood sexual trauma, including challenges with homophobia and misogyny. It examined the impacts of sexualized trauma on male identity and explored ways that survivors might work to re-story these experiences. It also considered power and silence and how they related to dominant masculinity and experiences of trauma through a Foucauldian lens. Finally, I considered and critically analyzed some possible male-centred responses and alternatives to traditional office-based talk therapy with male survivors. The following chapter will provide a brief overview of the study including strengths and limitations, implications for social work practice, and an overall conclusion.

CHAPTER SEVEN: CONCLUSION

The purpose of this study was to critically examine the potential impact of "toxic masculine culture" - defined here as a strict and rigid adherence to traditional masculine norms at their most extreme form - on the experiences of young male childhood sexual trauma survivors and their healing processes. The study also examined how we might work to reduce the stigma, shame, and silence that surrounds this phenomenon as a means of offering agency to survivors. It is intended to contribute to public discourse around the experiences of young men and boys and add to the literature around male early childhood sexual trauma. The rationale for the study came from a desire to break the silence that surrounds the phenomena in our current mental health systems and bring attention to a population of young men and boys whose experiences have been under recognized, underreported and understudied. The findings support a more gendered approach to addressing the unique needs of male sexual trauma survivors, something that is defined in the literature as an important part of the healing process for men and boys (Allagia et al, 2007; Gartner, 2018; Fisher & Goodwin, 2008). This focus is arguably almost nonexistent at present in our mental health systems in Nova Scotia, yet this study demonstrated how clinical service providers might be poised to intervene in important gender specific ways in the lives of young male survivors.

Chapter One outlined the framework of this study and provided some background to the research topic. It also offered a working definition of the term "toxic masculinity" that places it within the context of male survivors' experiences with early childhood sexual trauma and abuse. Chapter Two explored relevant literature pertaining to hegemonic or dominant masculinity, "toxic masculinity", early childhood development, male socialization, the sequelae of male childhood sexual trauma, and a brief overview of treatment and healing processes. Chapter Three

contained the methodology section, which described the qualitative research design, including some discussion of theoretical foundations, an outline of the recruitment process, and an overview of ethical considerations. It centred the study as a narrative inquiry, chosen as a means of understanding knowledge and processing experience (Connelly & Clanindin, 2010) as it honoured the lived experiences of the study's participants and their depictions of work with young male survivors of childhood sexual trauma. The study was also intended as a critical discourse analysis, influenced by the work of Foucault, that examined masculinity constructs and their relationship to experiences of male childhood sexual trauma, and deconstructed some of the thought, speech, and behaviour that contribute to social constructions (Fook, 2002), while it illuminated some of the ways discourse both constitutes and constrains (Healy, 1999). Using a Foucauldian approach allowed for some acknowledgement of what has gone unrecognized, unexpressed, or unspoken when considering discourse related to young male survivors.

Chapter Four presented the participants narratives as they describe their work as trauma therapists. The purpose of this chapter was to provide the reader with as much of the rich textual detail and depth of the individual participant's experiences as possible that emerged during the study. Chapter Five discussed the findings of the research in a results section through seven main themes: *Hidden Survivors* (which examined the silence that surrounds the phenomenon that renders its victims invisible); *Some Moral Distress* (which described challenges and constraints participants as therapists face within a medical model of care); *Dangerous Disclosures* (which detailed some of the challenges faced by survivors in disclosing their trauma); *Challenging and Renegotiating Masculinity* (which considered how participants as therapists work with male survivors to deconstruct and reconstruct different versions of masculinity and address issues in toxic masculine culture); *Misogyny/Fear of the Feminine* (which examined how participants

challenge dominant discourses around women as weak and narratives of femininity as bad within the context of patriarchal and misogynistic society); *Homophobia: Am I a Fag?* (which considered myths that perpetuate homophobia following male on male abuse within the context of heteronormative and homophobic culture); and *Re-Storying Trauma* (which detailed how participants as therapists work with clients to understand resiliencies and vulnerabilities and being the healing work)

Participant stories reflected the challenges young males who have experienced childhood sexual trauma face within the context of male socialization masculinity construction, and how by treating victimhood outside of masculine cultural norms they are both constrained and excluded as victims and survivors (Hvlaka, 2007). Study findings suggested that a more gendered based approach to assessment and treatment for male childhood sexual trauma survivors was warranted, with a significant focus on helping those impacted to challenge some of the more damaging or harmful norms of masculinity while working to separate and preserve the positive aspects of masculine identity. The study also saw a crucial component of therapy was related to meaning making and re-storying as means of processing a trauma narrative for survivors who are struggling to heal so as to not continue to operate from a place of guilt or shame. Participants also acknowledged a general lack of fit between traditional standard therapeutic offerings for some men and boys, and advocated for more creative and open ways to approach interventions other than office based talk therapy, including the endorsement of outdoor therapy techniques, incorporating elements of art or music therapy strategies, or the creation of smaller therapeutic group settings.

Chapter Six provided an in-depth analysis and discussion of the study findings. It related research findings consistent with the literature around how toxic masculine culture, with strict

and rigid adherence to traditional masculine norms, can have a negative impact on healing processes for young male survivors. It critiqued the medical model approach to mental health and current brief interventions as limiting and pathologizing in terms of trauma treatment. It also examined implications of silence around this phenomenon and described how myths and assumptions can contribute to and perpetuate the stigma.

The study explored, within the context of male sexual trauma, homophobia and misogyny as an outcome of heteronormative and patriarchal society that subordinates women and all things considered feminine, and expressed an urgency to address these tensions directly with survivors. Misogynistic and homophobic attitudes served to separate men and boys from healing processes and in challenging their masculine identity, potentially influenced negative interactions and engagement with women and LGBTQI2S+ peoples. It also served to critically examine the impacts of sexualized trauma on male identity and explored ways that survivors might re-story these experiences. It highlighted the importance of re-storying trauma, and the process of finding meaning in the traumatic experience as crucial to therapeutic recovery.

The study also argued that gender bias exists among clinicians when assessing for and treating male trauma survivors, and consider ways to correct the bias and address stigma.

Finally, it spoke to themes consistent with the literature regarding the importance of a male centered lens on trauma experience for young men and boys, and more gender responsive alternatives to therapy. Overall, this study demonstrated that childhood sexual trauma has a potentially profound impact on male identity formation, and for young men and boys it becomes a uniquely gendered experience that must be a central focus of our therapeutic processes in order to promote healing.

Strengths and Limitations

Narrative Research as Methodology

The study had a number of strengths and limitations related to the research methodology. A narrative analysis approach was chosen as it allowed for a deeper reflection and understanding of the nature of embedded stories in larger social, cultural, familial, and institutional settings (Clandinin, 2013). The issue of power relations is often a primary concern of narrative research (Clandinin and Connelly, 2000), making the utilization as methodology even more relevant to this thesis topic at hand. By thinking narratively about a phenomenon through research we challenge dominant discourses that view that phenomenon as fixed or unchanging (Clandinin, 2013). As a researcher who serves as both a survivor of childhood sexual trauma and as a mental health clinician with insider with knowledge of mental health systems, it was crucial, as Clandinin (2013) notes, to pay close attention to my positioning within the research, and to understand that as researchers we are all part of the storied landscapes we are studying.

The advantages of this methodology are many in light of understanding and exploring the experience of male childhood sexual abuse under study, particularly through the gathering of thick, rich narratives from service providers that narrative research entails. Narrative research allows researchers to understand experience, and an analysis of people's stories allows for hidden assumptions to come to the surface (Bell, 2002; Clandinin and Connelly, 2002). It also serves to illuminate the temporal aspect of experience, as it recognizes that understanding of people and events can often change over time (Bell, 2002; Connelly & Clandinin). In this case, it served to unpack and better understand the relationship between male childhood sexual trauma and male identity construction and allowed participants to reflect on how their understanding of that relationship changed over the course of their varied careers.

Transferability

A potential disadvantage of narrative research is the challenges in navigating its inherent subjectivity and the privileged voice of the researcher in deciding what stories to highlight and what stories to downplay or omit (Clandinin & Connelly, 2004; Josselson, 2013; Wertz, 2011). Narrative research can be time consuming in terms of the dedicated time in the filed interviewing as well as the amassed amount of rich thick data from interview transcripts and field notes (Clandinin & Connelly, 2004). Indeed, as researcher I transcribed all twelve interviews – which lasted on average two hours each – and compiled these transcripts in addition to field notes recounting all steps of the research process and reflective journals I kept throughout, which amounted to hundreds and hundreds of pages of data. However, I would counter that claim with the plethora of data being, as Josselson (2013) notes, a good problem for a researcher to have in terms of the depth and richness of material available related to the issues under study, and state that the level of reflexivity was necessary in maintaining the integrity of the work. Furthermore, narrative research seems particularly poignant when applied to the experience of early childhood sexual trauma, because, as a methodology, it is defined less by generalities and certainties and more by its imagined alternative possibilities (Clandinin & Connelly, 2004; Wertz, 2011).

Sampling

Purposeful and snowball sampling methods led to the recruitment of participants mostly from the IWK Health Centre or the Nova Scotia Health Authority, and a number of private practice clinicians. Aside from one psychiatrist, participants were divided between social workers and certified counselling therapists. Within their domains, they were representative of some of the specialized clinicians doing trauma work, but other potential participants might have included, among others, psychiatric nurses, occupational therapists, or child and youth care workers. Since

most participants held either a Master of Social Work or Master of Counselling Degree, it is challenging to know whether their views were representative of other trained disciplines in the field. However, of the twelve participants, I was fortunate to have a sample that was relatively diverse across age, gender, race, and sexual orientation.

Participant Interviews

All interviews were semi structured and at times conversational in tone, allowing for rich, thick descriptions of trauma work. In retrospect, I would have liked to have tailored more questions around specific therapeutic approaches and interventions, as when writing the findings I was left with, for instance, some tension around understanding the seeming focus on narrative approaches and how that mapped on to expectations of offering brief solution focused therapy that, as an insider I knew to be time sensitive and limited. A possible interpretation of that tension is that participants were more resistant to systemic constraints by pushing for therapeutic approaches and interventions that were more intensive and time consuming, but this was not fully unpacked and explored in this research.

Implications for Social Work Practice

Social workers pursue social change, particularly on the behalf of vulnerable and oppressed peoples in society. Our work seeks to promote sensitivity and knowledge around oppression, cultural and ethnic diversity, and social justice. Social workers are uniquely positioned to respond to the needs of young male sexual trauma survivors because of our understanding of complex trauma, our insight into intersecting identities, our awareness of the pervasive influence of gender constructs, and our emphasis on strengths-based work and empowerment. Unique to our profession is our ability to take a developmental perspective in understanding the impacts of trauma across the lifespan, and to take into account the intersection

of gender, sexual orientation, culture, race, and family of origin, and the way in which that may shape a trauma survivor's resiliencies and challenges. As a result, I believe more than any other profession, we have the capacity to look at a male survivor's history of trauma through a gendered lens and consider treatment approaches with men and boys as a uniquely gendered experience.

Future Research

The most challenging work I faced over the course of this study was deciding which parts of participant stories to include and which to omit, and when to highlight my voice and that of the voice that of participants. There were various narrative threads that I might have followed but instead chose not to make them a focus. For instance, participants were very interested in engaging in conversations around on issues related to sexual intimacy with adult male survivors, particularly around sexual addiction, sexual exploration, and sexual dysfunction, but due to the focus on adolescent experiences, this was only explored on the surface level. Another interesting narrative was related to experience of trans males- two participants discussed work with trans men who adopted hypermasculine behaviours following transition, and believed they needed to engage in aggressive and sometimes reckless behaviour as a means of staking claim to their masculine identity. Another participant described working with a trans male who had been physically and emotionally abused as a younger child and described their transition as a means of rejecting their feminine identity and adopting one far removed from it. These narratives warrant further research to unpack and explore experiences of sexuality and gender identity.

Other research areas to explore could relate to men and boys' experiences with diagnoses of Borderline Personality Disorder and how described symptomology could mask potential early trauma history. Another point of interest in this study was how participants related misogyny and

feelings of anti-femininity to male sexual trauma survivors, and therefore a deeper exploration of possible correlation between early childhood sexual abuse and perpetration of violence is warranted. I hope to do further research in this area of male childhood sexual trauma, which could include aspects of these issues described. My hope is to work on a major study that will focus on the voices of male survivors as an adjunct to this thesis work.

Conclusion

Throughout the duration of this study, colleagues and friends were often curious and intrigued about my thesis work and its progression until I explained the topic. They were interested in the toxic masculinity part, and curious about the trauma piece, but once I specified my focus was on male childhood sexual abuse, the conversation inevitably took an awkward turn or I would feel a cold, hard wall of resistance. Not only do people struggle with the implications of male childhood sexual abuse as the terrible act itself, but they struggle with the mere mention of it. As Joanna noted in her work, this phenomenon shuts people down at a surface level because they cannot manage the emotional states attached to it.

Fortunately, from their unique perspectives, all twelve of my participants were incredibly insightful, knowledgeable and gracious in sharing their stories and those of their clients. As I noted earlier, I leave this work a better mental health social worker than before thanks to their wisdom, their teachings, and their influence. When I began the research, I had an assumption that I would likely work with eight to twelve women, as women predominantly make up the sample of therapists that was the base of my recruitment. But instead, I started to receive e-mails and calls from men interested in the research topic and hoping for a chance to speak. What was very striking to me, in most cases, were the conversations I had with the male participants. As they shared the pain and the struggles with dominant expectations around masculinity that impacts

their clients, they also shared personal insights of their own challenges and details of their own tragedies. These men spoke of past trauma, of battles with addictions, and the stressors they faced trying to live up sometimes to a hegemonic ideal that was even outside their privileged grasp.

As I reflected on these interviews, I noticed a pattern – each began with some stiff posturing, some awkward small talk, before quickly getting down to business, but then moments after that I could see them each relax and start to unfurl. For some, that allowed me just the slightest peek inside to some of their own fears and vulnerabilities. For others who shared their past triumphs and tragedies more openly, it allowed for a deeply personal look into how they had overcome significant traumatic experiences to now use their gifts and become instruments to help others. By inviting them to have this conversation around toxic masculine culture, it gave them some permission and gave them some space to unpack and explore an issue that, even within this environment of mental health, is a mostly silent discourse.

Originally, I had naively hoped I had hoped to come away with some comprehensive treatment plan for how to work with male survivors – some agreed upon model to follow or sequential steps to take away and "fix" this problem – but I soon discovered that would not happen. However, what I did come away with was hope and determination. Hope came with the creative, passionate, resourceful and meaningful work that is offered by clinicians like these twelve participants, and the way that they challenge and resist systemic barriers in their efforts to provide the best care possible for all trauma survivors. Determination came from recognizing the need, now more than ever, to raise my voice and advocate further for male sexual trauma survivors who continue to be forced into silence. One in six boys are survivors of sexual abuse. These boys often grow up in culture of masculine expectations that can serve to amplify their

trauma and move them away from the capacities they need to heal. This is because those toxic notions of masculinity tell boys to reject those capacities – things like expressing feelings and emotions, accepting help from others, and acknowledging pain and hurt. Masculinity is not innately bad or inherently harmful, but male survivors of sexual trauma suffer in its shadow. I believe we can change that, and it starts with a conversation.

References

- Addis, M. E., & Mahalik, J. R. (2003). Men, masculinity, and the contexts of help seeking. *American Psychologist*, 58(1), 5–14.
- Addis, M. E., Mansfield, A. K., & Syzdek, M. R. (2010). Is "masculinity" a problem? Framing the effects of gendered social learning in men. *Psychology of Men & Masculinity*, 11(2), 77-90. https://doi.org/10.1037/a0018602
- Alaggia, Ramona. (2005). Disclosing the Trauma of Child Sexual Abuse: A Gender Analysis.

 Journal of Loss and Trauma. 10. 10.1080/15325020500193895.
- Alaggia R. (2010). An ecological analysis of child sexual abuse disclosure: considerations for child and adolescent mental health. *Journal of the Canadian Academy of Child and Adolescent Psychiatry*. 19(1), 32–39.
- Alaggia, R., & Millington, G. (2008). Male child sexual abuse: A phenomenology of betrayal. *Clinical Social Work Journal*, *36*(3), 265-275. https://doi.org/10.1007/s10615-007-0144-y
- American Psychological Association, Boys and Men Guidelines Group. (2018). APA guidelines for psychological practice with boys and men.
- http://www.apa.org/about/policy/psychological-practice-boys-men-guidelines.pdf Augusta-Scott, T. (2020). Exploring trauma and masculinity among men who perpetrate
- intimate partner violence. In C. Brown and J. MacDonald (Eds.). Critical clinical social work. Counter storying for social justice. Canadian Scholar's Press.
- Augusta-Scott, T., Scott, K., & Tutty, L. M. (2017). *Innovations in interventions to address intimate partner violence: Research and practice*. Taylor & Francis.
- Banet-Weiser, S., & Miltner, K. M. (2015). #MasculinitySoFragile: Culture, structure, and networked misogyny. *Feminist Media Studies*, *16*(1), 171-174.

- Beasley, C. (2005). Gender and sexuality: Critical theories, critical thinkers. SAGE.
- Beasley, C. (2013). Mind the gap? Masculinity studies and contemporary gender/Sexuality thinking. *Australian Feminist Studies*, 28(75), 108-124.
- Berger, J. M., Levant, R., McMillan, K. K., Kelleher, W., & Sellers, A. (2005). Impact of gender role conflict, traditional masculinity ideology, Alexithymia, and age on men's attitudes toward psychological help seeking. *Psychology of Men & Masculinity*, *6*(1), 73-78. https://doi.org/10.1037/1524-9220.6.1.73
- Berger, P. L., & Luckmann, T. (1966). The social construction of reality: A treatise in the sociology of knowledge (1st ed.). Penguin UK.
- Biblarz, T. J., & Stacey, J. (2010). How does the gender of parents matter? *Journal of Marriage* and Family, 72(1), 3-22. https://doi.org/10.1111/j.1741-3737.2009.00678.x
- Bisin, A., & Verdier, T. (2010). The economics of cultural transmission and socialization. *Handbook of Social Economics*, 339-416.
- Bockting, W., Benner, A., & Coleman, E. (2009). Gay and bisexual identity development among female-to-Male transsexuals in North America: Emergence of a transgender sexuality. *Archives of Sexual Behavior*, 38(5), 688-701.
- Bracken, P., Khalfa, J., & Thomas, P. (2007). Recent translations of Foucault on mental health. *Current Opinion in Psychiatry*, 20(6), 605-608. https://doi.org/10.1097/yco.0b013e3282f08782
- Brannon, Robert. 1976. The male sex role: Our culture's blueprint of manhood, and what it's done for us lately. In The forty-nine percent majority: The male sex role, edited by Deborah S. David and Robert Brannon, 14-15, 30-32. Addison-Wesley.

- Braun, V., & Clarke, V. (2006). Successful qualitative research: A practical guide for beginners. SAGE Publications.
- Briere, J. (1988). The long-term clinical correlates of childhood sexual victimization. *Annals of the New York Academy of Sciences*, *528*(1), 327-334. https://doi.org/10.1111/j.1749-6632.1988.tb50874.x
- Briere, J., & Scott, C. (2015). Complex trauma in adolescents and adults. *Psychiatric Clinics of North America*, 38(3), 515-527. https://doi.org/10.1016/j.psc.2015.05.004
- Brown, C. (2007). Situating knowledge and power in the therapeutic alliance. In C. Brown & T. Augusta-Scott (Eds.). *Narrative therapy*. *Making meaning, making lives* (pp.3-22). Sage.
- Brown, C. (2013). Women's narratives of trauma: (Re) storying uncertainty, minimization and self-blame. *Narrative Works: Issues, Investigations, & Interventions*, *3*(1). Retrieved from https://journals.lib.unb.ca/index.php/NW/article/view/21063
- Brown, C. (2018). The dangers of trauma talk: Counter storying co-occurring strategies for coping with trauma. *Journal of Systemic Therapies*. 37(3), 42–60.
- Brown, C. (2020). Feminist narrative therapy and complex trauma: Critical clinical work with women diagnosed as "borderline." In C. Brown and J. McDonald (Eds.). *Critical clinical social work: Counter storying for social justice (pp.)*. Canadian Scholars Press.
- Brown, C., & Augusta-Scott, T. (2007). Narrative therapy: Making meaning, making lives. Sage.
- Brown, C. G., & Stewart, S. H. (2007). Exploring perceptions of alcohol use as self-medication for depression among women receiving community-based treatment for alcohol problems. *Journal of Prevention & Intervention in the Community*, 35(2), 33-47. https://doi.org/10.1300/j005v35n02_04

- Brown, M. The Sad, the Mad and the Bad: Co-Existing Discourses of Girlhood. *Child Youth Care Forum* 40, 107–120 (2011). https://doi.org/10.1007/s10566-010-9115-5
- Bruner, J. (1991). The Social Construction of Reality. Critical Inquiry, 18(1), 1-21.
- Bruner, J. (1986). Actual minds, possible worlds. Harvard University Press.
- Burns, S. M., & Mahalik, J. R. (2008). Conformity to masculine norms inventory--22 items. https://doi.org/10.1037/t27381-000
- Butler, J. (1988). Performative acts and gender constitution: an essay in phenomenology and feminist theory. *Theatre Journal*, 40(4), 519-531.
- Butler, J. (1999). Gender trouble: Feminism and the subversion of identity. Routledge.
- Buswell, L., Zabriskie, R. B., Lundberg, N., & Hawkins, A. J. (2012). The relationship between father involvement in family leisure and family functioning: The importance of daily family leisure. *Leisure Sciences*, *34*(2), 172-190. https://doi.org/10.1080/01490400.2012.652510
- Caine, V., Estefan, A., & Clandinin, D. J. (2013). A return to methodological commitment:

 Reflections on narrative inquiry. *Scandinavian Journal of Educational Research*, *57*(6),

 574-586. https://doi.org/10.1080/00313831.2013.798833
- Cassese, J. (2000). Gay men and childhood sexual trauma: Integrating the shattered self.

 Haworth Press.
- Chase, Susan. (2005). Narrative Inquiry: Multiple Lenses, Approaches, Voices. Narrative Inquiry: Multiple Lenses Approaches Voices.
- Cicchetti, D., & Rogosch, F. A. (2018). A developmental psychopathology perspective on substance use. *Oxford Scholarship*Online. https://doi.org/10.1093/oso/9780190676001.003.0002

- Clandinin, D. J. (2013). Engaging in narrative inquiry. Routledge
- Clandinin, D. J., & Connelly, F. M. (2004). *Narrative inquiry: Experience and story in qualitative research*. Jossey-Bass.
- Clandinin, D., & Huber, J. (2010). Narrative inquiry. *International Encyclopedia of Education*, 436-441. https://doi.org/10.1016/b978-0-08-044894-7.01387-7
- Clandinin, D. J., Caine, V. & Lessard (2018). The relational ethics of narrative inquiry. NY: Routledge.
- Clandinin, D. J., & Rosiek, J. (2019). Mapping a landscape of narrative inquiry. *Journeys in Narrative Inquiry*, 228-264. https://doi.org/10.4324/9780429273896-15
- Coady, N., & Lehmann, P. (2016). Revisiting the general-eclectic approach. In *Theoretical* perspectives for direct social work practice: A generalist-eclectic approach (3rd ed.). Springer Publishing Company.
- Cochran, S. V. (2010). Emergence and development of the psychology of men and masculinity.

 In J. C. Chrisler & D. R. McCreary (Eds.), *Handbook of gender research in psychology,*Vol. 1. Gender research in general and experimental psychology (pp. 43-58). NY:

 Springer.
- Cochran, S. V., & Rabinowitz, F. E. (2000). Practical resources for the mental health professional. Men and depression: Clinical and empirical perspectives. UAcademic Press.
- Cohen, J. A., Mannarino, A. P., & Deblinger, E. (2017). *Treating trauma and traumatic grief in children and adolescents* (2nd Ed.). Guilford Publications.
- Coles, T. (2007). Negotiating the field of masculinity. *Men and Masculinities*, *12*(1), 30-44. https://doi.org/10.1177/1097184x07309502

- Collin-Vézina D, Daigneault I., & Hébert M. (2013). Lessons learned from child sexual abuse research: prevalence, outcomes, and preventive strategies. *Child Adolescent Psychiatry Mental Health*. 7(22). doi: 10.1186/1753-2000-7-22
- Collins, P.H. (2015). Intersectionality's definitional dilemmas. Annual Review of Sociology, 41 (1-20).
- Connell, R. (1995). Masculinities. Polity.
- Connell, R. W., & Messerschmidt, J. W. (2005). Hegemonic masculinity. *Gender & Society*, 19(6), 829-859.
- Corbett, A. (2016). Psychotherapy with male survivors of sexual abuse: The invisible men.

 Karnac.
- Courtenay, W. H. (2000). Constructions of masculinity and their influence on men's well-being:

 A theory of gender and health. *Social Science & Medicine*, 50(10), 1385–

 1401. https://doi.org/10.1016/S0277-9536(99)00390-1
- Crenshaw, K. (1991). Mapping the Margins: intersectionality, identity politics, and violence against women of color. Stanford Law Review, 43, (1241-1299).
- Currie, J., & Spatz Widom, C. (2010). Long-term consequences of child abuse and neglect on adult economic well-being. *Child Maltreatment*, *15*(2), 111-120. https://doi.org/10.1177/1077559509355316
- Creswell, J. W. (2013). Qualitative inquiry and research design: Choosing among five approaches. Sage.
- Cruikshank, J. (2000). Social life of stories: Narrative and knowledge in the Yukon Territory.

 UBC Press.

- Danchev, D., & Ross, A. (2014). Research ethics for counsellors, nurses and social workers.

 SAGE. https://doi.org/10.4135/9781473915169
- D'Andrea, W., Ford, J., Stolbach, B., Spinazzola, J., & Van der Kolk, B. A. (2012).

 Understanding interpersonal trauma in children: Why we need a developmentally appropriate trauma diagnosis. *American Journal of Orthopsychiatry*, 82(2), 187-200. https://doi.org/10.1111/j.1939-0025.2012.01154.x
- De Boise, S. (2019). Editorial: Is masculinity toxic? *NORMA*, *14*(3), 147-151. https://doi.org/10.1080/18902138.2019.1654742
- Denov, M. S. (2004). The long-term effects of child sexual abuse by female perpetrators. *Journal of Interpersonal Violence*, *19*(10), 1137-1156. https://doi.org/10.1177/0886260504269093
- Denzin, N. K., & Lincoln, Y. S. (2017). The SAGE handbook of qualitative research. Sage.
- Dewey, J. (2008). The early works, 1882-1898: 1895-1898. Early essays. SIU Press.
- Dorais, M. (2002). *Don't tell: The sexual abuse of boys* (2nd ed.). McGill-Queen's Press MQUP.
- Dube, S., Anda, R., Whitfield, C., Brown, D., Felitti, V., Dong, M., & Giles, W. (2005). Long-term consequences of childhood sexual abuse by gender of victim. *American Journal of Preventive Medicine*, 28(5), 430-438. https://doi.org/10.1016/j.amepre.2005.01.015
- Eastman, J. T. (2017). *The southern rock revival: The Old South in a new world*. Rowman & Littlefield.
- Easton, S. D. (2014). Disclosure of child sexual abuse among adult male survivors. *Clinical Social Work Journal*, 41(4), 344-355. https://doi.org/10.1007/s10615-012-0420-3

- Easton, S. D., Saltzman, L. Y., & Willis, D. G. (2014). "Would you tell under circumstances like that?" Barriers to disclosure of child sexual abuse for men. *Psychology of Men & Masculinity*, 15(4), 460–469. https://doi.org/10.1037/a0034223
- Edwards, S. R., Bradshaw, K. A., & Hinsz, V. B. (2014). Denying rape but endorsing forceful intercourse: Exploring differences among responders. *Violence and Gender*, *1*(4), 188-193. https://doi.org/10.1089/vio.2014.0022
- Englar-Carlson, M. (2014). Introduction: a primer on counselling men. In M. P. Evans, T. Duffy, & M. Englar-Carlson (Eds.), a counselor's guide to working with men. John Wiley & Sons.
- Evans, M. P., Duffey, T., & Englar-Carlson, M. (2013). Introduction to the special issue: Men in counseling. *Journal of Counseling & Development*, 91(4), 387-389. https://doi.org/10.1002/j.1556-6676.2013.00108.x
- Fagot, B., Rodgers, C. M., & Leinbach, M. (2000). Theories of gender socialization. In *The developmental social psychology of gender*. Psychology Press.
- Fairclough, N. (2013). Critical discourse analysis: The critical study of language. Routledge.
- Finkelhor, David, and Turner, Heather. National Survey of Children's Exposure to Violence II, 1993-2012 [United States]. Ann Arbor, MI: Inter-university Consortium for Political and Social Research [distributor], 2016-10-11. https://doi.org/10.3886/ICPSR36177.v1
- Fisher, A., & Goodwin, R. (2008). *Men & healing: Theory, research, and practice in working with male survivors of childhood sexual abuse*. Report prepared for the Cornwall Public Inquiry at The Men's Project, Ottawa, Canada.
- Ford, J. D., & Courtois, C. A. (2016). Treating complex traumatic stress disorders in children and adolescents: Scientific foundations and therapeutic models. Guilford Press.

- Foucault, M. (1980). *Power/Knowledge: Selected interviews and other writings, 1972-1977*.

 Pantheon.
- Foucault, M. (1991). Experiences of madness. *History of the Human Sciences*, *4*(1), 1-25. https://doi.org/10.1177/095269519100400101
- Foucault, M. (1991). The Foucault reader. P. Rabinow (Ed.). Pantheon.
- Fraser, N. (1989). *Unruly practices: Power, discourse and gender in contemporary social theory*. University of Minnesota Press.
- Gagnier, C., & Collin-Vézina, D. (2016). The disclosure experiences of male child sexual abuse survivors. *Journal of Child Sexual Abuse*, 25(2), 221-241. https://doi.org/10.1080/10538712.2016.1124308
- Gartner, R. B. (2000). Betrayed as boys: Psychodynamic treatment of sexually abused men.

 Guilford Press.
- Gartner, R. B. (2017a). Healing sexually betrayed men and boys: Treatment for sexual abuse, assault, and trauma. Routledge.
- Gartner, R. B. (2017b). Understanding the sexual betrayal of boys and men: The trauma of sexual abuse. Routledge.
- Gelman, S. A., Taylor, M. G., & Nguyen, S. P. (2004). Mother-child conversations about gender: Understanding the acquisition of essentialist beliefs: I.

 Introduction. *Monographs of the Society for Research in Child Development, 69*(1), 1–14. https://doi.org/10.1111/j.1540-5834.2004.06901002.x
- Gibb, B. E., & Abela, J. R. (2007). Emotional abuse, verbal victimization, and the development of children's negative inferential styles and depressive symptoms. *Cognitive Therapy* and *Research*, 32(2), 161-176. https://doi.org/10.1007/s10608-006-9106-x

- The Good Men Project. (2020). https://goodmenproject.com/
- Griffith, D. M. (2018). "Centering the margins": Moving equity to the center of men's health research. *American Journal of Men's Health*, 12(5), 1317-1327. https://doi.org/10.1177/1557988318773973
- Grossman, F. K., Spinazzola, J., Zucker, M., & Hopper, E. (2017). Treating adult survivors of childhood emotional abuse and neglect: A new framework. *American Journal of Orthopsychiatry*, 87(1), 86-93. http://dx.doi.org/10.1037/ort0000225
- Gruenewald, J., & Kelley, K. (2014). Exploring Anti-LGBT homicide by mode of victim selection. *Criminal Justice and Behavior*, 41(9), 1130-1152. https://doi.org/10.1177/0093854814541259
- Gruenfeld, E., Willis, D. G., & Easton, S. D. (2017). "A very steep climb": Therapists' perspectives on barriers to disclosure of child sexual abuse experiences for men. *Journal of Child Sexual Abuse*, 26(6), 731-751. https://doi.org/10.1080/10538712.2017.1332704
- Gruenfeld, E., Willis, D. G., & Easton, S. D. (2017). "A very steep climb": Therapists' perspectives on barriers to disclosure of child sexual abuse experiences for men. *Journal of Child Sexual Abuse*, 26(6), 731-751. https://doi.org/10.1080/10538712.2017.1332704
- Halpern, H. P., & Perry-Jenkins, M. (2015). Parents' gender ideology and gendered behavior as predictors of children's gender-role attitudes: A longitudinal exploration. *Sex**Roles, 74(11-12), 527-542. https://doi.org/10.1007/s11199-015-0539-0
- Herman J. L. (2015). Trauma and recovery: The aftermath of violence--from domestic abuse to political terror. (2nd ed.) NY: Basic Books.
- Hlavka, H. R. (2017). Speaking of stigma and the silence of shame. *Men and Masculinities*, 20(4), 482-505. https://doi.org/10.1177/1097184x16652656

- hooks, b. (2005). The will to change: Men, masculinity, and love.

 Washington Square Press.
- Iwamoto, D. K., Brady, J., Kaya, A., & Park, A. (2018). Masculinity and depression: A longitudinal investigation of multidimensional masculine norms among college men. *American Journal of Men's Health*, 12(6), 1873-1881.
 https://doi.org/10.1177/1557988318785549
- Jaffee, S. R., Moffitt, T. E., Caspi, A., & Taylor, A. (2003). Life with (or without) father: The benefits of living with two biological parents depend on the father's antisocial behavior. *Child Development*, 74(1), 109-126. https://doi.org/10.1111/1467-8624.t01-1-00524
- Jewkes, R., Morrell, R., Hearn, J., Lundqvist, E., Blackbeard, D., Lindegger, G., Quayle, M., Sikweyiya, Y., & Gottzén, L. (2015). Hegemonic masculinity: Combining theory and practice in gender interventions. *Culture, Health & Sexuality*, 17(sup2), 112-127. https://doi.org/10.1080/13691058.2015.1085094
- Kapalka, G. M. (2008). Managing students with ADHD in out-of-class settings. *Emotional and Behavioural Difficulties*, 13(1), 21-30. https://doi.org/10.1080/13632750701814641
- Kaplow, J. B., Hall, E., Koenen, K. C., Dodge, K. A., & Amaya-Jackson, L. (2008).
 Dissociation predicts later attention problems in sexually abused children. *Child Abuse*& Neglect, 32(2), 261-275. https://doi.org/10.1016/j.chiabu.2007.07.005
- Katz, J. (2006). The macho paradox: Why some men hurt women and how all men can help. Sourcebooks, Inc.
- Kehily, M. J., & Nayak, A. (1997). 'Lads and Laughter': Humour and the production of heterosexual hierarchies. *Gender and Education*, *9*(1), 69-88.

- https://doi.org/10.1080/09540259721466
- Kia-Keating, M., Grossman, F. K., Sorsoli, L., & Epstein, M. (2005). Containing and resisting masculinity: Narratives of renegotiation among resilient male survivors of childhood sexual abuse. *Psychology of Men & Masculinity*, 6(3), 169-185. doi:10.1037/1524-9220.6.3.169
- Kia-Keating, M., Sorsoli, L., & Grossman, F. K. (2009). Relational challenges and recovery processes in male survivors of childhood sexual abuse. *Journal of Interpersonal Violence*, 25(4), 666-683. doi:10.1177/0886260509334411
- Kilmartin, C., Addis, M. E., Mahalik, J. R., & O'Neil, J. M. (2013). Teaching the psychology of men: Four experienced professors describe their courses. *Psychology of Men & Masculinity*, *14*(3), 240-247. https://doi.org/10.1037/a0033254
- Kilmartin, C., & Smiler, A. (2015). *The masculine self* (5th ed.).

 McGraw-Hill Humanities/Social Sciences/Languages.
- Kim, J., & Cicchetti, D. (2006). Longitudinal trajectories of self-system processes and depressive symptoms among maltreated and nonmaltreated children. *Child Development*, 77(3), 624-639. https://doi.org/10.1111/j.1467-8624.2006.00894.x
- Kimmel, M. (2015). The gendered society reader (3rd ed.). Oxford University Press.
- Kimmel, M. S. (2012). Men's lives (9th ed.). M. A. Messner (Ed.). Prentice Hall.
- Kimmel, M. (2008). Masculinity as homophobia: Fear, shame and silence in the construction of gender identity. In H. Brod & M. Kaufman (Eds.), *Theorizing Masculinities* (2nd ed.). SAGE Publications.

- Kiselica, M. S., & Englar-Carlson, M. (2010). Identifying, affirming, and building upon male strengths: The positive psychology/positive masculinity model of psychotherapy with boys and men. *Psychotherapy: Theory, Research, Practice, Training*, 47(3), 276-287.
- Kiselica, M. S., Englar-Carlson, M., & Horne, A. M. (2011). *Counseling troubled boys: A guidebook for professionals*. Routledge.
- Kivel, P., (2011). *Men's work: How to stop the violence that tears our lives apart.* Hazelden.
- Kupers, T. A. (2005). Toxic masculinity as a barrier to mental health treatment in prison. *Journal of Clinical Psychology*, 61(6), 713-724.
- Lab, D. D., Feigenbaum, J. D., & De Silva, P. (2000). Mental health professionals' attitudes and practices towards male childhood sexual abuse. *Child Abuse & Neglect*, *24*(3), 391–409. https://doi.org/10.1016/S0145-2134(99)00152-0
- Lew, M. (2004). Victims no longer: The classic guide for men recovering from sexual child abuse (2nd ed.). HarperCollins.
- Levant, R. F. (2011). Research in the psychology of men and masculinity using the gender role strain paradigm as a framework. *American Psychologist*, 66(8), 765-776.
- Levant, R. F., & Pollack, W. S. (Eds.). (1995). A new psychology of men.

 Basic Books.
- Levant, R.F. & Richmond, K. (2007). A review of research on masculinity ideologies using the male role norms inventory. *The Journal of Men's Studies*, 15(2), 130–146.
- Lippa, R. A. (2005). Gender, nature, and nurture. Routledge.
- Lisak, D. (1994). The psychological impact of sexual abuse: Content analysis of interviews with male survivors. *Journal of Traumatic Stress*, 7(4), 525-548. https://doi.org/10.1002/jts.2490070403

- Lisak, David. (1995). Integrating a critique of gender in the treatment of male survivors of childhood abuse. Psychotherapy: Theory, Research, Practice, Training. 32. 258-269. 10.1037/0033-3204.32.2.258.
- Lisak, D. (2017). Male gender socialization and the perpetration of sexual abuse. *Childhood Socialization*, 311-330. https://doi.org/10.4324/9781315081427-24
- Liu, W. M. (2005). The study of men and masculinity as an important multicultural competency consideration. *Journal of Clinical Psychology*, 61(6), 685–697. doi:10.1002/jclp.20103
- Liu, W. M., & Concepcion, W. R. (2010). Redefining Asian American identity and masculinity.

 In W. M. Liu, D. K. Iwamoto & M. Chae (Eds.), *Culturally responsive counseling*interventions with Asian American men (pp. 127–144). Routledge.
- Mahalik, J. R., & Burns, S. M. (2007). Masculinity and perceived normative health behaviors as predictors of men's health behaviors. *Social Science & Medicine*, *64*(11), 2201-2209. https://doi.org/10.1016/j.socscimed.2007.02.035
- Mahalik, J. R., Good, G. E., & Englar-Carlson, M. (2003). Masculinity scripts, presenting concerns, and help seeking: Implications for practice and training. *Professional Psychology: Research and Practice*, *34*(2), 123-131.
- Mahalik, J. R., Levi-Minzi, M., & Walker, G. (2007). Masculinity and health behaviors in Australian men. *Psychology of Men & Masculinity*, 8(4), 240-249.
- Manne, K. (2015). Ameliorating misogyny. *Oxford Scholarship Online*. https://doi.org/10.1093/oso/9780190604981.003.0003
- Marks, J. L., Lam, C. B., & McHale, S. M. (2009). Family patterns of gender role attitudes. *Sex Roles*, 61(3-4), 221-234. https://doi.org/10.1007/s11199-009-9619-3

- Martin, C. L., Ruble, D. N., & Szkrybalo, J. (2002). Cognitive theories of early gender development. *Psychological Bulletin*, 128(6), 903-933. https://doi.org/10.1037/0033-2909.128.6.903
- McCarry, M. (2007). Masculinity studies and male violence: Critique or collusion? *Women's Studies International Forum*, 30(5), 404-415. https://doi.org/10.1016/j.wsif.2007.07.006
- McCarthy, J., & Deady, R. (2008). Moral Distress Reconsidered. *Nursing Ethics*, *15*(2), 254–262. https://doi.org/10.1177/0969733007086023
- McDermott, R. C., Schwartz, J. P., & Trevathan-Minnis, M. (2012). Predicting men's anger management: Relationships with gender role journey and entitlement. *Psychology of Men & Masculinity*, *13*(1), 49-64. https://doi.org/10.1037/a0022689
- McDermott, R. C., & Schwartz, J. P. (2013). Toward a better understanding of emerging adult men's gender role journeys: Differences in age, education, race, relationship status, and sexual orientation. *Psychology of Men & Masculinity*, *14*(2), 202-210. https://doi.org/10.1037/a0028538
- McKenzie-Mohr, S., & Lafrance, M. N. (2011). Telling stories without the words: 'Tightrope talk' in women's accounts of coming to live well after rape or depression. *Feminism & Psychology*, 21(1), 49-73. doi:10.1177/0959353510371367
- McGuffey, C. S. (2008). "Saving masculinity:" Gender reaffirmation, sexuality, race, and parental responses to male child sexual abuse. *Social Problems*, *55*(2), 216-237. https://doi.org/10.1525/sp.2008.55.2.216
- McMahon, A. (1993). Male readings of feminist theory: The psychologization of sexual politics in the masculinity literature. *Theory and Society*, 22(5), 675-695. https://doi.org/10.1007/bf00993542

- Mesman, J., & Groeneveld, M. G. (2017). Gendered parenting in early childhood: Subtle but unmistakable if you know where to look. *Child Development Perspectives*, *12*(1), 22-27. https://doi.org/10.1111/cdep.12250
- Messner, M. (2004). On patriarchs and losers: rethinking men's interests. *Berkeley Journal of Sociology*, 48, 74-88.
- Messerschmidt, J. W. (2019). The salience of "hegemonic masculinity". *Men and Masculinities*, 22(1), 85-91. https://doi.org/10.1177/1097184x18805555
- Moylan, C. A., Herrenkohl, T. I., Sousa, C., Tajima, E. A., Herrenkohl, R. C., & Russo, M. J. (2009). The effects of child abuse and exposure to domestic violence on adolescent internalizing and externalizing behavior problems. *Journal of Family Violence*, 25(1), 53-63. https://doi.org/10.1007/s10896-009-9269-9
- Nova Scotia Department of Community Services. (2015). *Breaking the silence: A coordinated response to sexual violence in Nova Scotia*. https://www.deslibris.ca/ID/247419
- O'Neil, J. M. (2013). Gender role conflict research 30 years later: an evidence-based diagnostic schema to assess boys and men in counseling. *Journal of Counseling & Development*, 91: 490-498.
- Paechter, C. (2007). Being boys, being girls: Learning masculinities and femininities. United McGraw-Hill.
- Pappas, S. (2019). *APA issues first-ever guidelines for practice with men and boys*. https://www.apa.org. https://www.apa.org/monitor/2019/01/ce-corner
- Parrott, D. J. (2009). Aggression toward gay men as gender role enforcement: Effects of male role norms, sexual prejudice, and masculine gender role stress. *Journal of Personality*, 77(4), 1137-1166. https://doi.org/10.1111/j.1467-6494.2009.00577.x

- Pascoe, C. J. (2007). Dude, you're a fag: Masculinity and sexuality in high school. University of California Press.
- Payne, M. (2014). Modern social work theory. Macmillan International Higher Education.
- Perlman, S. B., Kalish, C. W., & Pollak, S. D. (2008). The role of maltreatment experience in children's understanding of the antecedents of emotion. *Cognition & Emotion*, 22(4), 651-670. https://doi.org/10.1080/02699930701461154
- Peters, J., Nason, C., & Turner, W. M. (2009). Development and testing of a new version of the hypermasculinity index. *Social Work Research*, 31(3) 171-182, https://doi.org/10.1037/t57853-000
- Peretz, T. (2016). Why study men and masculinities? A theorized research review. *Graduate Journal of Social Science*, 12(3), 30-43.
- Pleck, J.H. (1995). The gender role strain paradigm. In R.F. Levant and W.S. Pollack (Eds.), *A new psychology of men* (pp.11–32). Basic Books.
- Pleck, J.H. (1981). The myth of masculinity. The MIT Press.
- Pinnegar, S., & Daynes, J. (2007). Locating narrative inquiry historically: thematics in the turn to narrative. In D. J. Clandinin (Ed.), *Handbook of narrative inquiry: Mapping a methodology*. SAGE Publications.
- Pollak, S. D., Messner, M., Kistler, D. J., & Cohn, J. F. (2009). Development of perceptual expertise in emotion recognition. *Cognition*, *110*(2), 242-247. https://doi.org/10.1016/j.cognition.2008.10.010
- Preves, S. E. (2003). *Intersex and identity: The contested self*. Rutgers University Press.
- Public Health Agency of Canada (2010). Canadian Incidence Study of Reported Child Abuse and Neglect 2008: Major Findings. Ottawa.

- Rabinowitz, F. E., & Cochran, S. V. (2002). Deepening psychotherapy with men. https://doi.org/10.1037/10418-000
- Randall, J., & Munro, I. (2010). Foucault's care of the self: A case from mental health work. *Organization Studies*, *31*(11), 1485-1504. https://doi.org/10.1177/0170840610380809
- Reigeluth, C. S., & Addis, M. E. (2016). Adolescent boys' experiences with policing of masculinity: Forms, functions, and consequences. *Psychology of Men & Masculinity*, 17(1), 74-83. https://doi.org/10.1037/a0039342
- Riessman, C. K. (2008). Narrative methods for the human sciences. Sage.
- Roberts, M. (2005). The production of the psychiatric subject: Power, knowledge and Michel Foucault. *Nursing Philosophy*, 6(1), 33-42. https://doi.org/10.1111/j.1466-769x.2004.00196.x
- Robinson, V. (2003). Radical revisionings? The theorizing of masculinity and (radical) feminist theory. *Women's Studies International Forum*, 26(2), 129-137. https://doi.org/10.1016/s0277-5395(03)00016-5
- Roesler, T. A., & McKenzie, N. (1994). Effects of childhood trauma on psychological functioning in adults sexually abused as children. *The Journal of Nervous and Mental Disease*, 182(3), 145-150. https://doi.org/10.1097/00005053-199403000-00003
- Rogers, W. S. (2001). Theories of child development. *Children in Society*, 202-214. https://doi.org/10.1007/978-1-137-24714-8_22
- Salter, M. (2019, February 27). *The problem with a fight against toxic masculinity*. The Atlantic. https://www.theatlantic.com/health/archive/2019/02/toxic-masculinity-history/583411/

- Sánchez, F. J., Westefeld, J. S., Liu, W. M., & Vilain, E. (2010). Masculine gender role conflict and negative feelings about being gay. *Professional Psychology: Research and Practice*, 41(2), 104–111.
- Schrock, D., & Schwalbe, M. (2009). Men, masculinity, and manhood acts. *Annual Review of Sociology*, 35(1), 277-295. https://doi.org/10.1146/annurev-soc-070308-115933
- Schuetze, P. & Eiden, R. (2005). The relationship between sexual abuse during childhood and parenting outcomes: Modeling direct and indirect pathways. Child abuse & neglect. 29. 645-59. 10.1016/j.chiabu.2004.11.004.
- Schwartz, J. P., Magee, M. M., Griffin, L. D., & Dupuis, C. W. (2004). Effects of a group preventive intervention on risk and protective factors related to dating violence. *Group Dynamics: Theory, Research, and Practice*, 8(3), 221-231. https://doi.org/10.1037/1089-2699.8.3.221
- Seidler, Z. E., Rice, S. M., River, J., Oliffe, J. L., & Dhillon, H. M. (2017). Men's mental health services: The case for a masculinities model. *The Journal of Men's Studies*, *26*(1), 92-104. https://doi.org/10.1177/1060826517729406
- Sirotich, F., Martin, S., & Ruhinda, S., Vaz, J., & Moffatt, K. (2012). Yearning to break silence: Reflections on the functions of male silence. 10.3138/9781442660786-007.
- Sivagurunathan, M., Orchard, T., MacDermid, J. C., & Evans, M. (2019). Barriers and facilitators affecting self-disclosure among male survivors of child sexual abuse: The service providers' perspective. *Child Abuse & Neglect*, 88, 455-465. https://doi.org/10.1016/j.chiabu.2018.08.015

- Solomon, S. E., Rothblum, E. D., & Balsam, K. F. (2005). Money, housework, sex, and conflict: Same-sex couples in civil unions, those not in civil unions, and heterosexual married siblings. *Sex Roles*, *52*(9-10), 561-575. https://doi.org/10.1007/s11199-005-3725-7
- Sorsoli, L., Kia-Keating, M., & Grossman, F. K. (2008). "I keep that hush-hush": Male survivors of sexual abuse and the challenges of disclosure. *Journal of Counseling Psychology*, 55(3), 333-345. https://doi.org/10.1037/0022-0167.55.3.333
- Stainton Rogers, W., & Stainton Rogers, R. (2001). The psychology of gender and sexuality: An introduction. McGraw-Hill Education
- Stermac, L., Del Bove, G., & Addison, M. (2004). Stranger and Acquaintance Sexual Assault of Adult Males. *Journal of Interpersonal Violence*, 19(8), 901-915.
- Suleiman, S. (2008). Judith Herman and Contemporary Trauma Theory. *Women's Studies Quarterly*, 36(1/2), 276-281.
- Sweeney, A., Clement, S., Filson, B., & Kennedy, A. (2016). Trauma-informed mental healthcare in the UK: What is it and how can we further its development? *Mental Health Review Journal*, 21(3), 174-192. https://doi.org/10.1108/mhrj-01-2015-0006
- Swerdfager, T. (2016). Theorizing resistance: Foucault, cross-cultural psychiatry, and the user/Survivor movement. *Philosophy, Psychiatry, & Psychology*, 23(3-4), 289-299. https://doi.org/10.1353/ppp.2016.0033
- Teram, E., Stalker, C., Hovey, A., Schachter, C., & Lasiuk, G. (2006) Towards male-centric communication: sensitizing health professionals to the realities of male childhood sexual abuse. *Issues in Mental Health Nursing*, 27:5, 499-517.
- Thomason, M. E., & Marusak, H. A. (2017). Toward understanding the impact of

- trauma on the early developing human brain. Neuroscience, 342, 55-67
- Thompson, E. H., & Cracco, E. J. (2008). Sexual aggression in bars: What college men can normalize. *The Journal of Men's Studies*, *16*(1), 82-96. https://doi.org/10.3149/jms.1601.82
- Trocmé, N. M., Tourigny, M., MacLaurin, B., & Fallon, B. (2003). Major findings from the Canadian incidence study of reported child abuse and neglect. *Child Abuse & Neglect*, 27(12), 1427-1439. https://doi.org/10.1016/j.chiabu.2003.07.003
- Turner, S., Taillieu, T., Cheung, K., & Afifi, T. O. (2017). The relationship between childhood sexual abuse and mental health outcomes among males: Results from a nationally representative United States sample. *Child Abuse & Neglect*, 66, 64-72. https://doi.org/10.1016/j.chiabu.2017.01.018
- Ungar, M., Dumond, C., & Mcdonald, W. (2005). Risk, resilience and outdoor programmes for at-risk children. *Journal of Social Work*, 5(3), 319-338. https://doi.org/10.1177/1468017305058938
- United Nations. Commission on the Status of Women. (2000). Report on the forty-fourth session of the Economic and Social Council, United Nations
- Valentino, K., Cicchetti, D., Rogosch, F. A., & Toth, S. L. (2008). True and false recall and dissociation among maltreated children: The role of self-schema. *Development and Psychopathology*, 20(1), 213-232. https://doi.org/10.1017/s0954579408000102
- Van der Kolk K. (2014). The body keeps the score: Brain, mind, and body in the healing of trauma. IDreamBooks Inc.
- Waling, A. (2018). Defining "masculinity". In A. Whaling *White masculinity in contemporary Australia (pp.*, 110-133). https://doi.org/10.4324/9781315207766-5

- Waling, A. (2019). Problematizing 'toxic' and 'healthy' masculinity for addressing gender inequalities. *Australian Feminist Studies*, 34(101), 362-375.
 https://doi.org/10.1080/08164649.2019.1679021
- Walrath, C. M., Ybarra, M. L., Sheehan, A. K., Holden, E. W., & Burns, B. J. (2006). Impact of maltreatment on children served in community mental health programs. *Journal of Emotional and Behavioral Disorders*, 14(3), 143-156.
 https://doi.org/10.1177/10634266060140030201
- Wang, C. C., & Geale, S. K. (2015). The power of story: Narrative inquiry as a methodology in nursing research. *International Journal of Nursing Sciences*, 2(2), 195-198. https://doi.org/10.1016/j.ijnss.2015.04.014
- Wetherell, M., & Edley, N. (1999). Negotiating Hegemonic Masculinity: Imaginary Positions and Psycho-Discursive Practices. *Feminism & Psychology*, *9*(3), 335–356. https://doi.org/10.1177/0959353599009003012
- White, M., & Epston, D. (1990). *Narrative means to therapeutic ends*. New York: W. W. Norton.
- White, A. M., & Peretz, T. (2009). Emotions and redefining Black masculinity. *Men and Masculinities*, 12(4), 403-424. https://doi.org/10.1177/1097184x08326007
- Whiteman, S. D., McHale, S. M., & Crouter, A. C. (2010). Family relationships from adolescence to early adulthood: Changes in the family system following Firstborns' leaving home. *Journal of Research on Adolescence*, 21(2), 461-474. https://doi.org/10.1111/j.1532-7795.2010.00683.x
- Willig, C. (2013). Introducing qualitative research in psychology. McGraw-Hill Education

- Young, N. D., Michael, C. N., & Jean, E. D. (2020). Masculinity in the making: Managing the transition to manhood. Rowman & Littlefield Publishers.
- Zurbriggen, E. L. (2010). Gender, sexuality, and the authoritarian personality. *Journal of Personality*, 78(6), 1801-1826. https://doi.org/10.1111/j.1467-6494.2010.00670.x



APPENDIX A – INFORMATION AND CONSENT

Information and Consent Form (Interview)

Study title: Toxic Masculinity, Childhood Sexual Abuse, and the

Challenges to How Young Men Heal

Investigators: Colin J. Morrison, MSW Candidate, Principal Investigator

Dr. Catrina Brown, Thesis Supervisor

Introduction and Purpose:

You have been invited to take part in a research study exploring the experiences of mental health practitioners and community advocates who work with young males who identify as early childhood sexual trauma survivors. The study is voluntary, and before you decide whether you wish to be involved, you need to understand the purpose of the study, what risks might be involved, and what benefits may be gained. This information and consent form explains the study.

The purpose of this study is to explore, through the work and experiences of mental health practitioners and community advocates, the influence of "toxic masculinity" in society today on young male socialization and consider how it may serve to impact and disrupt the healing process for young male survivors of early childhood sexual trauma.

You have been considered for this study because you have been identified as having experience providing therapeutic counselling, support, and advocacy in a mental health and addictions or community setting with young males who have been impacted by early childhood sexual trauma.

Please ask the principal investigator (also known as the lead researcher) or thesis supervisor to clarify anything you do not understand or would like further information about. We hope that all your questions are answered to your satisfaction before deciding whether to participate in this research study

If you decide not to take part or if you leave the study early, your professional practice will not be affected.

The researcher hopes to publish the results of this study in an academic journal and to present findings at academic conferences. No identifying material will be available when presenting the research findings. The purpose of presenting the findings of this research study are to contribute new knowledge and support toward the growing body of literature around mental health, addictions and therapeutic practice with young males who experience early childhood sexual trauma.

How will the researchers do the study?

The researcher will seek to recruit 8 to 12 participants to partake in this study.

The method of this qualitative study involves a commitment to attend a 60 to 90-minute interview with the researcher in a setting of the participant's choice. The participant will be asked a series of questions relating to their knowledge and experience with early childhood sexual trauma, role of dominant masculinity in male socialization, and healing practices specific to working with young male's survivors.

The interview will be conducted in an agreed upon location that best serves you, whether that be in your place of work, a public place (library, coffee shop etc.) or on Dalhousie University campus. Interviews will be audio recorded and later transcribed as part of the data collection and research findings.

You may also be asked if you know other practitioners or advocates who could be potential participants in this study. If you agree, you may be asked to pass on a recruitment email. The interviews will be conducted in the Fall of 2017.

Potential Harms and Burdens.

There are no expected harms of the study. It is possible that, given the nature of the research and the discussion centred on early childhood sexual experience, some participants may be triggered in a way that is emotionally or psychologically distressing.

To mitigate any foreseeable risks, the principal investigator will make himself available for debriefing after the interview as well as continued debriefing and discussion post interview as needed. For IWK/NSHA employees, the possibility of accessing EAP services in relation to any distress from the interview may also be referenced or encouraged.

To protect your information and anonymity, the principal investigator will not keep your name or other information that may identify you with the interview transcript, and instead use only pseudonyms. Files that link your name to the pseudonym will be kept in a secure place separate from the interview data. Although no one can absolutely guarantee confidentiality, using a pseudonym makes the chance much smaller that someone will ever be able to link your name and involvement in this study.

What are the potential benefits?

Taking part may be of no help to you personally. It is hoped that what is learned will be of future benefit to others, and that the participants' knowledge, expertise and insight will help contribute to the conversation around the often under studied and under explored phenomena of male early childhood sexual traumatic experiences.

Can I withdraw from the study?

Participants can opt to withdraw from the study any time prior to the submission of the final Master thesis document, which has a projected completion date of April/May 2018. After this date, it will not be possible to remove your data.

If you wish to withdraw your consent please inform the principal investigator at colin.morrison@dal.ca or by telephone at 902.292.3763. If your data has already been collected, you can decide whether you want any or all the information that you have contributed up to that point removed. If you choose to withdraw from this study, your decision will have no effect on your professional practice.

Are there any conflicts of interest?

As the principal investigator, I wish to acknowledge that I have been employed with the IWK Mental Health and Addictions Program for more than 15 years in the role of youth care worker. I have also completed a social work practicum with NSHA Addictions Community Based Services in 2015-2016 and with the IWK Suspected Trauma and Abuse Response Team in 2017. As such, I could potentially have some formal or informal relationship with participants from the IWK or the NSHA as a past or present mental health colleague. In that regard, I will ensure not to use coercion in recruitment, and remind participants of their right to withdraw from the study at any point in time. Potential or interested participants can also have the option of communicating their interest or fielding questions through my supervisor Dr. Catrina Brown if they so choose and will be informed of that option during recruitment.

Costs and reimbursements.

Participants will not be reimbursed for taking part in the study.

Participating in this study may result in added costs to you such as costs for parking and/or transportation to attend the individual interview.

How will my privacy be protected?

Protecting your privacy is an important part of this study. Every effort to protect your privacy will be made. If the results of this study are presented to the public, participants will not be identified in the study.

The principal investigator will keep the information about you confidential, to the extent permitted by applicable laws. Even though the risk of identifying you from the study data is very small, it can never be completely eliminated.

The principal investigator will keep any identifying information about you in a secure and confidential location. All research records and personal information will be stored in locked cabinet in the principal investigator's home office separate from other data. Electronic data will be stored in an encrypted file on a password protected computer used solely for purposes of this research and also locked in principal investigator's home office. All research data will be retained and safely stored for five years and then destroyed as per IWK research ethics policy and IWK Information Technology standards and policy.

Due to the ongoing monitoring of research activities, there is a possibility that the IWK Research Ethics Board may request to view study data or contact participants directly over the course of the study for quality assurance purposes.

Your personal information will not be shared with others without your direct permission.

What if I have study questions, concerns, or problems?

For questions, clarification, or concerns, please contact the principal investigator, Colin J. Morrison, at 902 292 3763 or colin.morrison@dal.ca. You can also contact the thesis supervisor/supervising investigator, Dr. Catrina Brown at catrina.brown@dal.ca, or the IWK supervising investigator, Coleen Flynn, MSW RSW, at coleen.flynn@iwk.nshealth.ca. You may also wish to contact the IWK Research Ethics Office by contacting Bev White, REB Manager, by e-mail at bev.white@iwk.nshealth.ca or by telephone at (902) 470-8520.

You have the right to all information that could help you decide about participating in this study. You also have the right to ask questions about this study and your rights as a research participant, and to have them answered to your satisfaction before you make any decision.

You have the right to ask questions and to receive answers throughout this study.

You have the right to access, review, and request changes to your study data.

You have the right to be informed of the results of this study once the entire study is complete.

The Research Ethics Board (REB) and people working for or with the REB may also contact you personally for quality assurance purposes.

If you have any questions regarding your specific rights to privacy and confidentiality as research participants, you may contact the IWK Research Ethics Office directly by contacting Bev White, REB Manager, by e-mail at bev.white@iwk.nshealth.ca or by telephone at (902) 470-8520.

How will I be informed of study results?

The results of this research should be available May 2018. If you are interested in receiving the summary results, please leave your preferred contact information in the space provided on the bottom of the information and consent signature page.

On the following IWK signature page, you will be asked if you agree (consent) to join this study. If the answer is "yes", please sign the form in the space provided.

Thank you for your interest in this research study.



APPENDIX B – E-MAIL RECRUITMENT (MANAGERS)

Title of Study: Toxic Masculinity, Male Childhood Sexual Trauma, and the Challenges to How Young Men Heal

Principal Investigator: Colin James Morrison, MSW Candidate, Dalhousie University

Dear Manager:

I am connecting with to provide information on a study I am conducting related to childhood traumatic sexual experiences in male youth, and I am hoping you can assist me in recruiting participants from amongst your wider clinical team.

The purpose of this study is to gather information about the current practice of mental health practitioners and community activists who work with young males under the age of 19 who have experienced early childhood sexual trauma. Current literature suggests that the phenomena of male early childhood sexual trauma is under-recognized, the psycho-social impact of male sexual trauma is not well understood, and male specific services are sorely underdeveloped. This study intends to add to the literature and contribute to the conversation around these early traumatic experiences, while considering improved interventions in treatment when working with these vulnerable male youths.

I am seeking input from mental health practitioners and community advocates who have worked directly with males under the age of 19 who have experienced early childhood sexual trauma.

I ask that you please open and **forward** the attached recruitment letter entitled E-Mail Recruitment (Mental Health Practitioners/Advocates) throughout your clinical team. The attached letter will provide more detailed information on the study itself and the level of commitment involved.

I wish to acknowledge that, as well as an MSW candidate, I am a youth care worker with IWK Child & Adolescent Mental Health and have been employed in this service for more than 15 years.

If you would like further information to assist you before proceeding, regarding myself as researcher or this study, please contact me directly by e-mail at colin.morrison@dal.ca or by telephone at 902 292 3763, or contact thesis supervisor Dr. Catrina Brown at catrina.brown@dal.ca

I thank you in advance for help with recruitment in what I hope will be a helpful and informative research study that will help clarify and better our response to early male childhood sexual trauma experiences.

Thank you, Colin James Morrison, MSW Candidate, Dalhousie University School of Social Work



APPENDIX C - Email Recruitment (Mental Health Practitioners/Advocates)

Dear Mental Health Practitioner/Advocate:

I am conducting a study that hopes to explore, through experiences and insights of mental health practitioners and community advocates, the influence of "toxic masculinity" in society and a consideration of how it may serve to impact and disrupt the healing process for young male survivors of early childhood sexual trauma.

The purpose of this study is to gather information about the current practice of mental health practitioners and community activists who work with young males under the age of 19 who have experienced early childhood sexual trauma. Current literature suggests that the phenomena of male early childhood sexual trauma is under-recognized, the psycho-social impact of male sexual trauma is not well understood, and male specific services are sorely underdeveloped. This study intends to add to the literature and contribute to the conversation around experiences of young males with early childhood sexual trauma. Through the lens of mental health practitioners and community advocates who navigate therapeutic relationships with those impacted, the study will also focus on the impact of masculinity in its many forms on boyhood socialization and healing processes from trauma.

I am seeking input from mental health practitioners and community advocates who have worked directly with males under the age of 19 who have experienced early childhood sexual trauma.

Individuals who volunteer for this study will have the opportunity to participate in a 60-90-minute interview in person in a location of their choosing. During the interview, participants will be asked questions related to their practice and understanding of early childhood sexual trauma and the psycho-social impacts on young male survivors.

This study has been reviewed by the IWK Health Centre Research Ethics Board (REB). There is no reimbursement offered for participation in this study.

If you are interested in participating in this study or would like more information to assist you in making your decision regarding participation, please contact Colin J. Morrison, MSW Candidate, by e-mail at colin.morrison@dal.ca or by telephone at 902 292 3763, or thesis supervisor Dr. Catrina Brown at catrina.brown@dal.ca If interested, you will then be provided with further detailed information about the study as well as the participant information and consent form. If you know of other individuals who may be interested in participating, please feel free to discuss this project with them and ask them to contact this researcher directly for further information, or feel free to circulate this email to any interested parties.

Thank You

Colin J. Morrison, MSW Candidate Dalhousie University School of Social Work

N; K

APPENDIX D - Information and Consent Signature Page

Project Title: Toxic Masculinity, Male Childhood Sexual Trauma, and the Challenges to How Young Men Heal	
Researcher : Colin James Morrison, MSW Candidate, School of Social Work, Dalhousie University	
I, (please print), volunteer to participate in the research project titled Toxic Masculinity, Male Childhood Sexual Trauma, and the Challenges to How Young Men Heal. I understand that the project is designed to gather information about the experiences of mental health clinicians and community advocates who work with young males who identify as early childhood sexual trauma survivors	
My participation in this project is voluntary. I understand that I will not be paid for my participation. I also understand I may withdraw and discontinue participation at any time without penalty.	
Participation involves being interviewed by the lead researcher. The interview will last approximately 60-90 minutes. I agree to the audio recording of the interview and acknowledge the researcher may also take notes during the session.	
I understand that the researcher will not identify me by name in any data collected, and that my confidentiality as a participant in this study will be protected.	
agree direct quotations from my interview may be used without any identifying information.	
understand that this research study has been reviewed and approved by the IWK/NSHA Research Ethics Board.	
have read and understand the explanation provided to me. I have had all my questions answered o my satisfaction, and I voluntarily agree to participate in this study.	
My signature below indicates my consent to participation:	
Name Signature Date f you wish to receive a copy of the research summary upon completion, please provide contact information and the preferred manner in which you wish to be notified, such as mailing inddress or e-mail address, in the space below. If you do not wish to be notified, please leave blank:	

APPENDIX E – INTERVIEW GUIDE

- 1) Can you tell me your approximate age, gender, and place of employment?
- 2) Can you tell me about your role as a mental health practitioner/mental health advocate? (prompts: type of work or advocacy; type of agency your work with)
- Can you describe specific clinical training that you've had? (prompts: degrees, workshops,
 training sessions)
- 4) Is there a theoretical approach/philosophy that you subscribe to that guides your work?
- 5) When working with young males with experiences of early childhood sexual trauma, what was the presenting issue upon referral and what triggered the help seeking?

 (prompts: behavioural issues? mental health challenges? being "forced" to seek help vs "wanting" to seek help?)
- 6) Are there specific strategies you employ that are unique to your experiences with young male survivors?
 (prompts: are there ways you approach this work with young males that differ from how you might work with other populations?)
- 7) Can you describe any challenges and barriers young male survivors face? (prompts: are there issues in the young men's' lives that may impede or impact disclosure? In their environment or community? Within larger society?
- 8) Have these young males usually sought help previously? How do they describe this?
- 9) Do these young males have a hard time expressing what has happened and what its impact or effect has been on them? Why do you think this might be? (What are some commonly expressed emotions? Examples anger, shame, fear, disgust?)
- 10) How do you work to help these young men unpack and understand their experiences?

- 11) Do these young males understand the experience differently if the perpetrator is male versus female?
- 12) In your experience, do the young males blame themselves for the abuse?
- 13) Does the experience of being socialized as "male" impact on how they deal with the sexual abuse, and if so in what ways?
- 14) How might you personally define cultural expectations around masculinity in our world today? How are they influenced and enacted?
- 15) How do expectations around dominant masculinity –or "toxic masculinity" in its most extreme form impact the young males you work with?
- 16) Are there issues or challenges that present regarding a young male's sexual identity formation/sexual orientation that have been impacted by experiences of early childhood sexual trauma? If so, how do you address it?
- 17) Are there specific myths around the phenomena of male early childhood sexual abuse that you see perpetuated in society? (e.g. males "cannot" be abused, all sexual contact for males as pleasurable, sexual orientation as a factor/role in "causing the abuse")

 Are there particular "truths" you wish were more readily acknowledged?
- 18) Do you work to promote healing and foster resilience in young males? If so, how do you do this?
- 19) What is the impact of stigma around early childhood sexual trauma experiences for young males? How do you address the stigma in your practice?
- 20) Is there anything we may have missed, or you wish to expand?

APPENDIX F – RESEARCH APPRROVAL LETTER (RENEWAL)



5850/5980 University Avenue PO Box 9700, Halifax Nova Scotia B3K 6R8 Canada tel: 902.470.8888 www.iwk.nshealth.ca

Approval – Annual Renewal September 1, 2019

Principal Investigator: Colin James Morrison

Title: Toxic Masculinity, Childhood Sexual Trauma, and the Challenges to How Young Men

Heal

Project #: 1022764

On behalf of the IWK Research Ethics Board (IWK-REB) I have examined the application for annual renewal. This request for annual renewal was considered at the July 16, 2019 REB meeting. I am pleased to confirm the Board's approval to continue the study.

The IWK-REB approval will expire on September 01, 2020.

	1
	5,775 1 1 1 1 1 1 2 2 2 2
Adam Huber	
Co-Chair, Research Ethics Board	

This statement is in lieu of Health Canada's Research Ethics Board Attestation: *The Research Ethics Board for the IWK Health Centre operates in accordance with:*

- Food and Drug Regulations, Division 5 "Drugs for Clinical Trials involving Human Subjects"
- The Tri-Council Policy Statement: Ethical Conduct for Research Involving Humans TCPS(2)
- International Conference on Harmonization Good Clinical Practice Guidelines ICH-GCP
- FWA #: FWA00005630 / IORG #: IORG0003102 / IRB00003719